Doc# 2301808080 Fee \$88.00

RHSP FEE:\$9.00 RPRF FEE: \$1.00 Karen a. Yarbrough

COOK COUNTY CLERK

DATE: 01/18/2023 11:41 AM PG: 1 OF 99

This instrument was prepared by:

Cheric Strong
Office of the General Counsel
Chicago Housing Authority
60 East Van Buren Street, 12th Floor
Chicago, Illinois 60605

And after recording, this instrument should be returned to:

Cherie Strong
Office of the General Counsel
Chicago Housing Authority
60 East Van Buren Street. 12th Floor
Chicago, Illinois 60605

41065234P (12 0F36)

GROUND LEASE (AUFORABLE)

This Ground Lease (this "Lease") is made as of the 10th day of January, 2023, by and between:

CHICAGO HOUSING AUTHORITY, an Illinois municipal corporation, having an office at 60 East Van Buren Street, 12th Floor, Chicago, Illinois 60605 ("Landlord" or "CHA"), and HEARTLAND HOUSING, INC., an Illinois not for profit corporation having an office at 208 South LaSalle Street, Suite 1300, Chicago, Illinois 60604 (the "Initial Tenant"), or its permitted successors and assigns. The Initial Tenant, in its capacity as initial tenant under this Lease, together with RS Affordable I LLC, an Illinois limited liability company (the "Owner") as the successor to the Initial Tenant following consummation of the Permitted Assignment (as hereinafter defined) and any successors and/or permitted assigns, are hereinafter eferred to as "Tenant".

RECITALS:

A. Landlord is the owner of fee simple title to all that certain real property located in the City of Chicago, Illinois, more particularly described in Exhibit A attached hereto (the "Real Estate"). In accordance with appropriate resolutions adopted by Landlord, Landlord desires to facilitate the development on the Real Estate consisting of (i) the new construction of three (3) midrise buildings at 1002 South Racine Avenue, 1257 West Roosevelt Road and 1357 West Roosevelt Road in Chicago, Illinois 60608 and related improvements ("Roosevelt Square 3B") and (ii) the development of fifteen (15) rental units at 907 – 909 South Ada Street in Chicago, Illinois 60608 ("Museum"). Roosevelt Square 3B shall include one hundred fifteen (115) affordable low-income housing tax credit ("LIHTC") units, of which seventy five (75) units will

be public housing units ("Public Housing Units"). The fifteen (15) rental units at the Museum shall be affordable LIHTC units, of which five (5) will be Section 8 project-based voucher units (the "PBV Units"). All one hundred and thirty LIHTC units are hereafter referred to as the "Affordable Units". Roosevelt Square 3B and the Museum, including all Affordable Units within them, are hereafter referred to as the "Development".

- B. The Development is a part of a larger redevelopment project including, but not limited to, thirty eight (38) buildings located at several scattered sites along Arthington Street, Taylor Street, Lytle Street, Racine Avenue, Roosevelt Road, Blue Island Avenue, Washburn Street, and 13th Street in Chicago, Cook County, Illinois, ("Roosevelt Square 1") and the Market Development. (as hereafter defined), which shall be financed, rehabilitated, constructed, leased, and operated 25 a single project that includes the Development (the "Redevelopment").
- C. Landford and Tenant have agreed to enter into this Lease in order to implement the Development and to facilitate the rehabilitation, if any, construction and operation and financing thereof.

AGREEMENT:

ARTICLE 1

Lease of the Keal Estate/Term of Lease

1.01 <u>Lease; Term.</u> Landlord, for and in consideration of the rents to be paid and of the covenants and agreements hereinafter contained to be kept and performed by Tenant, hereby leases to Tenant, and Tenant hereby leases from Landlord, the Real Estate;

Together with all right, title and interest of Landlord, if any, in and to any streets, driveways, sidewalks, parkways or alleys adjacent thereto or included within the Real Estate; and

Together with all right, title and interest of Landlord, if any, in, to and under all agreements, easements, rights of way, gores of land, air rights, sewer rights, water courses and water rights, and all privileges, liberties, tenements, and appurtenances whatsoever in any way belonging, relating or appertaining to the Real Estate or which hereafter shall in any way belong relate or be appurtenant thereto, whether now owned or hereafter acquired by Landlord, and the Estate, rights, title, interest, property, possession, claims and demands whatsoever, at law or in equity of Landlord in and to the same:

Subject, however, to all agreements, easements, encumbrances and other charges or matters affecting the Real Estate listed on Exhibit B attached hereto (the "Permitted Exceptions") and subject to the provisions of Section 1.02.

TO HAVE AND TO HOLD the same, subject to the Permitted Exceptions, for a term commencing on the date of this Lease referenced on Page 1 (the "Commencement Date") and ending on August 31, 2103 (the "Expiration Date" and such period from the Commencement Date to the Expiration Date, the "Term"), unless this Lease shall sooner be terminated as hereinafter provided, upon and subject to the covenants, agreements, terms, provisions, conditions

٠,

and limitations hereinafter set forth, all of which Tenant covenants and agrees to perform, observe and be bound by.

- Addition of Vacated Public Alleys and Rights-of-Way. The parties acknowledge 1.02 that all vacations and dedications necessary for, or contemplated in connection with, the Development have been completed prior to the execution of this Lease. In the future, land lying within or comprising existing public alleys and rights-of-way adjacent to portions of the Real Estate may be vacated by the City of Chicago (the "City") and acquired by Landlord and certain streets, passages and other rights-of-way may be dedicated to Governmental Authorities (as that term is hereinafter defined). In each case, with the prior written approval of Tenant and all Leasehold Mortgagees, which shall not be unreasonably withheld or delayed, the foregoing shall automatically, and without the necessity of amending this Lease, be included in (or excluded from, in the event of any such dedication of a portion of the Real Estate to Governmental Authorities), the Real Estate. Upon completion of the alleys and rights-of-way vacation process, however, if requested by Landlor's or Tenant or any Leasehold Mortgagee, the parties shall promptly execute an amendment to this Lease to include a revised legal description for the Real Estate conforming to the revised configuration of the Real Estate, including such vacated portions of the alleys and rights-of-way (or such dedicated portions of the Real Estate). Following completion of construction of the Development, no such dedication shall be permitted without the prior written consent of Tenant and all Leasehold Mortgagees, which consent shall not be unreasonably withheld or delayed.
- Market Units and Commercial Lease. The Landlord, on approximate even date 1.03 herewith is entering into a lease with RS Market 11.LC, an Illinois limited liability company, as tenant, for the construction and operation of ninety-(we (92) market rate units ("Market Units"), commercial retail property, and other related improvements referred to hereinafter as the "Market **Development**". During construction of the Market Development, Owner will be a party to the lease for the Market Development as nominee on behalf of RS Market LLLC. While the Market Units and related improvements may be adjacent to the Affordable Units, the lease agreement for 75 the Market Development shall be wholly separate from this Lease.

ARTICLE 2

Definitions

- The terms defined in this Section shall, for all purposes of this Lease, have the 2.01 following meanings:
- "Affiliate" shall mean, with respect to any person or entity, any other person or entity directly or indirectly controlling, controlled by or under common control with such person or entity. A person or entity shall be deemed to control another person or entity if such person or entity possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of such person or entity, whether through the ownership of voting securities, general partnership or limited liability company interests, by contract or otherwise and shall include, with respect to any so-called "Illinois land trustee" that is at any time the landlord or tenant under this Lease, such land trustee's beneficiary and all persons or entities having the power of direction under such land trust, but shall not include any Leasehold Mortgagee.

- (b) "ACC" shall mean the Consolidated Annual Contributions Contract between HUD and CHA, dated as of December 11, 1995, as amended by the Mixed Finance ACC Amendment, dated of even date herewith and incorporating the Public Housing Units, as the same may be further amended from time to time.
 - (c) "Affordable Units" shall have the meaning given in Recital A.
- (d) "Applicable Public Housing Requirements" shall mean all requirements applicable to public housing, including, but not limited to, the Housing Act, HUD regulations thereunder (and, to the extent applicable, any HUD-approved waivers of regulatory requirements), the ACC, the Mixed Finance ACC Amendment, the HOPE VI Grant Agreement, Landlord's Moving to Work Agreement with HUD, HUD notices (including any notice of fund availability under which Landlord received an award of HOPE VI funds for use in connection with the Project), the HUD-approved Declaration of Restrictive Covenants in favor of HUD, the CHA's admissions and occupancy policies applicable to Roosevelt Square 3B, as set forth in the CHA's approved Moving to Work Annual Plan, and all applicable Federal statutory, regulatory and executive order requirements, as those requirements may be amended from time to time.
- (e) "Building" shall mean a building (including without limitation a low-rise or mid-rise building, a townhome unit and any ancillary building) included in and constructed as part of the Development, or any part thereof. "Buildings" shall mean all of the Buildings comprising the Development. The Market Units in the Buildings comprising Roosevelt Square 3B have been vertically separated from the portion of the Development that is Roosevelt Square 3B via plats of survey that delineate the air rights parcels comprising the Market Development.
- (f) "CHA Mortgage Loan" shall thean a loan secured by a Leasehold Mortgage in favor of CHA, in its role as Lender (or a nontiree for CHA).
 - (g) "City" shall mean the City of Chicago.
 - (h) "Code" shall have the meaning given in Section 9.01(b).
 - (i) "Commencement Date" shall have the meaning given in Section 1.01.
- (j) "Construction Completion Deadline" shall mean thirty-six (26) months after the date of this Lease, the date by which Tenant must complete the construction of the Buildings comprising the Development, evidenced by either a substantial completion conficate from the Owner's architect and either a permanent or temporary certificate of occupancy issued by the City, which date shall be extended by the period of any Unavoidable Delay.
 - (k) "Deductible" shall have the meaning given in Section 8.01.
- (l) "Default" shall mean any condition or event that constitutes or would, after notice or lapse of time, or both, constitute an Event of Default.
- (m) "Developer" shall mean Roosevelt Square 3B LLC, an Illinois limited liability company.

- (n) "Development" shall have the meaning given in Recital A.
- (o) "Encumbrances" shall have the meaning given in Section 9.02.
- (p) "Environmental Agreement" shall mean that certain Remediation Agreement dated as of January 10, 2023, between CHA, Developer and the Owner, relating to the Development.
- (q) "Environmental Laws" shall mean the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. 9601 et seq., as amended; the Resource Conservettor and Recovery Act, 42 U.S.C. 6901 et seq., as amended; the Clean Air Act, 42 U.S.C. 7401 et seq., as amended; the Clean Water Act, 33 U.S.C. 1251 et seq., as amended; the Occupational Setty and Health Act, 29 U.S.C. 655 et seq., as amended: the Illinois Environmental Protection Act, 415 II.CS 5/1 et seq., as amended, and its implementing regulations, including 35 ILL. Admin. Code secs, 740 and 742, as amended: and any other federal, state, local or municipal laws, statutes, regulations, rules or ordinances imposing liability or establishing standards of conduct relating to Hazardovs Materials or protection of the environment, as any of the foregoing may be amended and in effect from time to time.
 - (r) "Event of Default" shall have the meaning provided in Section 10.01.
 - (s) "Excluded Enviror...ental Condition" shall mean the following:
 - (i) all Pre-Existing Environmental Conditions below ground surface at the Real Estate on the date of this Lease except to the extent any such Pre-Existing Environmental Condition is exacerbated by the actions or conduct of Tenant, Owner, Developer and/or any of their agents, contractors, subcontractors, employees, tenants or invitees and except to the extent that Tenant, Owner or Developer has responsibility for such Pre-Existing Environmental Condition under the express terms of the Environmental Agreement or if not addressed under the Environmental Agreement, then under applicable law;
 - (ii) any environmental condition at any off site disposal facility attributable to the Hazardous Materials removed from the Property as part of the Environmental Work (as defined in the Environmental Agreement) prosuant to the Environmental Agreement prior to the issuance of the NFR Leaver for the Development and the completion of all Other Environmental Tasks (as defined in the Environmental Agreement) to which the NFR Letter is not applicable and for which Landlord is the generator pursuant to the Environmental Agreement; provided, however, that this clause (ii) shall not include (A) any such environmental condition at an off-site disposal facility arising from any such Hazardous Materials that Tenant, Owner, Developer or any of their respective contractors or consultants fail to properly characterize as such environmental condition in connection with the work and activities performed by Tenant, Developer or their agents, contractors or consultants pursuant to the Environmental Agreement; (B) any Hazardous Materials that are removed from the Development for which Landlord is not the generator under the express provisions of the

Environmental Agreement; and (C) any Hazardous Materials that are the subject of Other Environmental Tasks;

- (iii) any environmental condition caused or exacerbated by Landlord or any of its agents, contractors, subcontractors or employees after the date of this Lease, except to the extent that Tenant, Owner or Developer has responsibility for such environmental conditions under the express terms of the Environmental Agreement or, if not addressed under the Environmental Agreement, then under applicable law: and
- (iv) with respect to Tenant, any environmental condition that first occurs after a Terminating Event with respect to Tenant unless and to the extent that such confironmental condition was caused or exacerbated by Tenant, Owner or Developer or their agents, contractors, subtenants, employees, tenants, guests or invitees prior to such Terminating Event or otherwise occurred during the period of Tenant's tenancy under this Lease; provided however, that if the Terminating Event with respect to Tenant consists of the sale or transfer of Tenant's interest in the Tenant Property, any environmental condition that first occurs after such sale or transfer shall not be an Excluded Environmental Condition with respect to the purchaser or transfered of Tenant's interest in the Tenant Property.
- (t) "Expiration Date" shall have the meaning provided in Section 1.01.
- (u) "Final Completion" shall have the meaning provided in Section 5.01.
- (v) "First Leasehold Mortgage" shail mean the Leasehold Mortgage of the First Leasehold Mortgagee.
- (w) "First Leasehold Mortgagee" shall meen the Leasehold Mortgagee whose Leasehold Mortgage is most senior in priority of lien, as identified on Exhibit C attached hereto.
- (x) "Full Insurable Value" shall mean the replacement cost (excluding, as to the insurance required pursuant to Section 7.01, foundation and excaration costs) of the Improvements, as determined, at the request of Landlord (not more frequently that at three-year intervals), at Tenant's expense, by an architect, engineer, contractor, appraiser, appraised company, or insurance company, selected by Tenant and approved by Landlord, which approval chall not be unreasonably withheld.
 - (y) "Full Restoration" shall have the meaning given in Section 8.01.
- (z) "Governmental Authority" or "Governmental Authorities" shall mean any one or more of the federal, state and local governmental or quasi-governmental body or bodies having jurisdiction at any time or from time to time during the Term over the Real Estate or the Property, or any part thereof, or the construction, repair, maintenance, operation or use thereof.
- (aa) "Hazardous Condition" shall have the meaning given in the Environmental Agreement.

- (bb) "Hazardous Material" means any pollutant or contaminant or any hazardous, toxic or radioactive waste, substance or material, in any form, including, but not limited to, those pollutants, contaminants, wastes, substances, or materials defined or classified as such in, or listed in or regulated under any Environmental Laws, including without limitation, polychlorinated biphenyls (PCBs), petroleum or any petroleum-based or petroleum-derived products, lead-based paint, asbestos or asbestos-containing materials, mold, urea formaldehyde and radioactive materials.
- (cc) "HOPE VI Grant Agreement" shall mean the HOPE VI Revitalization Grant Agreement relating to the revitalization of the Property, by and between HUD and CHA, date as of February 12, 2001, as may be amended.
- (34) "Housing Act" shall mean the United States Housing Act of 1937, as amended and as matchereafter be amended from time to time or any successor legislation, together with all regulations in promenting the same.
- (ee) "HUD" shall mean the United States Department of Housing and Urban Development, or any successor mereto.
- "Impositions" shall mean all taxes (including payments in lieu of taxes), assessments, special assessments, use and occupancy taxes, water and sewer charges, rates and rents, charges for public utilities, excises, levies, license and permit fees and other charges, general and special, ordinary and extraordinary, foresten and unforeseen, of any kind or nature whatsoever, which shall or may during the Term be assessed, levied, charged, confirmed or imposed upon or become payable out of or become a tien on the Property, Leasehold Estate or the Real Estate, or any part thereof, or any appurtenances increto but specifically excluding amounts related to an Excluded Environmental Condition and/or payable by Landlord under the Environmental Agreement; provided, however, that if at any time during the Term the present method of taxation or assessment shall be so changed that there shall be substituted in whole or in part for the types of taxes, assessments, levies, assessed or imposed on real estate and the Improvements thereon a capital levy or other tax levied, assessed or improved on the rents received by Landlord from said real estate or the rents reserved herein or any part thereof, then any such capital levy or other tax shall, to the extent that it is so substituted, be deemed to be included within the term "Impositions." Impositions affecting the Property or the Real Escate shall be those attributable to the Improvements, the Leasehold Estate, and/or the fee simple ownership of the Real Estate.
- (gg) "Improvements" shall mean any structures and other improvements, including the Buildings, equipment, fixtures, furnishings and appurtenances, now or at any time hereafter erected or located on the Real Estate.
- (hh) "Investor" shall mean Hudson Roosevelt Square LLC, a Delaware limited liability company, and its successors and assigns, and Hudson SLP LLC, a Delaware limited liability company, and its successors and assigns.
- (ii) "Landlord" or "CHA" shall mean Chicago Housing Authority, or its successors in interest.

- plus the rate announced from time to time by JP Morgan Chase Bank, N.A. (or any successor thereto), as its "corporate base rate," "prime rate," "reference rate" or other similar rate and in effect on the date interest first begins to accrue with respect to any sum that becomes payable pursuant to any provision or provisions of this Lease, or (ii) in the event such bank has ceased announcing any such rate, then such rate as may be announced by the Chicago branch of such other national bank as Landlord shall reasonably designate as its "prime rate" "reference rate" or other similar rate, plus 3%, or (iii) if Landlord fails to designate another bank, then the rate of interest on 90-day Treasury Bills issued by the United States government having an issue date as near as may be practicable to and preceding such date plus 6%; provided, however that if the Lease Interest Rate as so determined shall exceed the maximum rate allowed by law, then the "Lease Interest Rate shall change concurrently with each announced change in such "corporate base rate," "prime rate," "rete ence rate" or other similar rate, or Treasury Bill rate.
- (kk) Lease Year" shall mean a calendar year. The first full Lease Year during the Term shall commence on the Commencement Date, or if the Commencement Date is not the first day of a calendar year, on the first day of the calendar year next following the Commencement Date. Each succeeding Lease Year shall commence on the January 1 immediately following the December 31 of the preceding Lease Year. If the Commencement Date is not January 1, that portion of the Term that is prior to beginning of the first full Lease Year shall be a partial Lease Year. If the Expiration Date is not December 31 that portion of the Term that is after the end of the last full Lease Year shall be a partial Lease Year.
- (II) "Leasehold Estate" shall mean the leasehold estate of Tenant in the Real Estate created by this Lease more particularly described in Exhibit A attached hereto.
- (mm) "Leasehold Mortgage" shall mean any rior gage, deed of trust, assignment of rents and leases, Uniform Commercial Code security agreement and financing statement, or similar security instrument created by Tenant and which constitutes a lien or security interest on the Tenant Property or any part thereof in favor of (i) the Initial Leasebald Mortgagees identified on Exhibit C attached hereto pursuant to and in accordance with the provisions of Section 9.03, (ii) a Lender providing any Permitted Refinancing or (iii) any Lender making any other loan to Tenant that is consented to in writing by Landlord, which consent may be with a granted or granted with such conditions as Landlord may require, in Landlord's sole and absolute discretion.
- (nn) "Leasehold Mortgage Loan" shall mean a loan secured by a Leasehold Mortgage.
- (00) "Leasehold Mortgagee" shall mean the owner or owners, holder or holders from time to time of any Leasehold Mortgage (including trustees under deeds of trust).
- (pp) "Lender" shall mean any commercial real estate lender, state or national bank, commercial or savings bank, pension fund, real estate investment trust, or governmental agency or instrumentality, or any HUD-approved mortgagee, or any Affiliate of the foregoing, authorized to make loans secured by real property located in the State of Illinois.

- (qq) "Loan Documents" shall mean any and all documents, instruments or agreements in effect from time to time evidencing and securing a Leasehold Mortgage Loan.
- (rr) "Low Rent Housing Abatement" shall mean any real estate tax abatement provided to a portion of the Real Estate or the Improvements pursuant to 35 ILCS 200/18-177, as the same may hereafter be amended from time to time.
- (ss) "Manager" shall mean RS 3B LLC, an Illinois limited liability company, the manager of the Owner, and any permitted successor.
- (tt) "Mixed Finance ACC Amendment" shall mean that certain Mixed Finance Amendment to the Consolidated Annual Contributions Contract executed by Landlord and HUD dated January 10, 2023.
 - (uu) "Net Insurance Proceeds" shall have the meaning given in Section 8.02.
- (vv) "NFR Letter" shall mean a "No Further Remediation" letter issued by the Illinois Environmental Protection Agency pursuant to the Site Remediation Program, 415 ILCS 5/58 et. seq., as amended from time to time, with respect to any portion of the Real Estate.
- (ww) "Owner" shall ocen RS Affordable 1 LLC, an Illinois limited liability company.
- (xx) "Organizational Documents" shall mean: (i) with respect to a corporation, its articles of incorporation and by-laws; (ii) with respect to a general partnership, its partnership agreement; (iii) with respect to a limited partnership and limited partnership agreement; and (iv) with respect to a limited liability company, its articles of organization and operating agreement; in each case as amended prior to such entity becoming Tenant under this Lease and as amended from time to time thereafter; provided, however, that no amendment to any Organizational Document that materially adversely affects the rights of Landlord may be made after such entity becomes Tenant hereunder except as required by law, consented to in writing by Landlord or is made to effect a transfer or substitution of interests in Tenant which does not otherwise require the consent of Landlord hereunder, and any amendment that contravenes this prohibition shall be null and void.
- (yy) "Partial Restoration" shall mean all work in connection with Pestoration that is less than a Full Restoration (see Article 8). A Partial Restoration may be applicable when the sum of the Net Insurance Proceeds plus Deductible plus all other monies provided by any Person for such Restoration are insufficient to accomplish a Full Restoration. Examples of Partial Restorations include: (1) if a six-flat is destroyed, building a three-flat or townhomes on the lot; (2) if the top unit in a three-flat is destroyed, making the Building into a two-flat; and (3) if an end townhome unit is destroyed, not rebuilding that unit and making the adjoining unit into an end unit.
- (zz) "PBV HAP Contract" shall mean that certain Section 8 Project-Based Voucher Housing Assistance Payments Contract for the PBV Units by and between Landlord, as contract administrator, and Owner dated substantially even date herein.

- (aaa) "PBV Requirements" shall mean all HUD requirements applicable to the PBV Units, including but not limited to 24 CFR Part 983 and the PBV HAP Contracts.
 - (bbb) "PBV Units" shall have the meaning given in Recital A.
- (ccc) "Permitted Assignment" shall mean the assignment of the Leasehold Estate by the Initial Tenant to the Owner occurring (i) subsequent to the full execution and delivery of this Lease by the Landlord and the Initial Tenant and (ii) pursuant to an Assignment and Assumption and Amendment of the Ground Lease between Initial Tenant and the Owner.
- (ddd) "Permitted Exceptions" shall have the meaning given in Section 1.01. "Permitted Exceptions" shall also include the easements and licenses, if any, hereafter granted or consented to by Landlord in accordance with Section 9.01(d) and any deed restrictions required by the NFR Leater.
- (eec) "Permitted Refinancing" shall mean (a) with respect to any loan secured by a Leasehold Mortgage, any refinancing of such loan permitted under the documents evidencing or securing a CHA Mortgage Loan including without limitation, a refinancing of the initial Leasehold Mortgage Loan securer; by the First Leasehold Mortgage; (b) with respect to any loan secured by a Leasehold Mortgage, any refinancing of such loan (other than as provided in (a)), consented to in writing by Landlord, which consent may be withheld, granted or granted with such conditions as Landlord may require, in Landlord's sole and absolute discretion; and (c) any additional loan secured by a Leasehold Mortgage that is either expressly permitted under the terms of this Lease or consented to in writing by Landlord, which consent may be withheld, granted or granted with such conditions as Landlord may require, in Landlord's sole and absolute discretion.
- (fff) "Permitted Transfer" shall mean" (a) the Permitted Assignment; (b) after the completion of the construction of the Development, a sale or transfer of the Tenant Property or any portion thereof to a person reasonably acceptable to Landlord; (c) the removal of a general partner, limited partner, member or manager of Owner pursuant to a Removal Right in accordance with the provisions of Owner's Organizational Documents, and the substitution of a replacement general partner, limited partner, member or manager, as the case may be, reasonably acceptable to Landlord, (d) the transfer of the Tenant Property, or any portion thereof, to a Leasehold Mortgagee (or any nominee of such Leasehold Mortgagee) by foreclosure or deed in lieu of foreclosure or to a third party purchaser at a foreclosure sale in accordance with Section 9.03(a); (e) a transfer permitted under the documents evidencing and/or securing the CHA Mortgage Loan; and (f) any other transfer consented to by Landlord in writing. Notwithstanding the foregoing, a Permitted Transfer does not include the sale or transfer of commercial and/or retail space, if any in the Development, without the written approval of Landlord.
- (ggg) "Person" shall mean any person, corporation, partnership, limited liability company or other legal entity.
- (hhh) "Plans and Specifications" shall mean the plans and specifications for the Development, which have been approved by Landlord and the Leasehold Mortgagees, as such plans and specifications are amended from time to time with the written consent of Landlord and the Leasehold Mortgagees, if and as required by the applicable loan documents.

- (iii) "Pre-Existing Environmental Condition" shall mean any Hazardous Materials and any Hazardous Condition present on, under or in the Real Estate on the date of execution of this Lease or the Environmental Agreement, whether known or unknown.
- (jjj) "Proceeds" shall mean, in the case of damage to or destruction of the Improvements, the sum of the Net Insurance Proceeds plus the Deductible, and, in the case of a condemnation or other taking (or conveyance in lieu thereof), the awards (or compensation paid) therefor.
 - (kkk) "Property" shall mean the Real Estate and the Improvements.
- (III) "Protected Persons" shall mean Landlord or Tenant, as the context so requires, and such party's respective members, managers, partners, officers, directors, agents, employees, advicers, attorneys, consultants and Affiliates, and, in the case of Landlord, shall include its officials and members of its Board.
- (mmm) **Public Housing Units**" shall mean the seventy-five (75) Affordable Units.
 - (nnn) "Real Estate" shall have the meaning given in Recital A.
- (000) "Remediation" shall have the meaning set forth in the Environmental Agreement.
- (ppp) "Removal Right" shall mean the right, if any, given in Owner or Tenant's Organizational Documents, to remove an officer, affector, general partner, limited partner, member, manager or managing member of Tenant or Owner, and designate a substitute, provided that Tenant or Owner shall comply (or cause its members to comply) with the terms and conditions of the Loan Documents, as applicable.
 - (qqq) "Rent" shall have the meaning given in Section 3.01.
- (rrr) "Requirements" shall mean any and all present and future laws, statutes, ordinances, codes, rules, regulations, orders or other requirements of any Governmental Authority and of any applicable fire rating bureau or other body exercising similar functions; applicable to or affecting the Real Estate, Leasehold Estate or the Property, or any part thereof, including without limiting the generality of the foregoing, the ordinances of the City.
 - (sss) "Restoration" shall have the meaning given in Section 8.01.
- (ttt) "Substantially Commenced" shall mean with respect to any Building to be constructed, that the footings and foundation walls of such Building have been completed (including all necessary excavation work) and one or more of Borrower or Leasehold Mortgagees has disbursed equity and/or loan proceeds to pay the "hard costs" of such construction.
- (uuu) "Tenant" shall mean the party named as Initial Tenant herein; provided, however, that whenever this Lease and the Leasehold Estate shall be assigned or transferred in the manner specifically permitted herein, then from and after such assignment or transfer and until the

2301808080 Page: 12 of 99

UNOFFICIAL COPY

next such assignment or transfer, the term "Tenant" shall mean the permitted assignee or transferee named therein, as if such transferee or assignee had been named herein as Tenant. From and after the Permitted Assignment, the Initial Tenant will be released from, and permitted assignee or transferee shall assume, any and all obligations under this Lease arising from and after the Permitted Assignment, including, without limitation, the obligations set forth in Section 11.02.

(vvv) "Tenant Property" shall mean the Improvements and the Leasehold Estate.

(www) "Term" shall mean the term of this Lease described in Section 1.01.

Tenant Property to another party not affiliated with such Tenant (other than the Permitted Assignment of as a Permitted Transfer); (ii) transfer of title to such Tenant's interest in the Tenant Property pursuant of foreclosure of, or deed in lieu of foreclosure with respect to, any mortgage or other security instrument securing loans or advances with respect to the Development; (iii) termination of this Lease; or (iv) loss of such Tenant's possession of the Tenant Property pursuant to the appointment of a receiver or pursuant to the exercise by any Leasehold Mortgagee of its right to become a mortgagee in possession. In the event that any of the foregoing occur with respect to less than all of the Tena it Property, or if there is a partial termination of this Lease, then a Terminating Event shall be deemed to have occurred only with respect to such portion of the Tenant Property or that portion of the Property affected by such partial termination.

(yyy) "Unavoidable Delay" s'iall mean a delay beyond the reasonable control of Tenant and without the fault or negligence of Tenant, such as: (1) an act of God; (2) fire; (3) flood; (4) epidemic/pandemic; (5) quarantine restriction; (6) civil disorder; (7) enemy action; (8) strike, lockout or other labor dispute; (9) unavailability of labor or materials: (10) freight embargo; (11) action or inaction of Landlord; (12) the act or failure to act of a contractor in the performance of a contract with Landlord; (13) the act or failure to act of any Governmental Authority; (14) injunctive relief or other legal proceedings of any court; (15) war; (16) terrorism; (17) the failure of the City to timely complete any public improvements necessary for Tenant to construct the Development; (18) unforeseen soil conditions, such as underground's orage tanks and building foundations and/or any Unforeseen Condition under the Environmental Agreement; (19) delays caused by Landlord (other than Landlord exercising its rights under this Lease and the Loan Documents) affecting the construction of the Development including delays resulting from the suspension by Tenant of remediation work due to the Landlord's failure to perform its obligations pursuant to Section 4 of the Environmental Agreement; and (20) unusually severe weither.

2.02 "The words "herein," "hereof" or "hereunder" and words of similar import refer to provisions contained in this Lease as a whole and not to any particular section or subdivision thereof. All exhibits and riders referred to in the text of this Lease and attached hereto are incorporated into this Lease.

ARTICLE 3

Rent

- 3.01 Rent. From and after the Commencement Date through the Term, Tenant shall pay to Landlord at the place for which notices to Landlord are to be sent in accordance with Article 16, or to such other Person and/or at such other place as shall be designated from time to time by written notice from Landlord to Tenant, fixed rent at the rate of One Dollar (\$1.00) for each Lease Year ("Rent"). Landlord acknowledges that such Rent, in the amount of \$82.00, has been prepaid in full concurrently with the execution of this Lease.
- 3.02 No Partnership. Landlord and Tenant agree that they are not partners or joint venturers and that, except in respect to the proceeds of insurance and condemnation awards under the provisions of A. icle 8 and Article 12, they do not stand in any fiduciary relationship to each other.
- 3.03 Payment of Rent. All payments of Rent made to Landlord hereunder shall be made in lawful money of the United States of America and shall be paid to Landlord or to such other Person and/or at such other place as Landlord may designate from time to time in writing.
- 3.04 Net Lease. Tenant shair pay to Landlord throughout the Term all Rent, free of any charges, assessments, Impositions or deductions of any kind and without abatement, deduction or set-off. Under no circumstances or conditions, whether now existing or hereafter arising, or whether beyond the present contemplation of the parties, shall Landlord be expected or required to make any payment of any kind whatsoever hereunder or be under any other obligation or liability hereunder except as otherwise expressly set forth herein.
- 3.05 No Abandonment. Except to the extent provided in Section 8.01, no happening, event, occurrence, or situation during the Term, whether foreseen or unforeseen, and however extraordinary, shall relieve Tenant from its liability to pay the full Rent, or relieve Tenant from any of its other obligations under this Lease. Tenant waives any rights now or hereafter conferred upon it by statute, proclamation, decree, or otherwise, or to claim any abatement, diminution, reduction or suspension of the Rent on account of any such event, happening, occurrence or situation.
- 3.06 Reimbursements to Landlord; Arrearages. Tenant shall reimburse 1 and ord for all reasonable expenditures, costs, expenses and fees, including reasonable attorneys' tees made or incurred by Landlord in curing any Event of Default of Tenant for which Landlord has given Tenant the notice required under Section 10.01, if any. Such amounts shall become due upon delivery by Landlord, after the expiration of the notice and cure period afforded Tenant, if any, of written notice stating the amount of such expenditures, costs, expenses and fees by Landlord. Tenant shall also pay to Landlord upon delivery of notice by Landlord, all amounts payable to Landlord as reimbursements or indemnities pursuant to 6.03 and 6.04.
- 3.07 <u>Interest on Overdue Amounts</u>. All Rent and other amounts due to Landlord hereunder that are not paid prior to the expiration of the applicable cure period, if any, shall bear interest at the Lease Interest Rate from time to time in effect from the due date to the date received

by Landlord. Such interest shall be payable by Tenant to Landlord upon demand. The collection of such interest by Landlord shall not limit or modify any other right or remedy of Landlord under this Lease or otherwise available to Landlord by reason of Tenant's failure to pay such amount when due or by reason of any other Event of Default.

ARTICLE 4

Impositions

- 4.01 <u>Payment</u>. Throughout the Term, subject to the provisions of Section 4.04, Tenant shall pay or cause to be paid, as and when the same become due, all Impositions, except that:
- All Impositions attributable on the accrual basis to a calendar year or other period for which this Lease is in effect for less than the entire calendar year or other period shall be equitably apportioned (taking into account that Landlord may be entitled to exemptions or abatements) consistent with the time a party hereto held its respective interests in the Real Estate and Improvements;
- (b) Where any 'mposition is permitted by law to be paid in installments, Tenant may pay such Imposition in installments, as and when each such installment becomes due (Tenant acknowledges and agrees that Tenant is obligated to pay all such installments of any Imposition from which Landlord is or would be exempt, whether such installment is due prior to or after the Expiration Date or the date of any earlier termination of this Lease); and
- (c) Where any Imposition is envited to an abatement, refund, exemption or other diminution or reduction under law, whether avanuable to Landlord or Tenant, the parties shall use their best efforts, at Tenant's sole expense, to cause such benefits to be afforded to Tenant under this Lease.

4.02 Deposit of Impositions.

- (a) Tenant shall timely pay, as additional rent, all Impositions, and all premiums on insurance required to be carried under Article 7, as and when the same are ascertainable, billed, and due and payable without interest, penalty or fine. Within thirty (30) days after Landlord's written request, Tenant shall deliver reasonable proof of such payment to Landlord.
- (b) During the continuance of any Event of Default, Tenant agrees to deposit with Landlord on the first day of each and every month thereafter during the Term one-twelfth (1/12) of (a) all Impositions due and payable from Tenant during the next succeeding 12-month period, based on the most recent ascertainable Impositions, plus (b) annual premiums on insurance policies required to be carried by Tenant under Article 7. Further, upon the occurrence of any Event of Default, Tenant shall deposit, at least thirty (30) days prior to the due date of any Imposition, such additional amount as may be necessary to provide Landlord with sufficient funds in such deposit account to pay each such Imposition and annual insurance premium at least thirty (30) days in advance of the due date thereof. The rights granted hereunder to Landlord shall not be exclusive to Landlord's rights and remedies following an Event of Default by Tenant. Landlord shall have no obligation to pay interest to Tenant on any amounts deposited by Tenant. Landlord

shall apply any such deposits for the purpose held not later than the last day on which any such charges may be paid without interest or penalty. If, at any time, the amount of any Imposition or insurance premium is increased or Landlord receives reliable information from a Governmental Authority or insurer, as applicable, that an Imposition or insurance premium will be increased, and if the monthly deposits then being made by Tenant for such item (if continued) would not produce a fund sufficient to pay such item thirty (30) days prior to its due date, such monthly deposits shall thereupon be increased and Tenant shall deposit with Landlord, on demand by Landlord, additional sums in an amount which, when added to the monies then on hand for the payment of said item plus the increased one-twelfth (1/12) payments, shall be sufficient to pay such item at least thirty (30) days before the same becomes due and payable. For purposes of determining whether Landlord has on hand sufficient monies to pay any particular Imposition or insurance premium at least thirty (30), days prior to the due date therefor, deposits for each item shall be treated separately, it being the intention that Landlord shall not be obligated to use monies deposited for the payment of any item for the payment of another that is due and payable. Tenant shall not be required to make any specific deposit required under this Section if a deposit for the same purpose is made by Tenant to an excrow or otherwise to persons pursuant to a requirement by any Leasehold Mortgagee. This Section 4.12 shall not apply to any Leasehold Mortgagee (or any nominee of a Leasehold Mortgagee) that becomes Tenant hereunder through foreclosure or transfer by deed in lieu of foreclosure unless: (i) the Event of Default arises after such Leasehold Mortgagee becomes Tenant hereunder; or (ii) the Event of Default arises prior to such Leasehold Mortgagee becoming Tenant hereunder and such Event of Default is not an Incurable Default and is an Event of Default that such Leasehold Mortgagee is obligated to cure under the provisions of this Lease, and such Leasehold Mortgagee fails to commence or cure such Event of Default within the time and in the manner required by this Lease.

- Contest of Impositions. Tenant may, if it desires, contest the validity or amount of any Imposition, in whole or in part, by an appropriate proceeding diligently conducted in good faith. Tenant may conduct such a contest only after payment of the challenged Imposition or Tenant shall, at least fifteen (15) days prior to the date such Imposition is due: (i) have deposited with the First Leasehold Mortgagee or an escrow agent acceptable to Landlord an amount sufficient to pay such contested Imposition, together with interest and penalties thereon, which amount shall be applied to the payment of such Imposition, interest and penalties when the amount thereof shall be finally fixed and determined; or (ii) have provided to the First Leasehold Mortgagee or to Landlord a bond, letter of credit or other security reasonably acceptable to Landlord. Nothing herein contained, however, shall be construed as to allow such lapposition to remain unpaid for such length of time as shall permit the Property. Leasehold Estate or any part thereof, or the lien thereon created by such Imposition, to be sold or forfeited for the nonpayment of the same. If the amount so deposited as aforesaid shall exceed the amount of such Imposition, interest and penalties when finally fixed and determined, the excess (or the entire amount if no such payment is required) shall be released from the escrow to Tenant, or in case there shall be a deficiency, the amount of such deficiency shall be forthwith paid by Tenant.
- 4.04 Reduction of Impositions. Tenant, at its expense, may, if it shall so desire, endeavor at any time or times, upon prior written notice to Landlord, to obtain a lowering of the assessed valuation upon the Real Estate or the Property for the purpose of reducing taxes thereon and, in such event, Landlord will offer no objection and, at the request of Tenant, will cooperate with Tenant, but without expense to Landlord, in effecting such a reduction. Tenant shall be

authorized to collect any tax refund payable as a result of any proceeding Tenant may institute for that purpose and any such tax refund shall be the property of Tenant to the extent to which it may be based on a payment made by Tenant, subject, however, to the apportionment provisions contained in Section 4.01, after deducting from such refund the costs and expenses, including legal fees, incurred in connection with obtaining such refund.

4.05 <u>Joinder of Landlord</u>. Landlord shall not be required to join in any action or proceeding referred to in Sections 4.03 or 4.04 unless required by law or any rule or regulation in order to make such action or proceeding effective, in which event, any such action or proceeding may be taken by Tenant in the name of, but without expense to, Landlord. Notwithstanding the foregoing, Landlord shall execute, when and as required and requested to do so by Tenant in writing, all applications, affidavits and other documents required to obtain or maintain any tax abatement or exemption which may be available. Tenant hereby agrees to indemnify, defend and hold Landlord's Protected Persons harmless from and against all costs, expenses, claims, loss or damage, including reasonable attorney's fees, by reason of, in connection with, on account of, growing out of, or resulting from, any such action or proceeding.

Tax Divisions. Tenant shall, (1) within thirty (30) days following the execution and delivery of this Lease, send intilication to the Cook County Assessor's Office, Exemption Department, via certified mail, return receipt requested (with a copy to the City of Chicago, Department of Housing) of a change in the Leasehold Estate in accordance with 35 ILCS 200/9-185 and/or 35 ILCS 200/15-20, as applicable, and (2) within thirty (30) days following Tenant's receipt of the new Permanent Index Numbers for the Real Estate file a Petition for Division and/or Consolidation of Property ("Petition") with the Cock County Assessor's Office for a real estate tax division segregating the Landlord's fee interest in the Real Estate from Tenant's Leasehold Estate and ownership of the Improvements. Landlord and Tenant acknowledge that portions of the Real Estate are or may be included in tax parcels ("Shared Tax Parcels") that include land owned by Landlord other than the Real Estate ("Other Land"). Tenant is responsible for providing a copy of its filed Petition(s) to Landlord and monitoring the Division Report generated by the County Assessor's Office to ensure the assignment of new, separate tax parcel designation for the Real Estate and Leasehold Estate. Until such tax parcel re-designation occurs, Landlord agrees to pay or cause to be paid, when due, all property taxes assessed against the Other Land, and Tenant agrees to pay or cause to be paid; when due (or, if paid by Landford) to reimburse Landlord upon demand for) any property taxes attributable to the Real Estate, co any portion thereof that are taxed as part of a Shared Tax Parcel. Within fifteen (15) days of receipt of any new, separate tax parcel re-designation, Tenant shall provide Landlord with copies of all such notice of assignment of new, separate tax parcel re-designation for the Real Estate, Leasehold Estate and Other Land, if any. Landlord or Tenant may, if either shall so desire, contest the validity or amount of any such taxes, in whole or in part, by an appropriate proceeding diligently conducted in good faith. Any such contest by Tenant shall be in accordance with Section 4.01. Tenant will promptly forward on to Landlord copies of any property tax bills it receives covering the Other Land. Tenant's failure to fully comply with this Section shall constitute an Event of Default under Section 10.01 of this Lease.

2301808080 Page: 17 of 99

UNOFFICIAL COPY

ARTICLE 5

Improvements

- **Required Improvements.** Except in the event of termination pursuant to Section 5(d)(iv) of the Environmental Agreement. Tenant hereby covenants and agrees to commence and diligently pursue the construction of the Improvements on the Real Estate, in accordance with the Plans and Specifications, and obtain Final Completion. Landlord agrees that, upon receipt of written request from Tenant or a Leasehold Mortgagee, Landlord will cooperate with Tenant in applications for permits, licenses or other authorizations required for such Improvements; provided, however, that all expenses in connection therewith shall be borne by Tenant. Nothing herein shall relieve Tenant from the responsibility of achieving Final Completion except as expressly provided herein. "Final Completion" shall mean (1) delivery by Tenant to Landlord of a final certificate of occupancy from the City with respect to the last Building in the Development and (2) delivery by Tenaut to Landlord of a certificate of final completion from the Development's architect with respect to the last Building in the Development. Notwithstanding anything herein to the contrary, within ten (10) business days of receipt of any certificate of occupancy, Tenant shall forward copies of any such certificates of occupancy received to the Cook County Assessor's Office and the City of Chicago, Department of Housing. Tenant's failure to fully comply with this Section shall constitute an Event of Default under Section 10.01 of this Lease.
- Other Capital Improvements. With respect to any Major Capital Improvement other than the Buildings that Tenant desires to construct on the Real Estate, Tenant shall not commence construction unless Landlord shall have specifically approved such Major Capital Improvement and Tenant has complied with Section 5.93, provided that Landlord's approval shall not be unreasonably withheld or delayed. A "Major Capital Improvement" is a capital improvement involving an estimated cost of more than \$100,000.00, and shall include additional buildings and additions, alterations, renovations, restorations, replacements or rebuilding, whether or not required to be made in compliance with Tenant's obligations under this Article, or in connection with a Restoration made under Article 8 as a result of damage or destruction, or under Article 12 as a result of any taking pursuant to eminent domain. Notwinstanding the foregoing, Landlord's consent under this Article 5 shall not be required in connection with: (i) a Restoration of the Improvements under Article 8 or Article 12 to the condition that existed immediately prior to the casualty or condemnation; or (ii) a capital improvement that is required pursuant to Requirements.
- 5.03 <u>Major Capital Improvements Requirements</u>. Prior to the commencement of any Major Capital Improvement, the following shall be submitted to Landlord:
- (a) complete plans and specifications for the Major Capital Improvement prepared by a licensed architect which plans shall also include landscaping plans and specifications;
- (b) copies of all permits and licenses for the construction of the Major Capital Improvement issued by the appropriate Governmental Authority;

- (c) a signed construction contract or contracts for all of the work, material and equipment comprising the Major Capital Improvement in accordance with the plans and specifications delivered pursuant to Section 5.03(a), together with appropriate liability insurance policies; and
- (d) a copy of the written consent from each of the Leasehold Mortgagees approving the Major Capital Improvements, its inclusion in the respective Loan Documents and/or a copy of one or more commitments from a Lender or Lenders for loans to be made available to Tenant, on both a construction loan and long-term take-out loan basis, in an amount that, together with equity that is available and specifically allocated thereto, is sufficient to pay the budgeted costs of construction of the Major Capital Improvement.
- 5.04 <u>Demolition</u>. Except in connection with a Restoration under Article 8 or 12 or as provided in the Plans and Specifications, Tenant shall not demolish the Improvements, including any improvements to such Improvements required under Section 5.01 or any Major Capital Improvements permitted under Section 5.02, without the prior written consent of Landlord and subject to terms and conditions of the Loan Documents, as applicable.
- Accessibility and Visitability Requirements. The Redevelopment is subject to the accessibility requirements of the Fair Housing Act and Section 504 of the Rehabilitation Act of 1973 ("Section 504") and shall be designed and constructed in accordance with such requirements. To the greatest extent feasible, all units in the Buildings in the Redevelopment shall be built in accordance with the concept of Vistability, which recognizes that persons with disabilities should be able to enjoy the same privileges of accessibility to other living quarters outside their residence. "Visitability" means that: (a) at least one entrance is at grade (i.e. no steps), approached by an accessible route: and (b) the entrance door and all interior doors on the first floor are at least 34 inches wide, offering 32 inches of clear passage space. Redevelopment shall also be constructed to include common and exterior areas that comply with the design and construction requirements of all applicable federal, state, and local laws, including without limitation the Fair Housing Act and Section 504. The Redevelopment shall include 5% (11 units) mobility accessible units and 2% (5 units) hearing/visual impared units, which shall be constructed in accordance with Section 504 design and construction standard. The accessible units shall be distributed throughout the Redevelopment. Landlord acknowledges that the Development and Market Development, as designed, complies with the requirements of this Section 5.05.

ARTICLE 6

Use, Maintenance, Alterations, Repairs, Etc.

6.01 <u>Condition of Real Estate and Property</u>. Tenant has leased the Real Estate after a full and complete examination thereof, as well as the title thereto and its present uses and restrictions, and, except as expressly provided in the Environmental Agreement and except for Excluded Environmental Conditions (for which responsibility shall be determined under the Environmental Agreement, or if not addressed therein, then under applicable law), Tenant accepts the same "as is" and without any representation or warranty, express or implied, in fact or by law, by Landlord and without recourse to Landlord as to the title thereto, the nature, condition or usability thereof or the use or uses to which the Property or any part thereof may be put; provided,

however, that upon the commencement of this Lease, title to the Real Estate shall be subject only to the Permitted Exceptions. Except as expressly provided in the Environmental Agreement and except for Excluded Environmental Conditions (to the extent that responsibility for such Excluded Environmental Condition is imposed on Landlord under the Environmental Agreement, or if not addressed therein, then under applicable law), Landlord shall not be required to furnish any services or facilities or to make any repairs or alterations in or to the Property, throughout the Term. Other than the obligations of Landlord, if any, under the Environmental Agreement, Tenant hereby assumes the full and sole responsibility for the condition, operation, repair, replacement, maintenance and management of the entire Property; provided, however, that Tenant does not assume responsibility for Excluded Environmental Conditions (except to the extent that responsibility for such Excluded Environmental Conditions is imposed on Tenant, the Developer or the Owner ander the express provisions of the Environmental Agreement, or if not addressed therein, then under applicable law).

- 6.02 <u>Use of Poperty</u>. The Property shall be used and occupied only for multi-family residential uses and for uses incidental thereto and for no other purpose, unless Landlord has consented in writing to such other use, which consent may be granted, withheld or granted with such conditions as Landlord may require, in Landlord's sole and absolute discretion.
- <u>Prohibited Use</u>. Ten at shall not use or occupy the Property or permit the same to be used or occupied, nor do or permit in thing to be done in, on or to the Property, or any part thereof, in a manner that would in any way (a) materially violate any construction permit or certificate of occupancy affecting the Property or any Requirement, (b) make void or voidable any insurance then in force, or make it impossible to obtain fire or other insurance required to be furnished by Tenant hereunder. (c) cause or be apt to cause structural injury to the Property, or any part thereof, or (d) materially violate any material provision of this Lease; provided, however, that the foregoing shall not impose any obligation on Tevant with respect to any Excluded Environmental Condition (except to the extent that responsibility for such Excluded Environmental Condition is imposed on Tenant or the Owner under the express provisions of the Environmental Agreement or, if not addressed therein, then under applicable law) or with respect to any obligation of Landlord under the Environmental Agreement. Notwithstanding anything to the contrary contained in this Lease. Landlord may not terminate this Lease by reason of an act or omission of a resident or sub-tenant of the Development. Tenant shall promptly and diligently take all reasonable steps, in accordance with the provisions of such resident's or sub-tenant's lease and applicable Requirements, to evict such resident or sub-tenant who materially violates any material provision of his or her lease, which violation constitutes a material violation of a material provision of this Lease. Notwithstanding the first sentence of this Section 6.03. Tenant may, in good faith. upon prior written notice to Landlord (and where necessary in the name of, but without expense to, Landlord) and, after having secured Landlord to its reasonable satisfaction against loss or damage, by cash or by a letter of credit or surety bond in an amount, with an issuer or surety, and in form and substance reasonably satisfactory to Landlord, contest the validity of any Requirement and, pending the determination of such contest may postpone compliance therewith, provided that in no event shall such act or omission of Tenant: (i) subject Landlord to any fine or penalty or to prosecution for a crime: (ii) cause the Property, or any part thereof, to be condemned or to be vacated: or (iii) cause any material interference with the operation of the Development for the purposes set forth in Section 6.02 or the occupancy, use, benefit and enjoyment thereof by any resident or sub-tenant of the Development. Tenant shall indemnify, defend and hold harmless

Landlord's Protected Persons from and against any recovery or loss to which any Landlord's Protected Person may be subject or which any Landlord's Protected Person may sustain, including reasonable attorneys' fees and expenses incurred by any Landlord's Protected Person arising from any breach of this covenant or by reason of any action or proceeding which may be brought against any Landlord's Protected Person or against the Property, or any part thereof, by virtue of any Requirement, which do not arise out of any negligent act or willful misconduct of Landlord, or any event of default by Landlord hereunder. Landlord shall provide notice to Tenant of any action brought against Landlord that affects the Property, or any part thereof.

- Maintenance of Property. Subject to Section 8.01, Tenant shall make all necessary repairs to and replacements of the Improvements, interior and exterior, structural and nonstructural, ordinary and extraordinary, foreseen and unforeseen, and shall maintain and keep the Improvements in good and safe order, repair and condition. Tenant covenants and agrees that throughout the Teras (a) all building systems, facilities and equipment, including HVAC systems. common area lighting and the like, shall be maintained in good operating order and repair; and (b) the Real Estate and the Property shall, at all times, have adequate means of ingress and egress to and from the abutting pub ic streets and alleys. Tenant shall indemnify, defend and hold Landlord's Protected Persons harmless from and against any and all claims and demands arising from the failure of Tenant to perform the covenants contained herein or arising from any accident. injury or damage to any person or property that shall or may happen in or upon the Property or the Real Estate, or any part thereof, however caused, other than Landlord's negligent act or willful misconduct or any Excluded Environmental Condition, and shall keep the Real Estate and the Property free and clear of any and all mechanics' liens or other similar liens or charges incidental to work done or material supplied in or about the Property or the Real Estate subject to the provisions in Section 4.03 providing for contest of such liens.
- 6.05 <u>Waste</u>. Tenant shall not do, permit or sufferany waste, damage, disfigurement or injury to or upon the Property, or any part thereof, without repairing the same within a reasonable period of time. Tenant shall have the right at any time and from time to time to sell or dispose of any equipment or fixtures subject to this Lease that may have become absolete or unfit for use or that is no longer useful, necessary or profitable in the conduct of Tenant's business; provided, however, that Tenant shall have substituted or shall promptly substitute for the property so removed from the Property other equipment or fixtures at least of equal quality and utility in the performance of the particular function in question as that of the property so removed was performing an obsolete function and replacement thereof is not necessary or appropriate to maintain the operation or character of the Property, or its use and occupancy by residents of the Property and licensees or its overall value without impairment.
- 6.06 <u>Compliance with Requirements</u>. Except as expressly provided in the Environmental Agreement and except for Excluded Environmental Conditions (for which responsibility shall be determined under the Environmental Agreement, or if not addressed therein, then under applicable law), Tenant shall comply, at its own expense, with all Requirements during the Term and with the reasonable requests of any insurance company having a policy outstanding with respect to the Real Estate or the Property, or any part thereof, whether or not such Requirements or requests require the making of structural alterations or the use or application of portions of the Real Estate or the Property for compliance therewith, or interfere with the use and

enjoyment of the Property, and shall indemnify, defend and hold harmless Landlord's Protected Persons from and against all fines, penalties, and claims for damages of every kind and nature arising out of any failure to comply with any such Requirement or request. It is the intention of the parties that Tenant during the Term shall discharge and perform all obligations of Landlord, as well as all obligations of Tenant arising as aforesaid, and hold harmless Landlord's Protected Persons therefrom (except with respect to Excluded Environmental Conditions to the extent that responsibility for such Excluded Environmental Conditions is imposed on Landlord under the express provisions of the Environmental Agreement or, if not addressed therein, then under applicable law), so that at all times the Rent shall be net to Landlord without deductions or expenses on account of any such Requirement or request, whatever it may be. Tenant may, in good faith upon prior written notice to Landlord (and wherever necessary, in the name of, but without expense to, Landlord) and, after having secured Landlord to its reasonable satisfaction against loss or demage, by cash or by a letter of credit or surety bond in an amount, with an issuer or surety, and in 10 m and substance reasonably satisfactory to Landlord, contest the validity of any such Requirement of request and, pending the determination of such contest, may postpone compliance therewith, provided that in no event shall such contest or postponement: (i) subject Landlord to any fine or penalty or to prosecution for a crime; (ii) cause the Real Estate, Leasehold Estate or the Property, or any part thereof, to be condemned or to be vacated; or (iii) cause any material interference with the operation of the Property for the purposes set forth in Section 6.02 or the occupancy, use, benefit and enjoyment thereof by any resident or sub-tenant of the Development. Notwithstanding anything to the contrary in this Section 6.06, except as provided for in the Environmental Agreement, Tenant has no compliance responsibility or liability for matters existing prior to the commencement of the Term.

Exculpation of Landlord. Landlord shall not be responsible or liable for any 6.07 destruction, damage or injury to any property or to any person or persons at any time on the Property resulting from any casualty, occurrence or condition occurring or existing during the Term of this Lease (except as expressly set forth in the Environ nental Agreement), including without limitation those from steam, gas or electricity, or from waer, rain or snow, whether the same may leak into, issue or flow from or within any part of the Property or from the pipes or plumbing of the same, or from any other place or quarter; nor shall Landlord be in any way responsible or liable in case of any accident or injury (including death) to any of Tenant's residents. subtenants, employees or agents, or to any person or persons in or about the Property or the streets. driveways, sidewalks, parkways or alleys adjacent thereto; and Tenant agrees that it will not hold Landlord in any way responsible or liable therefor (except for actions caused by Landlord's negligent act or willful misconduct) and will further indemnify, defend and hold Landlord's Protected Persons harmless from and against any and all claims, liability, penalties, damages, expenses and judgments arising from injury to persons or property of any nature and also for any matter or thing arising out of or resulting as a direct or indirect consequence from the use or occupancy of the Property: provided, however, that the foregoing obligations shall not apply to Excluded Environmental Conditions. Nothing contained or required in this Section 6.07 shall be deemed to release Landlord from providing contracted services to Tenant and the Development, or from the consequences to Tenant and the Tenant Property of Landlord's negligent act, willful misconduct or breach of its representations, warranties and covenants set forth herein or in the Environmental Agreement or for liability arising from Excluded Environmental Conditions (for which responsibility shall be determined under the Environmental Agreement, or if not addressed therein, then under applicable law).

- 6.08 Exculpation of Leasehold Mortgagee. Until any Leasehold Mortgagee becomes a mortgagee in possession or the tenant under a new lease, no Leasehold Mortgagee shall, except to the extent of the gross negligence or willful misconduct of such Leasehold Mortgagee, its agents and employees on or about the Real Estate be responsible or liable for any destruction, damage or injury to any property or to any person or persons at any time on the Real Estate resulting from any casualty, occurrence or condition occurring or existing during the Term of this Lease, including without limitation those from steam, gas or electricity, or from water, rain or snow, whether the same may leak into, issue or flow from or within any part of the Real Estate or from the pipes or plumbing of the same, or from any other place or quarter; nor shall any Leasehold Mortgagee be in any way responsible or liable in case of any accident or injury (including death) to any of Tenant's residents, subtenants, employees or agents, or to any person or persons in or about the Real Estate or the streets, driveways, sidewalks, parkways or alleys adjacent thereto.
- 6.09 Lanciord's Right of Entry. Landlord shall have the right, upon reasonable advance notice to Terant and subtenants, when appropriate, on any business day, to enter upon the Property or the Real Estate, or any part thereof, for the purpose of ascertaining the condition thereof, or whether Tenant is observing and performing the obligations assumed by it under this Lease, or to make any repairs or perform any work, all without hindrance or molestation from Tenant, or anyone claiming by, through or under Tenant, whether as subtenant or otherwise. Each Leasehold Mortgagee, to the extent permitted in its Loan Documents, at its sole cost and expense with no liability or expense to Landlord upon reasonable advance notice to Landlord. Tenant and subtenants, when appropriate, on any business day, may enter upon the Property or the Real Estate. or any part thereof, for the purpose of ascert ining the condition thereof, or whether Tenant is observing and performing the obligations assume(1b) it under this Lease and its Loan Documents. or to make any repairs or perform any work, all without hindrance or molestation from Tenant, or anyone claiming by, through or under Tenant, whether as subtenant or otherwise. No Leasehold Mortgagee shall interfere with Landlord's exercise of its rights under this Lease or disturb or interfere with the occupancy of any subtenant. The above ment oned rights of entry shall be exercisable at reasonable times, at reasonable hours and on reasonable advance notice; provided, however, that entry may be made at any time without notice in the event of an emergency (although Landlord shall endeavor to give Tenant prior notice thereof). Nothing contained herein, however, shall impose or imply any duty on the part of Landlord to make any such repairs or perform any such work.
- 6.10 No Liens. Notice is hereby given that Landlord shall not be liable for any labor or materials furnished or to be furnished to Tenant upon credit, and that no mechanics or other lien for any such labor or material shall attach to or affect the estate or interest of Landlord in and to the Property or any part thereof.

6.11 <u>Compliance with HUD Requirements</u>.

- (a) The Public Housing Units shall be subject to and operated in accordance with all Applicable Public Housing Requirements.
- (b) The PBV Units shall be subject to and operated in accordance with the PBV Requirements.

ARTICLE 7

Insurance

- 7.01 <u>Maintenance of Insurance</u>. During the Term, Tenant shall, at its sole expense, obtain and maintain, or cause to be obtained and maintained policies of insurance satisfying the requirements set forth on Exhibit F.
- 7.02 Form of Policies. Except as provided in Section 8.02, any policies of insurance covering the Development during construction, shall expressly provide that any losses thereunder shall be adjusted with Tenant and all Leasehold Mortgagees as their interests may appear (or, absent a Leasehold Mortgagee, with Landlord). All such insurance shall be carried in the name of Tenant and lose discretive interests may appear. All such insurance may be in the form of a so-called "blanket policy" covering more than two properties, provided that the amount of coverage shall be not less than the aggregate of the Full Insurable Values of all covered properties and the policy shall include an "agrecial amount" endorsement on a no-coinsurance basis.
- 7.03 Evidence of Insurance and Payment. Upon the execution and delivery of this Lease, and thereafter not later than fineen (15) days prior to the expiration date of an expiring policy theretofore furnished pursuant to this Article, certificates of insurance evidencing the required coverages, bearing notations evidencing the payment of premiums or accompanied by other evidence satisfactory to Landlord of such payment, shall be delivered by Tenant to Landlord. Upon request from Landlord, Tenant shall deliver to Landlord duplicate originals or certified copies of the policies required by this Article 7.
- 7.04 Separate Insurance. Tenant shall not obtain separate insurance concurrent in form or contributing in the event of loss with that required in this Article to be furnished by or that may reasonably be required to be furnished by Tenant unless Landlord is included therein as an additional insured, with loss payable as required in this Lease. Tenant shall immediately notify Landlord of the obtaining of any such separate insurance and shall deliver duplicate originals or certified copies of the policy or policies so obtained as provided in Section 7.93.
- 7.05 <u>Cancellation</u>. Each policy of insurance delivered hereunde: shall contain an agreement by the insurer that such policy shall not be cancelled or materially altered without at least thirty (30) days' prior written notice given to Landlord and to each Leasehold Mortgagee named in such policy.

ARTICLE 8

Damage and Restoration

8.01 Damage or Destruction.

(a) In the event of any damage to or destruction of the Improvements during the Term, Tenant shall give Landlord immediate notice thereof and, unless the insurance proceeds are applied by a Leasehold Mortgagee to reduce its debt in accordance with Section 8.05, Tenant shall promptly and diligently restore, replace, rebuild and repair the same as nearly as possible to

their value, condition and character immediately prior to such damage or destruction, in accordance with the following provisions of this Article 8. Landlord shall in no event be called upon to restore, replace, rebuild or repair the Property, or any portion thereof, or to pay any of the costs or expenses thereof. All work in connection with such restoration, replacement, rebuilding and repairing, including all temporary repairs to the Property or repairs made for the protection of the Property pending the completion of the permanent restoration, replacement, rebuilding and repairing, is hereinafter collectively referred to as "Full Restoration." In the event of any damage to or destruction of the Improvements occurring during the Term, Tenant shall, upon demand, deposit with the First Leasehold Mortgagee (or, if none, with Landlord), or into the Restoration Escrow (as that term is hereinafter defined), the amount of any applicable deductible or selfinsurance (the "Deductible"). If the Net Insurance Proceeds (as that term is hereinafter defined) available for a particular Restoration (as that term is hereinafter defined) plus the amount of the Deductible, are insufficient to accomplish the Full Restoration, then Landlord may terminate this Lease with respect to the portion of Real Estate on which such damaged or destroyed Improvements were situated (and any related open space parcels) by written notice to Tenant and all Leasehold Mortgagers, unless: (1) within ninety (90) days after the amount of Net Insurance Proceeds has been determined Tenant deposits with the First Leasehold Mortgagee (or. if none. with Landlord) cash, a letter of credit and/or evidence satisfactory to the First Leasehold Mortgagee (or. if none, to Landlord) of the availability of funds (from a loan or otherwise) in an amount equal to the Restoration Deficiency (as that term is hereinafter defined); or (2) within ninety (90) days after the expiration of said 90-day period, any one or more of the Leasehold Mortgagees and/or any other Person so deposits the Restoration Deficiency with the First Leasehold Mortgagee (or, if none, with Land ord) or into the Restoration Escrow; or (3) within one hundred eighty (180) days after the amount of Net Insurance Proceeds has been determined, the First Leasehold Mortgagee agrees to a Partial Restocation and agrees to make the Net Insurance Proceeds available for such Restoration; or (4) at least one Leasehold Mortgagee is diligently proceeding to obtain such insurance proceeds and, if applicable, to exercise its rights with respect to the Restoration; provided, however, that the foregoing proh bition against termination shall no longer be applicable when the First Leasehold Mortgagee elects to caply such insurance proceeds to repay outstanding debt in lieu of Restoration. The First Leasehold Mongagee shall consult with all subordinate Leasehold Mortgagees with respect to the application of the Net Insurance Proceeds: provided however that in the event of any disagreement between the First Leasehold Mortgagee and any subordinate Leasehold Mortgagee over the application of the Net Insurance Proceeds, the decision of the First Leasehold Mortgagee, in its sole discretion, shart provail, subject to Section 8.01(b). The Full Restoration or Partial Restoration, as applicable, is hereinafter referred to as the "Restoration". As used herein, the term "Restoration Deficiency" skall mean additional funds in an amount sufficient, when added to the Net Insurance Proceeds available for a Restoration plus the Deductible, to complete such Restoration. If this Lease is terminated pursuant to this Section 8.01(a), with respect to a portion of the Real Estate only then, at the option of Landlord, Tenant shall, at Tenant's sole expense, demolish and/or remove such of the Improvements on such portion of the Real Estate as are designated by Landlord, provided that, if the costs of such demolition and removal exceed the Deductible, sufficient Net Insurance Proceeds are made available to Tenant for that purpose.

(b) The determination of whether the Proceeds are sufficient to rebuild and repair the Improvements so damaged or destroyed to a value, condition and character that is not materially different from what existed immediately prior to such damage or destruction (i.e. a Full

Restoration) and that such Restoration is feasible, shall be reasonably made by the First Leasehold Mortgagee in accordance with the requirements of its Leasehold Mortgage (or, if none, by The First Leasehold Mortgagee shall consult with all subordinate Leasehold Mortgagees with respect to application of Net Insurance Proceeds; provided however that in the event of any disagreement between the First Leasehold Mortgagee and any subordinate Leasehold Mortgagee over the application of Net Insurance Proceeds, the decision of the First Leasehold Mortgagee, in its sole discretion, shall prevail. If there is to be a Restoration, all Proceeds shall be deposited in an account with First Leasehold Mortgagee or, if there is no First Leasehold Mortgagee, in a construction disbursement escrow, among Landlord, Tenant, the Leasehold Mortgagees, if any, and a mutually acceptable title company (the "Restoration Escrow"), and disbursed to pay the costs of such Restoration. By accepting a Leasehold Mortgage, each Leasehold Mortgagee agrees to be bound by such determination of the First Leasehold Mortgagee and to make the Net Insurance Proceeds available for such Restoration. In the event of any such damage to or destruction of the Improvements, Tenant, Landlord or any Leasehold Mortgagee. shall have the right (ov. shall not be obligated) to deposit the Restoration Deficiency into the Restoration Escrow. Usless the Restoration Deficiency is deposited with the First Leasehold Mortgagee (or, if none, with Landlord) or into the Restoration Escrow, or the First Leasehold Mortgagee agrees to a Partial Restoration, within the time periods provided in Section 8.01(a), this Lease shall terminate as to the affected portions of the Real Estate and the provisions of Section 8.05 shall apply. Subject to the rights of each Leasehold Mortgagee, Applicable Public Housing Requirements and the PBV Requirements, any insurance proceeds remaining after completion of the Restoration shall be paid to Tenant.

- 8.02 Adjustment of Insurance Claims and Disbursements. Adjustment of any insurance claim shall, subject to the terms of any Leasehold Mortgage, be negotiated by Tenant. All insurance proceeds shall be deposited with the First Leasehold Mortgagee (or, if none, with Landlord) or into a Restoration Escrow, and administered is hereinafter set forth. All insurance proceeds received by Landlord or Tenant on account of such da nage or destruction, less the actual costs, expenses and fees, if any, incurred in connection with the adjustment of the loss (the "Net Insurance Proceeds"), shall be applied in accordance with the terms of this Article. Such Net Insurance Proceeds plus the Deductible shall be paid out from time to time as such Restoration progresses and is approved. All Proceeds held by the First Leasehold Mortgagee shall be held in trust in a separate bank account.
- 8.03 <u>Deficiencies</u>. If, at any time during the course of a Restoration, the projected Restoration Deficiency increases, Tenant shall either, before proceeding with the Pesteration, deposit with the First Leasehold Mortgagee (or, if none, with Landlord) cash, a letter of credit and/or evidence satisfactory to Landlord of the availability of funds (from a loan or otherwise) in an amount equal to the increase in the Restoration Deficiency, or deliver to the First Leasehold Mortgagee (or, if none, to Landlord) a surety bond from a company and in form and substance satisfactory to the First Leasehold Mortgagee (or, if none, to Landlord), for such increase in the Restoration Deficiency, the premium for which shall have been paid by Tenant. Thereupon, Tenant may proceed with the Restoration.
- **8.04** Landlord's Right to Complete. Subject to any rights of the First Leasehold Mortgagee under Section 8.01, if a Restoration is commenced or required to be commenced, and if Tenant shall fail to promptly and diligently commence and complete such Restoration, Landlord,

after first giving all Leasehold Mortgagees written notice and at least sixty (60) days thereafter to commence such Restoration and thereafter promptly and diligently complete such Restoration, may complete the same and apply the Net Insurance Proceeds plus the Deductible and any additional funds provided by Tenant to the cost of Restoration.

Leasehold Mortgages. Except as provided in Section 8.01: (1) all provisions of this Article 8 are subject to the rights of the First Leasehold Mortgagee and the provisions of the First Leasehold Mortgage; (2) the provisions of the First Leasehold Mortgage shall govern in the event of any conflict or inconsistency between the provisions of this Article 8 and the provisions of the First Leasehold Mortgage or any other documents governing the loan secured by the First Leasehola Mortgage; and (3) application of Net Insurance Proceeds shall be subject to the terms of the Leasehold Mortgages, and the respective priorities of the Leasehold Mortgages thereunder, including the Leasehold Mortgagees' rights, if any, to apply proceeds of insurance to the payment of outstanding debt oved by Tenant to such Leasehold Mortgagees in lieu of Restoration. In such an event, Landlord and Tenant shall adjust any remaining balance of insurance proceeds as their respective interests may be affected by such damage or destruction, and this Lease shall terminate as to the affected portions of the Real Estate. No termination of this Lease shall occur under this Article 8 so long as at least one Leasehold Mortgagee is diligently proceeding to obtain such insurance proceeds and, if appl cable, to exercise its rights with respect to the Restoration: provided, however, that the foregoing prohibition against termination shall no longer be applicable when the First Leasehold Mortgagee elects to apply such insurance proceeds to repay outstanding debt in lieu of Restoration.

ARTICLE 9

Title and Ownership; Leasebold Mortgage

9.01 Restrictions on Transfer.

- (a) Except for Permitted Transfers and Permitted Refinancings: (i) Tenant shall not at any time without the prior written consent of Landlord: (A) sell, assign, transfer, or convey all or any part of its interest under this Lease, or (B) sell, assign, transfer or convey all or any part of any structure or other Improvement located on the Real Estate; or (C) sublet all or any part of the Tenant Property except for subletting of the dwelling units to sublement, pursuant to subleases with a term (including options to extend or renew) not in excess of one (1) year, and (ii) there shall not be a change of control of Tenant or the sale, assignment, transfer or conveyance of any interest in Tenant, other than to First Leasehold Mortgagee. Landlord's consent to any of the foregoing may be granted, withheld or granted with such conditions as Landlord shall require, in its sole and absolute discretion. Notwithstanding any provisions to the contrary in this Lease, any sale, assignment, transfer or conveyance of any portion of the Development, including both Real Estate and Improvements thereon, must involve the sale, assignment, transfer or conveyance of the entire Development.
- (b) If a Permitted Transfer consisting of a sale, assignment, transfer or other conveyance of the Leasehold Estate occurs, other than a transfer by foreclosures or deed in lieu of foreclosure to a Leasehold Mortgagee, the transferee or assignce shall enter into an assumption agreement with Landlord by which it assumes all of Tenant's rights and obligations under this

Lease. Upon the consummation of such Permitted Transfer and the delivery to Landlord of such assumption agreement executed by the transferee or assignee, the transferee or assignee shall succeed to all rights and obligations of Tenant under this Lease, and shall be deemed a permitted assignee of Tenant, and Tenant making such sale, assignment, transfer or other conveyance shall be and hereby is relieved of any continuing obligations hereunder arising thereafter and such permitted assignee, by accepting such assignment, shall be deemed to have assumed all obligations hereunder arising after such assignment. Landlord acknowledges that Tenant may cause the Improvements, or a portion thereof, to qualify for housing "low income families", "very low income families", and/or "extremely low income families" for the period required under Section 42 of the Internal Revenue Code of 1986, as amended (the "Code"). Tenant may cause the Improvements, or a portion thereof, to qualify for other state and/or federal assistance, including but not limited to financing arranged through or insured by HUD. To the extent required by the Code and/or Fix as a condition to such qualification, and as may be required under Article 15. Tenant is authorize ato enter into restrictive covenants encumbering the Tenant Property pertaining to the use of the Tenant Property. Landlord agrees to enter into a subordination agreement relating to this Lease as to such restrictive covenants as may be required to obtain and maintain such qualifications.

- (c) Landlord shall not, without the prior written consent of Tenant and all Leasehold Mortgagees, mortgage or create a lien upon (i) all or any part of the Real Estate, or (ii) all or any part of its interest in this Lease or any Improvement.
- (d) The parties acknowledge that it may become necessary to grant easements and/or licenses over, under, upon and across the Real Estate for the provision of gas, electricity, telephone service, cable television, Internet access, we're, sewer, and other utilities, and to allow pedestrian and vehicular access, if necessary, to serve the laprovements. All such easements and licenses shall be subject to the prior written consent of Landlord, which shall not be unreasonably withheld or delayed. If required, Landlord shall grant or joir with Tenant in the grant of such easements and licenses, so as to subject Landlord's interest in the Real Estate to such easements and licenses. All costs in connection with such easements and licenses shall be borne by Tenant.
- Liens. Other than the Permitted Exceptions and the encumbrances listed on Exhibit D (the "Encumbrances"), and any Permitted Refinancing, Tenant shall not create or permit to be created or to remain, and shall promptly discharge, any lien (including but not limited to any mechanic's, contractor's, subcontractor's or materialman's lien or any lien, encumbrance or charge arising out of any Imposition, conditional sale, title retention agreement, chattel mortgage, security agreement, financing statement or otherwise) upon the Real Estate or the Tenant Property, or any part thereof, or the income therefrom, and Tenant shall not suffer any matter or thing whereby the estate, rights and interests of Landlord in the Real Estate or the Development, or any part thereof, will be impaired. Notwithstanding the foregoing prohibitions, Tenant shall have the right to contest any such lien upon compliance with the same conditions as are applicable to the contest of any Imposition under Section 4.03 or to provide title insurance over any such lien in a manner reasonably satisfactory to Landlord. If Tenant shall fail to cause any such matter to be discharged of record or contested in the foregoing manner, then Landlord may, but shall not be obligated to, in addition to any other right or remedy, discharge such lien at any time after delivery of notice to Tenant, either by paying the amount claimed to be due or by procuring the discharge of such lien by deposit or bonding proceedings or otherwise, and in any such event Landlord shall be entitled,

if it so elects, to compel the prosecution of an action for foreclosure of such lien by the lienholder and to pay the amount of judgment in favor of the lienholder with interest, costs and allowances. Any amount so paid by Landlord and all reasonable costs, expenses and fees incurred by Landlord in connection therewith shall be reimbursed by Tenant to Landlord. This Lease shall constitute notice that Landlord shall not be liable for any work performed or to be performed, or any materials furnished or to be furnished, at the Real Estate for Tenant or any subtenant upon credit, and that no mechanic's or other lien for such work or materials shall attach to or affect the estate or interest of Landlord in and to the Real Estate, unless such work or materials is specifically ordered by Landlord in writing. Notwithstanding anything to the contrary contained in this Section 9.02, the obligations of Tenant under this Section 9.02 shall not apply to any lien arising from an Excluded Environmental Condition.

9.03 **Leasehold Mortgage**.

in order to enable Tenant to finance a portion of the cost of construction of the Development, Tenant shall have the right, at or prior to commencement of construction or rehabilitation of the Improvements required under Section 5.01, to mortgage its Leasehold Estate, together with its ownership interest in the Improvements, and execute and record a Leasehold Mortgage or Mortgages with respect to both such estates, respectively (collectively, the "Initial Leasehold Mortgages"), to secure the repayment of a loan or loans made to Tenant by a Lender or Lenders (collectively, the "Initial Levsehold Mortgagees") in an aggregate amount not to exceed the estimated cost of the Development, or such other amount as is reasonably approved by Landlord. The Initial Leasehold Mortgagees and their addresses for purposes of notices are listed on Exhibit C. In addition to the Initial Leasehold Mortgages, Tenant shall have the right, provided that Tenant first obtains all consents and/or approvais required under this Lease, at any time and from time to time, to grant one or more Leasehold Moragages. Landlord's interest in the Real Estate or this Lease shall at no time be encumbered by and shall at no time be subject or subordinate to any Leasehold Mortgage (i.e. the foreclosure of any such Leasehold Mortgage shall not divest Landlord of its fee simple title or reversionary interest), except as to rights expressly granted to any Leasehold Mortgagee in this Lease. For purposes of this Article 3, the making of a Leasehold Mortgage shall not be deemed to constitute an assignment or transfer of this Lease or the Leasehold Estate, nor shall any Leasehold Mortgagee, as such, be deemed to be an assignee or transferee of this Lease or of the Leasehold Estate so as to require such Leasehold Mortgagee, as such, to assume the performance of any of the terms, covenants or conditions on the part of Tenant to be performed hereunder. A Leasehold Mortgagee may become the holder of the Leasehold Estate and succeed to Tenant's interest under this Lease by foreclosure of its Leasehold Mortgage (either in its own name or in the name of its nominee) or as a result of the assignment of Tenant's interest under this Lease in lieu of foreclosure, and any purchaser at any sale of Tenant's interest under this Lease in any proceeding for the foreclosure of any Leasehold Mortgage or the assignee or transferee of Tenant's interest under this Lease under any instrument of assignment or transfer in lieu of the foreclosure of any Leasehold Mortgage shall be deemed to be an assignce or transferee approved by Landlord and shall be deemed to have agreed to perform all of the terms, covenants and conditions on the part of Tenant to be performed hereunder arising from and after the date of such transfer and if the remediation work required in the Environmental Agreement, has not been fully performed, all of the terms, covenants and conditions to be performed under the Environmental Agreement from and after the date of such assignment, but only for so long as such purchaser, assignee or transferee is the owner of the Leasehold Estate. The security interests of Leasehold

Mortgagees, as mortgagees, shall be transferable to holders of certificates or other successors or assigns in connection with the securitization, sale or assignment of a Leasehold Mortgagee's Leasehold Mortgage Loan.

- (b) In addition to the Initial Leasehold Mortgagees identified on Exhibit C, if Landlord shall be notified in writing of the existence of any other Leasehold Mortgage, and provided that the Leasehold Mortgagee shall have designated in a written notice to Landlord the address of the Leasehold Mortgagee for the service of notices, then notice of any Event of Default by Tenant in the performance of the covenants of this Lease shall simultaneously be given to such Leasehold Mortgagee, and such Leasehold Mortgagee shall have the right, but not the obligation, within the respective periods as prescribed in Subsection 9.03(c), to take such action or to make such payments as may be necessary to cure any such default to the same extent and with the same effect as though done by Tenant. Landlord agrees to accept such cure when proffered.
- if there shall be an Event of Default by Tenant under this Lease, Landlord agrees that it will not enforce any rights it may have to terminate this Lease or invoke its right to take possession of the Tenam Property if: (i) any Leasehold Mortgagee shall cure the default within 180 days after expiration of the time for Tenant to cure said default, or if such default cannot reasonably be cured within said 180-day period, and any Leasehold Mortgagee in good faith commences within said 180-day period and thereafter diligently prosecutes all actions required to cure such default, such longer period as my be reasonably necessary; or (ii) within 180 days after notice of such default by Landlord to a Leasehold Mortgagee, such Leasehold Mortgagee commences legal proceedings (herein called "foreclosure proceedings") to foreclose the lien of its Leasehold Mortgage and if such Leasehold Mortgagee diligently proceeds with its foreclosure proceedings or obtains a deed in lieu of foreclosure (including seeking to be put in possession as mortgagee-in-possession or seeking to obtain the appointment of a receiver in such foreclosure proceedings). If such Event of Default is a non-monetary default which: (x) is personal to Tenant or by its nature cannot be cured by a Leasehold Mortgagee, or (y) requires possession of the Leasehold Estate and/or Improvements to cure, the foregoing 180-day periods provided by clause (ii) shall be extended for so long as such Leasehold Mortgagee is enjoined or stayed in any bankruptcy or insolvency proceedings filed by or against Tenant, provided that a Leasehold Mortgagee timely commences and diligently prosecutes a foreclosure proceeding. Regarding defaults, the nature of which are referred to in clause (x) of the immediately p ecoding sentence. the Leasehold Mortgagee's acquisition of the Leasehold Estate and Improvements pursuant to foreclosure, deed in lieu of foreclosure or otherwise shall be deemed a cure of such defaults. Regarding defaults, the nature of which are referred to in clause (y) in the immediately preceding sentence, after the Leasehold Mortgagee acquires the Leasehold Estate and Improvements, Leasehold Mortgagees shall have a reasonable period of time to cure such defaults, provided Leasehold Mortgagee diligently pursues such cure.
- (d) Nothing in this Article 9 shall require any Leasehold Mortgagee, as a condition to the exercise of rights provided under this Article 9, to cure any Event of Default of Tenant. The foregoing shall not be deemed to excuse a Leasehold Mortgagee from performing covenants relating to the construction (provided that no Leasehold Mortgagee has elected to discontinue or reduce the funding of its loan) or condition of Improvements on the Real Estate or other similar matters requiring access to and/or control of the Property from and after such time as such Leasehold Mortgagee acquires the Leasehold Estate by foreclosure or otherwise, provided

that the Construction Completion Deadline shall be extended accordingly during any such period given to initiate and complete any foreclosure proceeding. If no Leasehold Mortgagee commences and prosecutes either curative action or foreclosure proceedings as provided above, Landlord may invoke any or all of its remedies under this Lease, including the remedy of termination. In the event the purchaser at the foreclosure sale or the assignee of such purchaser or the recipient of any deed in lieu of foreclosure acquires the Leasehold Estate and Tenant's interest in the Improvements, such purchaser or assignee shall thereupon become Tenant under this Lease and hereby agrees to assume and perform each and all of Tenant's obligations and covenants hereunder from and after the date that such purchaser or assignee acquires the Leasehold Estate and Tenant's interest in the Improvements.

- In the event there is a Leasehold Mortgage listed on Exhibit C or a Leasehold Morigage of which Landlord has received notice as provided in Subsection 9.03(b), Landlord agrees in Cit will not accept a surrender of the Tenant Property or a cancellation of this Lease from Tenant prior to the expiration of the Term of this Lease and will not amend this Lease without in each case notifying each such Leasehold Mortgagee of Tenant's offer to surrender the Tenant Property. Such Leasehold Mortgagee may become the holder of the Leasehold Estate and succeed to Tenant's interest under this Lease (either in its own name or in the name of its nominee). shall be deemed to be an assignee or transferee approved by Landlord and shall be deemed to have agreed to perform all of the terms, coverants and conditions on the part of Tenant to be performed hereunder from and after the date that such Leasehold Mortgagee acquires the Leasehold Estate and Tenant's interest in the Improvements, so long as all monetary defaults of Tenant are cured, and if the remediation work required in the Environmental Agreement, has not been fully performed, all of the terms, covenants and conditions to be performed under the Environmental Agreement from and after the date of such assignment subject in each case to the performance by the Landlord of all of the terms, covenants and conditions on the part of Landlord to be performed hereunder and thereunder, but only for so long as such Leasthold Mortgagee is the owner of the Leasehold Estate. If no Leasehold Mortgagee becomes the 'iolder of the Leasehold Estate as provided above, Landlord may accept surrender of the Tenant Property and cancel this Lease.
- (f) Each Leasehold Mortgagee, by accepting its Leasehold Mortgage, agrees for the benefit of Landlord:
 - (i) that such Leasehold Mortgagee will use reasonable of facts to give to Landlord notice of all events of default declared by such Leasehold Mortgagee with respect to its Leasehold Mortgage Loan that give such Leasehold Mortgagee the right of acceleration, concurrently with or promptly after notice thereof is given to Tenant; and Landlord shall have the right, but shall not be obligated, to cure any such defaults on the part of Tenant within the time period, if any, allowed by the Leasehold Mortgage:
 - (ii) prior to commencing foreclosure proceedings or accepting a deed in lieu of foreclosure, such Leasehold Mortgagee shall give Landlord a written notice describing the action proposed to be taken by such Leasehold Mortgagee and stating the aggregate amount of the indebtedness then due and secured by the Leasehold Mortgage, and setting forth in reasonable detail the respective portions of said indebtedness attributable to principal, interest, attorneys' fees and expenses

and other costs, fees and expenses. Landlord shall have a period of sixty (60) days following receipt of such notice from such Leasehold Mortgagee during which period Landlord, at its election, shall give written notice to Leasehold Mortgagee of Landlord's intent to purchase from such Leasehold Mortgagee, without representation, warranty or recourse (other than as to the purchase price), the Leasehold Mortgage, the indebtedness secured thereby, and any other security held by such Leasehold Mortgagee for such indebtedness, for a purchase price equal to the amounts due such Leasehold Mortgagee under the Leasehold Mortgage documents, including without limitation, any actual costs and expenses of Leasehold Mortgagee and any prepayment premiums. Landlord shall have an additional sixty (60) days after notifying Leasehold Mortgagee of its intent to purchase to complete the transaction; and

- (g) So long as any Leasehold Mortgage is in existence, unless all Leasehold Mortgagees shall otherwise expressly consent in writing, the fee title to the Real Estate and the Leasehold Estate shall not merge, but shall remain separate and distinct, notwithstanding the acquisition of said fee title and Leasehold Estate by any single owner, other than by termination of this Lease by Landlord in compliance with the provisions of this Article 9.
- If Landlord o Tenant shall terminate this Lease, or if this Lease shall be (h) terminated by reason of the rejection of his Lease by a debtor in possession or a trustee or receiver appointed by a court of competent jurisdiction in bankruptcy or insolvency proceedings involving Tenant, then and in either such event, Landlor shall immediately seek to obtain possession of the Real Estate and title to the Improvements. Upo acquiring such possession and title, Landlord shall notify all Leasehold Mortgagees. Each of the Leasehold Mortgagees (in the order of priority of their respective Leasehold Mortgages) or a nominee of a Leasehold Mortgagee designated by such Leasehold Mortgagee by written notice to Landlord, shall have one hundred twenty (120) days from the date of such notice of acquisition to elect to take a new lease on the Real Estate and a conveyance of title to the Improvements. Landlord shall, subject to applicable bankruptcy laws and/or the order of a court of competent jurisdiction, enter into a new lease with a Leasehold Mortgagee (or with the nominee of such Leasehold Mortgagee), convey ide to the Improvements to such Leasehold Mortgagee (or such nominee) by quitclaim deed, and assign to such Leasehold Mortgagee (or such nominee) all leases where the residents or sub-tenants thereunde have attorned to Landlord provided that:
 - (i) such Leasehold Mortgagee has made written request of Landlord for a new lease of the Real Estate and a conveyance of the Improvements within the one hundred twenty (120) days next following the date of termination of this Lease aforesaid and either the prior written consent of any superior Leasehold Mortgagee is obtained or the one hundred twenty (120) days have expired without such superior Leasehold Mortgagee requesting a new ground lease and conveyance of Improvements hereunder; and
 - (ii) at the time of termination of this Lease, and at the time of such Leasehold Mortgagee's written request for a new lease and deed, and at the time of execution and delivery of such new lease by and between Landlord and such Leasehold Mortgagee (or the nominee of such Leasehold Mortgagee, as the case

may be), such Leasehold Mortgagee (or such nominee) shall have cured all defaults of Tenant under this Lease that can reasonably be cured by such Leasehold Mortgagee or shall be proceeding in accordance with Section 9.03(e): and

- (iii) at the time of execution and delivery of such new lease by and between Landlord and such Leasehold Mortgagee (or the nominee of such Leasehold Mortgagee, as the case may be), if the remediation work required in the Environmental Agreement has not been fully performed, such Leasehold Mortgagee (or the nominee of such Leasehold Mortgagee) shall enter into a new environmental agreement with Landlord in the form of the Environmental Agreement, pursuant to which such Leasehold Mortgagee (or the nominee of such Leasehold Mortgagee) in replacement of Tenant and Developer (as defined in the Environmental Agreement), shall be obligated to complete the obligations of Tenant and the Developer under the Environmental Agreement.
- ground lease, concurrently with the recordation thereof, such Leasehold Mortgagee shall: (A) cause the tenant's leasehold estate under the new ground lease and the fee simple title to the Improvements (collectively, the "New Lease Property") to be made subject to the field of all Leasehold Mortgages (and related documents) that encumbered the Property immediately prior to the termination of this Lease (the "Existing Documents"); and/or (B) execute or cause to be executed, and recorded new leasehold mortgages (and related documents), with the same terms as the Existing Documents, encumbering the New Leasehold Property; and (C) take all other actions necessary to place such Leasehold Mortgagees in the same position under the new ground lease as immediately prior to the termination of this Lease.

Such new lease shall have a term equal to the unexpired portion of the Term of this Lease, shall be deemed to be part of an uninterrupted leasehold estate, and shall, except as otherwise provided herein, be on the same terms and conditions as contained in this Lease. Landlord shall deliver possession of the Real Estate and Improvements immediately upon execution of the new lease. Upon executing a new lease, the Leasehold Mortgagee (or the nominee of such Leasehold Mortgagee, as the case may be) shall pay to Landlord the amount by which (a) the sum of any unpaid Rent due under this Lease (or which would have been due under this Lease if it had not been terminated), if any, from the date that Landlord obtains possession of the Real Estate and Improvements to the commencement date of the new lease, plus any Impositions that were liens on the Real Estate ana/or the Improvements and which were paid by Landlord, exceeds (b) any rent or other income received by Landlord from the Real Estate and/or the Improvements during the period after Landlord obtains possession to (but not including) the commencement date of the new lease. A Leasehold Mortgagee shall not have the right to elect to take a new lease if its Leasehold Mortgage is paid in full prior to the Leasehold Mortgagee notifying Landlord of its election to take a new lease. Concurrently with the execution, delivery and recording of the new lease, this Lease shall be deemed to have terminated.

(i) Notwithstanding any provisions to the contrary in Sections 6.03, 6.04, 6.06, 9.01, 9.02, 9.03, 11.02 or 20.04. Landlord agrees (without waiving any rights that Landlord may

have against any former Tenant) that: (a) any Leasehold Mortgage or purchaser at a foreclosure sale or assignce or transferce in lieu of foreclosure of a Leasehold Mortgage that succeeds to the Leasehold Estate and becomes a successor Tenant hereunder (a "Successor") shall not be responsible for any then existing indemnification obligation of the former Tenant; (b) such Successor shall not be required to cure a default that cannot be cured by the payment of money or the taking of affirmative action (an "Incurable Default"); and (c) failure by such Successor to cure an Incurable Default or to assume such existing indemnification obligations of the former Tenant shall not constitute a basis for not recognizing such Successor as the successor Tenant or for terminating this Lease.

9.04 Severability. At the request of Tenant, Landlord may in its discretion and with the consent of LOD and the Leasehold Mortgagees permit the release and transfer of a portion of all or a portion of the Development to enable a separate mortgage financing, in the event that Landlord, HUD, in restor and the Permitted Leasehold Mortgagees determine that such portion of the Development is underperforming in comparison to the other properties in the Redevelopment and, as a result, make continued operations of the entire Redevelopment impractical or unduly burdensome, (the "Release 'ke prirements"). The conditions for severability will be provided by Landlord, HUD and the Leasehold Mortgagees at the time the Leasehold Mortgagees determine that the Release Requirements have been met.

ARTICLE 10

Tenant Default: Right, and Remedies of Landlord

- 10.01 <u>Tenant's Event of Default</u>. Each of the following events shall be an "Event of Default" by Tenant under this Lease:
- (a) Tenant's failure to pay, when due, any installment of Rent or any other amount to be paid by Tenant under this Lease, and such failure shall continue for a period of thirty (30) days after written notice from Landlord specifying such failure;
- (b) Tenant shall be in default under Section 9.01(a) and such default shall continue for a period of thirty (30) days after written notice from Landlord specifying such failure;
- (c) if any insurance required to be maintained by Tenant shall lapse without replacement, so that any required coverage is not in effect and such failure shall commue for a period of ten (10) days following written notice of Tenant;
- (d) Tenant shall fail to perform or observe any other material obligation, term or provision under this Lease and such failure continues beyond sixty (60) days after written notice from Landlord to Tenant specifying such Event of Default; provided, however, that if Tenant in good faith commences within said 60-day period and thereafter diligently prosecutes all actions required to cure such default, Tenant shall be allowed a reasonable additional period to effect such cure;
- (e) a petition in bankruptcy is filed by or against Tenant, or if Tenant makes a general assignment for the benefit of creditors or is adjudged insolvent by any state or federal court, and in the case of any such involuntary petition, action or proceeding not initiated by Tenant

such petition, action or proceeding is not dismissed or stayed within ninety (90) days after the commencement of such petition, action or proceeding; and

- (f) Tenant shall fail to commence and complete the construction or rehabilitation of the Development in accordance with the Plans and Specifications prior to the Construction Completion Deadline and such failure continues for a period of thirty (30) days after written notice from Landlord specifying such failure; provided, however, that if Tenant in good faith commences within said 30-day period and thereafter diligently prosecutes all actions required to cure any such failure, Tenant shall be allowed a reasonable additional period to effect such cure.
- Lease for so long as the provisions of Section 9.03, 10.14 or any other provision of this Lease that expressly limits Landlord's ability to terminate this Lease precludes such termination. Otherwise Landlord, at its option, at any time thereafter during the continuance of such Event of Default, may give to Tenant and all Leasehold Mortgagees a notice of termination of this Lease, and, upon the date specified in such notice, which date shall be after all cure periods and foreclosure proceeding periods without a cure or foreclosure (or exercise by a Leasehold Mortgagee of other remedies contemplated by Section 9.03(2)) being effected, then this Lease and all of Tenant's rights under this Lease shall expire and terminate as if that date were the date herein originally fixed for the expiration of the Term of this Lease, and on the date so specified, Tenant shall vacate and surrender the Property to Landlord. If an Event of Default under Section 10.01(f) shall occur, then Landlord's right to terminate the entire Lease by reason of such Event of Default shall be limited as hereinafter provided.
- (a) If the construction of a Building on a lot (which for purposes of this Section 10.02 "lot" shall refer to any discrete part of the Real Estate "pon which a Building may be erected) has been Substantially Commenced and if a Leasehold Mortgagee has disbursed proceeds of its loan to pay for the so-called "hard costs" of such construction, the 1, Landlord may not terminate this Lease with respect to the Real Estate upon which such Building is being constructed.
- (b) If any Event of Default under Section 10.01(f) shall occur with respect to a lot on which construction of a Building has not been Substantially Commenced, then subject to Section 9.03. Landlord may terminate this Lease with respect to such lot, but this Lease shall remain in effect with respect to the remainder of the Real Estate.
- (c) If an Event of Default under Section 10.01(f) shall occur with espect to a lot on which construction of the Building has been Substantially Commenced and no Leasehold Mortgagee has advanced proceeds of its loan to pay costs of such construction or any subsequent construction of such Building, then, subject to Section 9.03, Landlord may terminate this Lease with respect to such lot, but this Lease shall remain in effect with respect to the remainder of the Real Estate.
- 10.03 <u>Transfer of Deposits, etc.</u> In the event of any termination of this Lease under Section 10.02, all unearned insurance premiums, all deposits theretofore made by Tenant with utility companies, any claims for refund of any Imposition, any pending claims for insurance proceeds or condemnation awards, and all fuel and supplies on the Property owned by Tenant

shall, subject to the rights of the Leasehold Mortgagees, be deemed to be and are hereby assigned to and transferred to Landlord to be applied in payment of Tenant's liability under this Lease.

- 10.04 <u>Re-entry</u>. In the event of termination of this Lease under Section 10.02 or by operation of law or otherwise, Landlord may without further notice re-enter and repossess the Property.
- 10.05 <u>Injunctive Relief</u>. In the event of any breach or threatened breach by Tenant of any of the covenants, agreements, terms or conditions contained in this Lease, Landlord shall be entitled, after expiration of any applicable notice and cure period, to injunctive relief against such breach or threatened breach, and shall have the right to invoke any right or remedy available at law or in equity or by statute or otherwise as though re-entry, summary proceedings and other remedies were not provided for in this Lease.
- 10.06 Re-letting by Landlord. If Landlord has terminated this Lease in accordance with Section 10.02, Landlord may re-let the Property or any part thereof and receive the rent, whether such rent is in the aggregate greater than or less than the Rent payable hereunder. Landlord shall not be responsible or liable in any way for failure to re-let the Property or any part thereof or for failure to collect any rent due on such re-letting, except as required by law to mitigate Landlord's damages.
- 10.07 Receipt of Monies: No Wa ver. No receipt of money by Landlord from Tenant after termination of this Lease shall reinstate, continue or extend the term of this Lease or of any notice of termination theretofore given to Tenant, or operate as a waiver of Landlord's right to enforce the payment of Rent and any other payment's or charges herein reserved or agreed to be paid by Tenant, then or thereafter falling due, or operate as a waiver of Landlord's right to recover possession of the Property by proper remedy, it being agreed that after service of notice to terminate this Lease or the commencement of any suit or summa y proceedings, or after final order for the possession of the Property, Landlord may demand and collect any monies due or thereafter falling due in any manner without affecting such notice, proceeding, order, suit or judgment, and all such monies collected shall be deemed paid on account of the use and occupancy of the Tenant Property or, at Landlord's election, on account of Tenant's liability hereund at
- 10.08 No Implied Waivers. Landlord's granting of a consent under this Lease, or Landlord's failure to object to an action taken by Tenant without Landlord's consent under this Lease, shall not be deemed a waiver by Landlord of its right to require such consent for any further similar act of Tenant. No waiver by Landlord of any breach of any of the conditions, covenants or agreements of this Lease shall be construed, taken or held to be a waiver of any other breach or be a waiver, acquiescence in or consent to any further or succeeding breach of the same term, condition, covenant or agreement. None of Tenant's covenants, agreements, obligations or undertakings under this Lease, and no breach thereof, may be waived, altered or modified except by a written instrument executed by Landlord.
- 10.09 <u>Remedies Not Exclusive</u>. Subject to provisions of Article 18 and other provisions of this Lease restricting Landlord's right to terminate this Lease, no right, power or remedy conferred upon or reserved to Landlord under this Lease or under law shall be considered exclusive of any other right, power or remedy, but such rights, powers and remedies shall be cumulative and

shall be in addition to every other right, power and remedy given hereunder, or now or hereafter existing at law or in equity, or by statute or otherwise, and every right, power and remedy of Landlord may be exercised from time to time and as often as occasion may arise or as may be deemed expedient, without precluding Landlord's simultaneous or later exercise of any or all other rights, powers or remedies. No delay or omission of Landlord in exercising any right, power or remedy arising from any default shall impair any such right, power or remedy or shall be construed to be a waiver of any such default or an acquiescence therein.

- 10.10 Waiver of Notice. Tenant expressly agrees that any notice of intention to re-enter provided in any statute or to initiate legal proceedings to that end shall run concurrently with any applicable police period provided hereby so that any required notice period shall not be longer than the longer of such statutory notice or notice required under this Lease. Tenant waives, for and on behalf of itself and all persons and parties claiming through or under it (other than any Leasehold Mortgagee), any and all right of redemption provided by any law or now in force or hereafter enacted or otherwise, for re-entry or repossession, or to restore the operation of this Lease, in case Tenant shall be dispossed by a judgment or by warrant of any court or judge, or in case of reentry or repossession by Landford or in case of any expiration or termination of this Lease.
- 10.11 <u>Suits for Damage</u>. Suits for damages or deficiencies, or for a sum equal to any installments of Rent, Impositions and other charges and payments hereunder shall be subject to the provisions of Article 18.
- 10.12 <u>Bankruptcy</u>. Nothing in this Article contained shall limit or prejudice the right of Landlord to prove and obtain as damages in any bankruptcy, insolvency, receivership, reorganization or dissolution proceeding an amount equal to the maximum allowed by any statute or rule of law governing such proceeding and in effect at the time when such damages are to be proved, whether or not such amount be greater, equal to or less than the amount of the damages referred to in any of the preceding Sections.
- 10.13 <u>Leasehold Mortgagee's Rights</u>. Notwithstanding the remedies afforded to Landlord under this Article 10, such remedies shall be subject to and sub-redinate to the Leasehold Mortgagees' rights granted herein.
- 10.14 <u>Investor's Rights in the Event of Tenant Default</u>. The following revisions shall apply for so long as Investor is a member of Tenant:
- (a) Landlord shall give Investor a duplicate copy of all notices of default or other notices that Landlord may give to or serve in writing upon Tenant pursuant to the terms of this Lease. No notice by Landlord to Tenant under this Lease shall be effective unless or until a copy of such notice has been provided to Investor.
- (b) Investor may, at its option and during the time specified for Tenant to cure any default hereunder, either pay any amount or do any act or thing required of Tenant by the terms of this Lease. All payments made and all acts performed by Investor during the cure period shall be effective to prevent a termination of this Lease to the same extent as if they had been performed by Tenant. Tenant hereby authorizes Investor to take any such action at the Investor's option and does hereby authorize entry upon the Property by Investor for such purpose.

- (c) In addition to all other rights of Investor hereunder, Investor's commencement of proceedings to exercise its Removal Right shall be deemed initiation of a cure for purposes of Sections 10.01 and 10.14 provided that each of the following conditions is satisfied:
 - (i) the default is one which cannot be cured only by payment of money:
 - (ii) In the reasonable opinion of Investor, removal of the Manager is necessary:
 - (iii) Investor notifies Landlord within 30 days following receipt of Landlord's default notice of Investor's intention to exercise the Removal Right and does in fact perform all required activity pursuant thereto; and
 - (iv) Investor is performing all other good faith commercially reasonable activity accessary to cure the default.

ARTICLE 11

Additional Pights and Remedies of Landlord

11.01 Performance by Landlord. If Tenant shall at any time fail to make any payment or perform any act to be made or performed by Tenant under this Lease and such failure continues beyond the cure period, if any, applicable thereto under this Lease, and provided that no Leasehold Mortgagee or Investor has cured such failure within the time period provided herein for such cure (provided, in the latter case, that any notice of default required by the terms of this Lease to be given to such Leasehold Mortgagee or Investor by Landlord has been given), Landlord may, at its option (but shall not be required to), make any such payment or perform any such act, and for such purpose Landlord may enter upon the Property and take all actions thereon as may be deemed by Landlord necessary or desirable. Any amount paid or incurred by Landlord in effecting or attempting to cure such failure shall be additional rent due from Tenant to Landlord, and shall be payable by Tenant upon demand.

11.02 Tenant to Provide Indemnification.

- (a) Unless arising from Landlord's negligent act or intentional run conduct or a breach of Landlord's obligations under this Lease or the failure of Landlord to perform its obligations under the Environmental Agreement, or until Landlord shall have re-entered the Property upon expiration or termination of this Lease, Tenant agrees to indemnify, defend and save Landlord's Protected Persons harmless against and from all liabilities, claims, suits, fines, penalties, damages, losses, charges, costs, expenses and fees (including reasonable attorney's fees) which may be imposed upon, incurred by or asserted against Landlord's Protected Persons by reason of any of the following occurring during the portion of the Term during which such indemnitor was Tenant hereunder:
 - (i) any use, non-use, possession, occupation, condition (other than Excluded Environmental Conditions to the extent that responsibility for such Excluded Environmental Conditions is imposed on Landlord under the express provisions of the Environmental Agreement or, if not addressed therein, then under

applicable law), operation, repair, maintenance or management of the Property, or any part thereof, or any occurrence of any of the same;

- (ii) any act or omission on the part of Tenant, resident or any subtenant, licensee or invitee, or any of its or their agents, contractors, servants, employees, licensees or invitees relating to the Property or this Lease;
- (iii) any accident, injury (including death) or damage, regardless of the cause thereof, to any person or property occurring in, on or about the Property or any part thereof;
- (iv) any contest permitted pursuant to the provisions of Section 4.03 or 6.06;
- (v) any litigation or proceeding related to the Property or this Lease to which Landlord becomes or is made a party without fault on its part, whether commenced by or against Tenant; and
- (vi) any costs which may be incurred by Landlord in enforcing any of the covenants, agreements, terms and conditions of this Lease (provided Landlord prevails in the enforcement proceeding).
- (vii) The obligations of Tenant under this Section 11.02(a) do not apply to Excluded Environmental Conditions.
- (b) As to claims for which Tenant does not receive timely notice, there shall be no obligation of Tenant to indemnify.

Tenant's obligations under this Section 11.02 shall survive the expiration or termination of this Lease.

11.03 <u>Excluded Environmental Conditions</u>. Except as expressly provided in the Environmental Agreement and this Lease, all liability and responsibility with respect to Excluded Environmental Conditions shall be determined pursuant to applicable federal, state or local law.

ARTICLE 12

Eminent Domain

Property or the Tenant Property, or such substantial portion of the Property or the Tenant Property as shall in the reasonable good faith judgment of Tenant, subject to the approval of the First Leasehold Mortgagee, which shall not be unreasonably withheld, make it economically unfeasible to continue to operate the remaining portion for the purposes herein stated, shall be taken by the exercise of the power of eminent domain, this Lease shall terminate on the date of vesting of title in the condemnor under such eminent domain proceedings, and all Rent and other sums payable by Tenant hereunder shall be prorated to the date of such vesting, and thereafter Tenant shall be relieved of all obligations to pay the Rent and to otherwise perform its agreements, obligations and

undertakings under this Lease except those that expressly survive the termination of this Lease. The award granted with respect to such eminent domain proceedings shall be divided between Landlord, Tenant and any Leasehold Mortgagees in the following order:

- (a) to the First Leasehold Mortgagee, an amount sufficient to obtain the release and satisfaction of the First Leasehold Mortgage;
- (b) to any and all other Leasehold Mortgagees, as their interests appear, an amount sufficient to obtain a release and satisfaction of the Leasehold Mortgages, with payment being made in full to such Leasehold Mortgagees according to the priorities of their Leasehold Mortgages:
- to Tenant, an amount equal to the sum of: (y) the greater of: (1) the fair market value of the Improvements and the fair market value of the unexpired Leasehold Estate, reduced by the amount if any, paid under the preceding Clauses (a) and (b); and (2) the replacement cost of the Improvements and the fair market value of the unexpired Leasehold Estate, reduced by the amount, if any, paid under the preceding Clauses (a) and (b); plus (z) the value of any low-income housing tax credits and Illinois Affordable Housing Tax Credits recaptured or not available in future years as a result of such taking; and
 - (d) the balance, if any, shall be paid to Landlord.

If this Lease is terminated under this Section, then Tenant shall, if so directed by Landlord, demolish and/or remove any damaged Improvements on any remaining Property at the sole cost and expense of Tenant provided that all condemnation proceeds allocable to the Tenant Property remaining after satisfaction of the indebtedness secured by any Leasehold Mortgages shall be available to Tenant. The obligation under this Lease to be notish and/or remove Improvements under the foregoing sentence shall not apply to any Leasehold Mortgagee (or nominee of a Leasehold Mortgagee) that succeeds to Tenant's interest under this Lease through foreclosure of its Leasehold Mortgage or deed-in-lieu thereof.

12.02 Partial Taking.

(a) If, during the Term, less than the entire Property of Tenant Property (excluding any substantial portion covered under Section 12.01 above) shall be taken by the exercise of the power of eminent domain, and, in the reasonable judgment of the First Leasehold Mortgagee, condemnation proceeds attributable to Tenant's interest in the Property are attricient to restore that portion of the Property remaining after the taking so as to be not materially different from the value, condition and character of the Property prior to such taking, this Lease shall not terminate but shall continue in full force and effect for the remainder of the Term, subject to the provisions of this Section 12.02. The amount of damages resulting to Landlord and Tenant, respectively, and to their respective interests in the Property and in, to and under this Lease, by reason of such exercise and partial taking under such eminent domain proceedings shall be separately determined and computed by the court having jurisdiction of such proceedings, and separate awards and judgments with respect to such damages to Landlord and Tenant shall be made and entered, and said awards shall, subject to Section 12.05, be paid to Landlord and Tenant, respectively, in accordance therewith; provided, however, that Tenant shall receive that portion of

the award made as consequential damages to the Improvements located on the remaining portion of the Property and Tenant, at its expense, shall forthwith restore the remaining portion of the Improvements to substantially the same value, condition and character as existed prior to such taking (to the extent such restoration is possible, taking into account the extent to which a portion of the Improvements have been removed as a result of the taking), using such part of the award received by Tenant in said eminent domain proceeds as may be necessary and, if the amount of such award is not sufficient, Landlord shall make its portion of the award available for such restoration. If Tenant is obligated to restore the Property, the proceeds of the award shall be deposited in the Restoration Escrow and disbursed to pay the costs of such restoration. If the sum of such awards is not sufficient, Tenant shall have the right, but not the obligation, to provide the additional funds required.

fithe First Leasehold Mortgagee reasonably determines that condemnation proceeds are insufficient to restore that portion of the Property remaining after the taking so as to be not materially different than the value, condition and character of the Property prior to such taking, and neither Tenarit nor any Leasehold Mortgagee deposits into the Restoration Escrow the additional funds necessary to satisfy such deficiency within ninety (90) days after the condemnation award, and Landford makes no commitment to provide additional funds within such ninety (90) days and deposits into the Restoration Escrow the additional funds necessary to satisfy such deficiency within one hundred twenty (120) days after the condemnation award, then the condemnation proceeds shall be applied as set forth in Section 12.01 and the requirements of Section 8.05 shall apply.

12.03 <u>Temporary Taking</u>. In the event of a taking for a temporary use, this Lease and the Term shall continue and the Rent thereafter due and payable shall be equitably reduced or abated. Tenant shall continue to perform and observe all of the other covenants, agreements, terms and conditions of this Lease. The entire amount of any proceeds with respect to such temporary taking shall be paid to Tenant.

12.04 Other Governmental Action. In the case of any governmental action not resulting in the taking of any portion of the Property or the Tenant Property but creating a right to compensation, this Lease shall continue in full force and effect without reduction or abatement of any Rent thereafter due and payable. If such governmental action results in any damage to the Improvements, Tenant shall be entitled to receive such portion of the proceeds (or all of the proceeds, if required for such purpose) estimated to be necessary to remedy any such damage and to compensate for the loss of low income housing tax credits and Illinois Affordable Housing Tax Credits, and Tenant shall proceed with reasonable diligence to make all repairs, replacements, restorations and improvements necessary so to remedy such damage to the extent economically feasible, and, if the amount of such proceeds is not sufficient, Tenant shall have the right, but not the obligation, to provide the additional funds required. Any balance remaining from such proceeds, or if no damage is involved then all of such proceeds, shall be divided between Landlord and Tenant as their respective interests may appear.

12.05 <u>Leasehold Mortgagees</u>. The rights granted to Landlord and Tenant under this Article 12 shall be subject to the rights and interests of the Leasehold Mortgagees under their respective Leasehold Mortgages (except as provided in Section 12.02(b)) pursuant to their respective priorities.

ARTICLE 13

Estoppel Certificates

Upon written request by either party, any Leasehold Mortgagee or the Investor, the party to whom the request was made will promptly certify to the requesting Person, or to any proposed assignee or grantee or mortgagee or trustee under deed of trust or trust deed or the proposed assignee of such mortgagee, deed of trust or trust deed, whether or not this Lease is valid and subsisting, whether or not it has been modified (and if there are modifications, stating them) and whether or not the party executing the certificate has knowledge of any default or breach by the other party under any of the terms of this Lease (and if any exists, stating them). If the party to whom a writer, request is directed under the preceding sentence shall fail to furnish the requested certificate within eventy (20) days after the receipt of such request, then by such failure such party shall be deemed to 1.3 /e certified to the requesting Person and to any proposed assignee or grantee or mortgagee or trustce under a deed of trust or trust deed, that this Lease is valid and subsisting, that there have been no modifications to this Lease, and that there are no known defaults or breaches by the other party under the terms of this Lease. Upon the issuance of a certificate of occupancy for the final Building in the Development by the City in its municipal capacity following completion of the construction or rehabilitation of the Improvements, Landlord shall give to Tenant an estoppel certificate (in recordable form) certifying all obligations set forth in Section 5.01 have been satisfied, and Tengot shall cause such certificate to be recorded.

ARTICLE 14

Surrender at End of Term; Title to Improvements

14.01 Surrender at End of Term. Upon the expiration of the Term, or earlier termination of the Lease, all Improvements then on the Real Estate, hall, together with all fixtures, equipment and other personal property owned by Tenant and used ir connection with the operation of the Development, shall become the property of Landlord without any payment or allowance whatever by Landlord on account of or for such Improvements, fixtures, equipment and personal property, whether or not the same or any part thereof shall have been constructed by, paid for, or purchased by Tenant. Tenant shall vacate and surrender possession of the Tenant Property to Landlord without delay, free and clear of all lettings, occupancies, and licenses, and free and clear of all liens, claims, encumbrances and security interests other than (i) the Permitted Exceptions, (ii) the rights of tenants in possession under leases (which shall expire not later than one (1) year after the end of the Term), (iii) those, if any, created by Landlord and (iv) those related to Excluded Environmental Conditions for which Tenant or the Owner is not responsible under the express provisions of the Environmental Agreement or, if not addressed therein, under applicable law. Tenant agrees to execute and deliver to Landlord such quit claim deeds, bills of sale, assignments or other instruments of conveyance as Landlord may reasonably deem necessary to evidence such transfer of title to Landlord. Tenant hereby waives any notice now or hereafter required by law with respect to vacating the Property at any such termination date.

14.02 <u>Title to Improvements</u>. Landlord acknowledges and agrees that throughout the Term and until expiration or earlier termination of this Lease, title to all Improvements shall be in Tenant's name and that Tenant has, and shall be entitled to, all rights and privileges of ownership

of such Improvements, including without limitation: (a) the right to claim depreciation or cost recovery deductions; (b) the right to claim the low-income housing tax credit described in Section 42 of the Code; (c) the right to amortize capital costs and to claim any other federal or state tax benefits attributable to the Tenant Property; and (d) the right to transfer such Improvements in accordance with the terms of this Lease; provided, however, that Tenant may not remove or substantially alter any of the Improvements (other than the disposition and replacement of equipment, appliances and other personal property in the ordinary course of business or in connection with the performance of its obligations under Section 6.04) without having first obtained the prior written consent of Landlord, which shall not be unreasonably withheld or delayed. If requested by Tenant, and to the extent legally permissible, Landlord will deliver a deed, in recordable form, evidencing Tenant's fee title to the Improvements.

ARTICLE 15

Landlord Defaults

- 15.01 <u>Landlord's Default</u>. Each of the following events shall be an event of default by Landlord under this Lease:
 - (a) Landlord's failure to lease by this Lease the Real Estate; or
- (b) Landlord's failure to materially perform any other term or provision to be performed by Landlord under this Lease, not otherwise described in subsection (a) hereof, and such failure shall continue beyond sixty (60) days after written notice received by Landlord from Tenant specifying such event of default; provided, however, that if Landlord in good faith commences within said 60-day period and thereafter differently prosecutes all actions required to cure such default, Landlord shall be allowed a reasonable period to effect such cure.

Upon an event of default by Landlord hereunder, Tenant shall have all of the rights and remedies afforded at law or in equity, subject to Section 18.01.

- 15.02 <u>Injunctive Relief</u>. Upon any event of default by Landlord, Tenant shall, in addition to any other remedies available to Tenant at law or in equity, be entitled to enjoin such breach or threatened breach, and shall have the right of specific performance, it being the agreement of the parties hereto that in certain circumstances of Landlord's event of default, Tenant's remedies at law may be inadequate to afford it the practical realization of the agreements herein reacle by the parties.
- 15.03 Remedies Not Exclusive. No right, power or remedy conferred upon or reserved to Tenant under this Lease, or under law, shall be considered exclusive of any other right, power or remedy, but such rights, powers and remedies shall be cumulative and shall be in addition to every other right, power and remedy given hereunder, or now or hereafter existing at law or in equity, or by statute. Every right, power and remedy given by this Lease may be exercised from time to time and as often as occasion may arise or may be deemed expedient, without precluding Tenant's simultaneous or later exercise of any or all other rights, powers or remedies. No delay or omission of Tenant to exercise any right, power or remedy arising from Landlord's event of

default shall impair any such right, power or remedy or shall be construed to be a waiver of any such default or an acquiescence therein.

- 15.04 <u>Waivers in Writing</u>. None of Landlord's covenants, agreements, obligations or undertakings, and no events of default of Landlord may be waived, altered, or modified except by a written instrument executed by Tenant and all Leasehold Mortgagees.
- 15.05 <u>Landlord's Representations</u>. Landlord hereby represents and warrants to Tenant that:
- (a) the entry by Landlord into this Lease with Tenant, and the performance by Landlord of all of the terms and conditions contained herein will not, or with the giving of notice or the passage of time, or both, would not, violate or cause a breach or default under any other agreement relating to the Property to which Landlord is a party or by which it is bound;
- (b) as of the Commencement Date, there is no tenant or other occupant of the Real Estate having any right or claim to possession or use of the Real Estate other than public or quasi-public utilities; and
- (c) as of the Commencement Date, there are no special assessments of which Landlord has received notice for sever, sidewalk, water, paving, gas electrical, or utility improvements or other capital expenditures, matured or unmatured, affecting the Real Estate.

ARTICLE 16

Notices

Unless otherwise specified, any notice, demand or request required hereunder shall be given in writing at the addresses set forth below, by any of the icllowing means: (a) personal service; (b) electronic communications, including but not limited a ejectronic mail; (c) overnight courier, receipt requested; or (d) registered or certified mail, return receipt requested.

If to Landlord: Chicago Housing Authority

60 East Van Buren Street, 12th Floor

Office

Chicago, Illinois 60605

Attention: Chief Executive Officer

with copy to: Chicago Housing Authority

60 East Van Buren Street, 12th Floor

Chicago, Illinois 60605 Attention: General Counsel

If to Tenant: RS Affordable I LLC

c/o Related Midwest

350 West Hubbard Street. Suite 300

Chicago, Illinois 60654 Attention: Jacques Sandberg

with copy to: DLA Piper LLP (US)

444 West Lake Street, 9th Floor

Chicago, Illinois 60606

Attention: Kimberlie Pearlman

And: Heartland Housing, Inc.

208 South LaSalle Street, Suite 1300

Chicago, Illinois 60604 Attention: Mark Kruse

Applegate & Thorne-Thomsen P.C. 425 South Financial Place, Suite 1900

Chicago, Illinois 60605

Attention: Greg C. Whitehead

If to Management Agent to: Related Management Company, L.P.

350 W. Hubbard Street, Suite 100

Chicago Illinois 60654

Attention: Joseph LaMantia, Senior Vice President

With a copy to: Relate (Management Company, L.P.

30 Hudson Yards, 72nd Floor New York, New York 10001

Attention: Jonathan Callahan, General Counsel

If to Investor, to: Hudson Roosevelt Square LLC

c/o Hudson Housing Capital LLC 630 Fifth Avenue, 28th Floor New York, New York 10111 Attention: General Counsel

With a copy to: Holland & Knight LLP

10 St. James Avenue, 12th Floor Boston, Massachusetts 02116 Attention: Dayna M. Hutchins, Esq.

If to HUD, to: United States Department of Housing and Urban Development

77 West Jackson Blvd., 26th Floor

Chicago, Illinois 60604 Attention: Regional Counsel

And: United States Department of Housing and Urban Development

77 West Jackson Blvd., 24th Floor

Chicago, Illinois 60604

Attention: Director of Public Housing

And: United States Department of Housing and Urban Development

451 Seventh Street, S.W. Washington, D.C. 20410

Attention: Assistant Secretary of Public and Indian Housing

If to City: Department of Housing

City of Chicago

121 North LaSalle Street, Room 1000

Chicago, Illinois 60602 Attention: Commissioner

In addition, concurrently with the giving of any notice or demand by Landlord to Tenant, or by Tenant to Landlord, Landlord or Tenant, as the case may be, shall furnish a copy of such notice to any Leasehold Mongagee, including the Leasehold Mongagees listed on Exhibit C at their respective addresses set forth therein, and to any other party listed on Exhibit C.

By written notice served in the foregoing manner, any party entitled to receive notices shall have the right to designate another person and another address to which notices and demands shall thereafter be sent. Each such notice or demand shall be deemed served, given and received when received or, when given by mail, shall be deemed served, given and received on the third business day after the mailing thereof.

ARTICLE 17

Miscellancovs

- 17.01 Covenants Running With Land. All terms, provisions, conditions, covenants, agreements, obligations and undertakings contained in this Lease shall, except as herein specifically limited or otherwise provided, extend and inure to be binding upon Landlord's successors and assigns and Tenant's successors and permitted assigns, as if such successors and assigns were in each case specifically named, and shall be construed a covenants running with the land. Wherever reference is made in this Lease to either party, it shall be held to include and apply to such successors and assigns.
- 17.02 <u>Amendments in Writing</u>. In no event shall this Lease or any terms, provisions or conditions hereof be deemed to be amended, modified or changed in any manner whatsoever, except and unless set forth and provided for in writing executed by Landlord and Tenant, and consented to in writing by any Leasehold Mortgagee.
- 17.03 Quiet Possession. Landlord represents and warrants that it has full right and power to execute and perform this Lease and to convey the rights and interest demised hereby. Landlord agrees that during the Term and so long as no Event of Default exists and is continuing hereunder, Tenant shall and may peaceably and quietly have, hold and enjoy the Real Estate demised hereby, subject to the Permitted Exceptions, without molestation or disturbance by or from Landlord or any party claiming by, through or under Landlord, and free of any encumbrance created or suffered by Landlord except those expressly described herein to which this Lease is made subject and subordinate.

- 17.04 <u>Time of Essence</u>. Time is of essence of this Lease and of the performance of the respective obligations, covenants and agreements of Landlord and Tenant hereunder. If the day for the performance of any obligation hereunder occurs on a calendar day other than a business day, the time for such performance shall be extended to the next business day.
- 17.05 <u>Approvals</u>. All approvals or consents required under the provisions hereof shall be in writing. Unless herein expressly otherwise provided, any approval or consent of Landlord shall be sufficiently given if signed by Landlord's Chief Executive Officer.
- 17.06 Representations and Warranties. Landlord has made no warranties or representations whatsoever with respect to the Real Estate and, except for the obligations of Landlord under the Environmental Agreement and as set forth in Section 6.01 above, Tenant accepts the Real Estate "as-is"; provided, however, that the foregoing shall not affect the obligations, if any, of Landlord under this Lease with respect to matters of title to the Real Estate and liens arising out of theorem and/or materials furnished to the Real Estate, or any portion thereof, by or on behalf of Landlord.
- 17.07 <u>Captions</u>. The table of contents and captions of this Lease are for convenience of reference only and in no way define, limit or describe the scope or intent of this Lease nor in any way affect this Lease.
- 17.08 <u>Partial Invalidity</u>. If any term, provision or condition of this Lease or its application to any Person or circumstance shall to any extent be invalid or unenforceable, the remainder of this Lease and the application of such term, provision or condition to Persons or circumstances other than those as to which it is held invalid or unenforceable shall not be affected thereby, and each term, provision and condition of this Lease shall be valid and be enforced to the fullest extent permitted by law.
- 17.09 Applicable Law. This Lease shall be construct and enforced in accordance with the law of the State of Illinois.
- 17.10 <u>Recording of Lease</u>. This Lease shall be recorded in its entirety with the Cook County Clerk's Office.
- 17.11 <u>Lease Not to be Construed Against Either Party</u>. The parties have each been represented by counsel in connection with the negotiation and drafting of this Lease. Accordingly, this Lease shall not be construed against or for either party.
- 17.12 <u>Cooperation</u>. Landlord and Tenant agree that they will cooperate with one another in all respects in furtherance of the Development. From time to time, Tenant may request modifications to this Lease to satisfy the requirements of financing sources, including without limitation government agencies and private lenders and equity sources. Landlord will use all reasonable efforts to accommodate such requests and will not unreasonably withhold or delay its approval and execution of modifications to this Lease that do not materially and adversely alter the basic terms hereof or Landlord's rights hereunder. Nothing herein shall impose upon Landlord any requirement to approve any modification or amendment to this Lease that would violate or contravene any applicable law or any contract or agreement to which Landlord is a party or which is binding on Landlord. Landlord agrees that it will, upon request of Tenant, from time to time.

enter into an amended and restated lease combining into one document the entire Lease and all modifications and amendments theretofore entered into. Tenant shall pay, or reimburse Landlord upon demand, for all reasonable out of pocket expenses incurred by Landlord in connection with any such modification or amendment.

ARTICLE 18

Exculpatory Provisions

Tenant, and any Person claiming by, through or under Tenant (including without limitation all Leasehold Mortgagees) that none of Landlord's covenants, undertakings or agreements herein set forth are made or intended as personal covenants, undertakings or agreements of Landlord, but are for the purpose of binding the premises demised hereby, and liability or damage for breach for nonperformance by Landlord shall be collectible only out of the Real Estate demised hereby or available insurance proceeds and no personal liability is assumed by nor at any time may be asserted or enforced against Landlord or any other Landlord's Protected Persons or any of its or their heirs, legal representatives, successors or assigns, all such personal liability, if any, being expressly waived and released by Tenant and each Person claiming by, through or under Tenant. Nothing contained in this Section 18.01, however, shall in any way or manner limit the full recourse of Tenant against Landlord under the Environmental Agreement, or under any non-monetary remedy granted Tenant in Section 15.02. This Section 18.01 shall not apply to HUD at any time HUD is Tenant under this Lease.

18.02 Exculpatory Provision - Tenant. Tenant, but not any partner, officer, director, shareholder, member or manager of Tenant, nor any employee or agent of any of the foregoing, shall be personally liable for payment or performance under this Lease, it being acknowledged that Landlord's exclusive rights and remedies hereunder shall be limited to Tenant's interest in this Lease and the Improvements and any other asset of Tenant and to the extent provided for in Section 10.02, for the termination of this Lease and re-entry and possession of the Property. No deficiency judgment shall be sought or obtained against Tenant or any partner, officer, director, shareholder, member or manager of Tenant, nor any employee or agent of any of the foregoing (collectively, "Exculpated Parties") for any amount due under this Lease; provided, however, that, except as hereinafter provided in this Section 18.02, nothing contained herein shall either relieve the Exculpated Parties from personal liability and responsibility, or limit Landlord's other rights and remedies against Tenant hereunder, either at law or in equity: (i) for fraudurent acts; (ii) for the fair market value of any personal property or fixtures removed or disposed of from the Property in violation of the terms of this Lease; (iii) for waste committed by Tenant with respect to the Property; (iv) for insurance proceeds and condemnation awards received by Tenant and not turned over to Landlord or used by Tenant for restoration or repair of the Property to the extent required under this Lease; and (v) for any rents or other income from the Tenant Property received by Tenant after an Event of Default under this Lease and not applied to the fixed and operating expenses of the Development. Notwithstanding the preceding sentence, if Tenant is a limited partnership, the liability of a limited partner of Tenant shall be limited to extent provided in the Illinois Uniform Limited Partnership Act (805 ILCS 210), or any successor thereto, or if Tenant is a limited liability company, the liability of a member of Tenant shall be limited to the extent provided in the Illinois Limited Liability Act (805 ILCS 180), or any successor thereto.

ARTICLE 19

Regulatory Agreements

19.01 <u>Regulatory Agreements</u>. Concurrently with, and as a condition to the Permitted Assignment, the Owner, as Tenant under this Lease pursuant to an Assignment and Assumption and Amendment of Ground Lease, shall enter into those agreements which shall be described in a revised <u>Exhibit E</u> (collectively, the "Regulatory Agreements") relating to the operation of the Development.

ARTICLE 20

Hazardous Materials

20.01 Proxibition Against Hazardous Materials. Tenant shall not cause any Hazardous Material to be generated, released, stored, buried or deposited over, beneath, in or on (or used in the construction of) the Real Estate or the Development, or any part thereof, from any source whatsoever, other than in accordance with this Lease, the Environmental Agreement, the applicable NFR Letters and applicable Environmental Laws. Except for Excluded Environmental Conditions (to the extent that Tenam is not responsible for such Excluded Environmental Conditions under the express provisions of the Environmental Agreement or, if not addressed therein, then under applicable law), Tenent shall not permit any Hazardous Material to be generated, released, stored, buried or deposited over, beneath, in or on (or used in the construction of) the Real Estate or the Property, or any part thereof, from any source whatsoever, other than in accordance with this Lease, the Environmental Agreement, the applicable NFR Letters and applicable Environmental Laws. Tenant shall (i) comply at its own cost and expense with all Environmental Laws; (ii) not cause or permit any Hazardous Materials to be brought upon, kept or used in or about the Property by Tenant, its agents, employees, contractors, guests or invitees unless all of the following requirements are satisfied at all times with respect thereto: (A) such Hazardous Materials are reasonably required in the ordinary course of Tenant's business which it is authorized to conduct on the Property under the terms of this Lease, (B) such Hazardous Materials are maintained only in such quantities as are reasonably necessary for the operations which Tenant is authorized to conduct on the Property under this Lease, (C) such Hazardous Materials are used strictly expecordance with the manufacturers' instructions therefor and in compliance with all applicable Environmental Laws, (D) such Hazardous Materials are not disposed of in or about the Property in a manner that would constitute a release or discharge thereof, and (E) at such Hazardous Materials are removed from the Property by Tenant upon the expiration or earlier termination of this Lease or upon termination of Tenant's right of possession of the Property; (iii) not dispose of Hazardous Materials in dumpsters at the Property; (iv) not discharge Hazardous Materials into building drains or sewers; (v) not cause or allow the release of any Hazardous Materials on, to, or from the Property; and (vi) arrange at its sole cost and expense for the lawful transportation and off-site disposal in accordance with all applicable Environmental Laws, of all Hazardous Materials that it generates.

- **20.02** Environmental Agreement. The covenants and obligations of Tenant under Sections 20.01 and 20.03 are in addition to the obligations of Tenant and the Owner under the Environmental Agreement.
- 20.03 Indemnity. Certain indemnification obligations of the Owner with respect to environmental matters are set forth in the Environmental Agreement. Accordingly, in the event of a conflict between application of the provisions of this Section 20.03 to the Initial Tenant and the Owner, as the assignee of Initial Tenant's Leasehold Estate and the indemnification obligations of the Owner under the Environmental Agreement, the indemnification obligations under the Environmental Agreement shall govern and control. Tenant shall indemnify, defend and hold harmless Landlord's Protected Persons, and any current or former officer, director, employee or agent of Landlord (collectively, the "Indemnitees") from and against any claims, demands, penalties, fines, l'abilities, settlements, damages, costs or expenses, including, without limitation, attorney's and cor. ultant's fees, investigation and laboratory fees, court costs and litigation expenses, known or unknown, contingent or otherwise, arising out of or in any way related to: (i) a violation of Section 25.01; (ii) any violation of an NFR Letter caused by Tenant, the Owner or Developer or any of their respective employees, agents, contractors, guests or invitees; or (iii) any exacerbation of a Pre existing Environmental Condition caused by Tenant, the Owner, Developer or any employee, agent or contractor of Tenant, the Owner or Developer; provided, however, that Tenant's obligations under this Section 20.03 shall not apply to Excluded Environmental Conditions except to the extent that responsibility for such Excluded Environmental Conditions is imposed on Tenant or the Owner under the express provision of the Environmental Agreement or, if not addressed therein, then under applicable law.
- **20.04** Survival. Tenant's obligations under this Article 20 shall survive the expiration or termination of this Lease.

ARTICLE 21

HUD PUBLIC HOUSING PROVISIONS

- 21.01 <u>Term.</u> This Lease shall be for a minimum term (i) commencing on the date of this Commencement Date, and (ii) unless otherwise provided by law, terminating or the latest to occur of: (A) expiration of the minimum period during which the Public Housing Units are required by law to be operated as public housing in accordance with the Housing Act; and (B) the expiration of 40 years from the date the Project becomes available for occupancy.
- **21.02** <u>Use of Development</u>. Tenant shall throughout the Term continuously use and operate the Development only for the following uses, and such other uses as are reasonably and customarily attendant to such uses: construction, development, marketing for lease and leasing of the Affordable Units in a manner which strictly satisfies the applicable requirements of this Lease, the Applicable Public Housing Requirements and the PBV Requirements.

21.03 Covenants Applicable to Public Housing Units.

(a) The Public Housing Units are subjected to, and benefitted by, the terms and conditions of the Applicable Public Housing Requirements. The provisions of the Applicable

Public Housing Requirements and this Section are intended to create a covenant running with the land and, subject to the terms and benefits of the Applicable Public Housing Requirements, to encumber and benefit the Development for the entire Term of this Lease. The Applicable Public Housing Requirements and this Section 21.03 shall be binding upon Landlord and Tenant and each of their respective successors and assigns, including, without limitation, any entity which succeeds to Tenant's interest in the Property by foreclosure or an instrument in lieu of foreclosure, and expressly include, but are not limited to, the following obligations:

- (b) Except as otherwise provided in the Housing Act, the Public Housing Units shall be operated under the terms and conditions applicable to public housing, as set forth in the Applicable Fublic Housing Requirements, during the 40-year period that begins on the date on which the Project becomes available for occupancy, as required by section 9(d)(3)(A) of the Housing Act (or any successor provision).
- (c) Except as otherwise provided in the Housing Act, the Project shall be maintained and operated under the terms and conditions applicable to public housing, as set forth in the Applicable Public Housing Requirements, during the 20-year period that begins on the latest date on which modernization with public housing capital funds is completed, as required by section 9(d)(3)(B) of the Housing Act (or any successor provision).
- (d) Except as otherwise provided in the Housing Act, no portion of the Project may be disposed of before the expiration of the 10-year period beginning upon the conclusion of the fiscal year for which such amounts were provided, as required by section 9(e)(3) of the Housing Act (or any successor provision).
- (e) Neither the Public Housing Units, for any part thereof, may be demolished other than in accordance with the Applicable Public Housing Requirements.
- (f) Tenant agrees that, with the exception of (A) the First Leasehold Mortgage and any mortgage(s) held by the Landlord that have been approved by HUD, or any other Permitted Exceptions listed in Exhibit B; (B) dwelling leases with eligible families for the Public Housing Units; and (C) normal uses associated with the operation of the Project, neither the Project nor any portion thereof shall be encumbered in any way, nor the assets of the Project pledged as collateral for a loan, without the prior written approval of Landlord and HUD.
- 21.04 Amendments to Plans and Specifications. Tenant shall take no action to effectuate any material amendments, modifications or any other alterations to the Indis and Specifications unless Landlord has approved such, in writing and in advance. Landlord's execution of this Lease also constitutes a certification to HUD that prior to making any such amendments, modifications or alterations to the Plans and Specifications that such amendments, modifications or alterations are in accordance with its design and construction standards at 24 CFR § 905.312.
- 21.05 <u>Alterations</u>. Tenant shall not make any alteration, improvement or addition to the Property having a cost greater than One Flundred Thousand Dollars (\$100,000), or such lesser amount as may be provided in the Plan and Specifications, or demolish any portion thereof, without first presenting to Landlord complete plans and specifications therefor and obtaining Landlord's

and HUD's written consent thereto (which consent shall not unreasonably be withheld so long as, in Landlord's and HUD's judgment such alteration, improvement, addition or demolition will not violate the Applicable Public Housing Requirements or this Lease, or impair the value of the Property). HUD's right under the preceding sentence shall be extinguished upon the release of the Declaration of Restrictive Covenants in favor of HUD encumbering the Development. Any improvements made to the Development by either party hereto shall be made only in good and workmanlike manner using new, materials of the same quality as the original improvements, and in accordance with all applicable building codes and the Applicable Public Housing Requirements.

21.06 Restoration. If any portion of this Lease conflicts with Section 11 of the ACC, the provisions of Section 11 of the ACC shall control.

21.07 Assignment and Subletting.

- (a) <u>Consent.</u> This Agreement shall be binding upon and inure to the benefit of the successors and assigns of the Landlord and the Tenant, except that Tenant may not assign or sublet its interest in this Lease without the prior written consent of the Landlord and HUD. Any attempted transfer without such consents shall be null and void.
- (b) Prohibited Transfers. Tenant agrees for itself and its successors and assigns in interest hereunder that it will not, other than by the First Leasehold Mortgage or other Leasehold Mortgage approved in Exhibit C: (1) assign this Lease or any of its rights under this Lease as to all or any portion of the Property, the Public Housing Units, the rest of the Improvements, the unit equipment or the Development generally, or (2) make or permit any voluntary or involuntary total or partial sale, lease, assignment, conveyance, mortgage, pledge, encumbrance or other transfer of any or all of the Property, the Public Housing Units, the rest of the Improvements, the unit equipment or the Development or the occupancy or use thereof, other than in accordance with the Applicable Public Housing Requirements and this Lease (including but not limited to (i) any sale at foreclosure or by the execution of any judgment of any or all of Tenant's rights hereunder, or (ii) any Transfer by operation of law), without first obtaining Landlord's and HUD's express written consent thereto.
- (c) <u>HUD Restrictions on Transfers.</u> In addition to the transfers described in paragraph (b) of this Section, no transfer, conveyance, or assignment shall be made, without the prior written approval of HUD, of: (1) any interest of a managing member, general partner, or controlling stockholder (any such interest being referred to as a "Controlling Interest") of the Tenant; or (2) a Controlling Interest in any entity that has a Controlling Interest in the Tenant; or (3) prior to the payment of full of all equity contributions required under the Mixed Finance Amendment to the ACC, any other interest in the Tenant, or in any partner or member thereof (each of such transfers, conveyances and assignments, together with the transfers described in paragraph (b) of this Section, is hereafter referred to as a "Transfer"). Notwithstanding the foregoing, HUD consent is not required where a business organization that has a limited interest (i.e., non-Controlling Interest and non-managing) in the Tenant transfers a non-Controlling Interest and non-managing interest in the business organization, provided that the Tenant: (1) provides HUD with written notice of such transfer; and (2) certifies to HUD that the new owner of the limited interest remains obligated to fund its equity contribution in accordance with the HUD-approved organizational documents of the Tenant. If Tenant requests HUD's consent to an internal

reorganization of the Tenant, or of any of the partners, members, or stockholders of Tenant, HUD will not unreasonably withhold or delay such consent.

- (d) Any person to whom any Transfer is attempted without such consent shall have no claim, right or remedy whatsoever hereunder against Landlord, and Landlord shall have no duty to recognize any person claiming under or through the same.
- Notwithstanding the foregoing, the prior approval of HUD shall not be required for the exercise by the investor member, i.e., limited partner, limited owner, etc. or its affiliates, of its rights to remove a Controlling Interest of the Owner or partner or member thereof and to designate one of the members, or an affiliate of one of the members as a substitute Controlling Inverest under the terms of the Owner's Amended and Restated Operating Agreement. dated as of subscantially even date herewith, as may be amended (the "Operating Agreement"), provided that HUD is given prior written notice of default under the Operating Agreement and of the exercise of the removal and appointment right therein (the "Notice"). However, HUD consent shall be required for the appointment of such substitute Controlling Interest to extend beyond a ninety (90) day period. Such 90-day period will commence on the date of the Notice (the "Interim Replacement Period"). With notice to HUD, the Interim Replacement Period may be extended for an additional 90 days to allow the substitute Controlling Interest of the Owner or partner or member thereof to find a replacement Controlling Interest acceptable to HUD and all other parties, provided that prior to the expiration of such additional 90-day period, the substitute Controlling Interest demonstrates that the Investor is continuing to fund (or has already funded) its equity contribution, as required under the Operating Agreement, and that the Project continues to be operated in a manner consistent with the Applicable Public Housing Requirements.
- Notwithstanding the foregoing, the prior approval of HUD shall not be required for the Controlling Interest in the Owner to be colleterally assigned and pledged to the First Leasehold Mortgagee, and to allow the First Leasehold Mortgagee to exercise any of its rights pursuant thereto, so long as the First Leasehold Mortgagee gives prompt written notice to HUD of the exercise of such rights at the time of such exercise of its rights under such collateral assignment (the "Pledge Notice"). However, the consent of HUD and the Authority shall be required for the appointment of any substitute Controlling Interest (including the First Leas Hold Mortgage or its Affiliates) extending beyond a 90-day period. Such 90-day period will commence on the date of the Pledge Notice (the "Pledge Replacement Period"). With notice to the Landierd and notice and prior written approval of HUD, the Pledge Replacement Period may be extended for an additional 90 days to allow the substitute Controlling Interest of the Owner to find a replacement Controlling Interest acceptable to HUD and the Landlord provided that prior to the expiration such additional 90-day period, the substitute Controlling Interest demonstrates that the Investor is continuing to fund (or has already funded) its equity contribution as required by the Owner's Operating Agreement and that the Project continues to be operated in accordance with the Applicable Public Housing Requirements.

- 21.08 <u>HUD's Rights on Event of Default</u>. Upon the occurrence of an Event of Default that also constitutes a substantial default under the ACC with respect to the Public Housing Units, HUD may:
- (1) require Landlord to convey to HUD its fee simple interest in the Project, and ensure the Tenant's conveyance to HUD of its leasehold interest in the project, if, in HUD's determination (which determination shall be final and conclusive), such conveyance of title is necessary to achieve the purposes of the Housing Act; or
 - (2) require Tenant to deliver possession and control of the Project to HUD; or
- exercise any other right or remedy existing under applicable law, or available at emity HUD's exercise or non-exercise of any right or remedy under the ACC shall not be constructed as a waiver of HUD's right to exercise that or any other right or remedy at any time.
- (4) If HUD acquires title to, or possession of, the Project, HUD shall reconvey, or redeliver possession of, the Project to the Landlord and Tenant in accordance with their respective interests in the Project (i) upon a determination by HUD that the substantial default under the ACC has been cured and that the Project will thereafter be operated in accordance with the terms of the ACC; or (ii) after the termination of HUD's obligation to make annual contributions available, unless there are any obligations or covenants of the Landlord to HUD that are then in default.
- (5) During the Term of this Lease, and so long as Tenant shall not be in default of its obligations hereunder, HUD agrees that in the event of a substantial default by Landlord under the ACC, HUD shall exercise any remedies or senctions authorized under the ACC, including taking possession of the Landlord's interest in the Project, in such a manner as not to disturb Tenant's rights under this Lease or the Regulatory Agreement.
- **21.09** Access. Tenant agrees to grant a right of access to the Landlord, HUD, the Comptroller General of the United States, or any of their authorized representatives, with respect to any books, documents, papers, or other records related to this Lease in order to make audits, examinations, excerpts, and transcripts.
- **21.10** <u>Amendment</u>. This Lease may be amended by mutual agreement of the Landlord and Tenant, subject to the prior written approval of HUD, and provided that all amendments must be in writing and signed by both parties and that no amendment shall impair the obligations of the Tenant to develop and operate the Project in accordance with the Applicable Public Housing Requirements.

21.11 Disclaimer of Partnership Status.

(a) Tenant and Landlord acknowledge that the proposed transfer to Tenant, or to any other participating party in the Project, of public housing funds for the development and operation of the Public Housing Units covered under this Lease shall not be deemed to be an assignment of such funds. Accordingly, neither Tenant, nor any other participating party, shall succeed to any rights or benefits of the Landlord under the ACC and the HOPE VI Grant

Agreement. Tenant further agrees to include this disclaimer in each of its agreements or contracts with any partner, participating party, or any other party involving the use of public housing funds for the Project.

(b) Nothing contained in the ACC or the HOPE VI Grant Agreement or in any agreement between Landlord and Tenant, nor any act of HUD or Landlord, shall be deemed or construed to create any relationship of third-party beneficiary, principal and agent, limited or general partnership, or joint venture involving HUD.

21.12 Conflicts.

- (a) In the event of a conflict or inconsistency between any requirement contained in this Lease (or between any requirement contained in any document referred to in this Lease, including any Leasehold Mortgage), and the Applicable Public Housing Requirements, the Applicable Public Housing Requirements shall in all instances be controlling.
- (b) In the event of a conflict between the provisions of this Article 21 and any other provision of this Lease, during the period this Article 21 is in effect, the provision of this Article 21 shall prevail.
- **21.13** <u>Project</u>. For the purposes of this Article 21 and the definition of Applicable Public Housing Requirements in Article 2 only **Project** shall have the same meaning given in the Mixed Finance ACC Amendment.

[Signatures appear on the following pages.]

2301808080 Page: 55 of 99

UNOFFICIAL COPY

IN WITNESS WHEREOF, this Lease is executed as of the date first written above by the duly authorized officers or representatives of the parties hereto.

LANDLORD:

CHICAGO HOÙSING AUTHORITY, an Illinois municipal corporation

Tracey Scott

Chief Executive Officer

C/e/7/5 O/F/CO

COOK COUNTY CLERK OFFICE RECOGNING OWISION 118 N. CLARK ST. ROOM 120 CHICAGO, R. SUGOZ-1327

> COOK COUNTY CLERK OFFICE RECORDING DIVISION 118 N. CLARK ST. ROOM 120 CHICAGO, IL 60602-1327

2301808080 Page: 56 of 99

UNOFFICIAL COPY

STATE OF ILLINOIS)
) SS. COUNTY OF COOK)
I, Mose M. Allen, a Notary Public, in and for said County, in the State aforesaid, DO HEREBY CERTIFY that Tracey Scott, the Chief Executive Officer of the Chicago Housing Authority, an Illinois municipal corporation, who is personally known to me to be the same person whose name is subscribed to the foregoing instrument as such Chief Executive Officer, appeared before me this day in person and acknowledged that he or she signed and delivered said instrument as his own free and voluntary act and as the free and voluntary act of said municipal corporation, for the uses and purposes therein set forth.
GIVEN order my hand and official seal this 6th day of January, 2013.
GIVEN onder my hand and official seal this <u>OM</u> day of <u>January</u> , 2013. Notary Public
for the M. alla
Notary Public
OFFICIAL SEAL ROSE M ALLEN NOTAR: PUBLIC, STATE OF ILLINOIS MY COMMISSION EXPIRES: 5/17/2025
Clert's Office

2301808080 Page: 57 of 99

UNOFFICIAL COPY

TENANT:

HEARTLAND HOUSING, INC., an Illinois not for profit corporation

Toul chuse

Vice President-and Executive Director

COOK COUNTY CLERK OFFICE RECONDACE, DIVISION 113 N. CLARK ST. ROCKI 120 CHICAGO, IL 60602-1387

COCK COUNTY CLERA OFFICE
RECORDING DIVISION
118 N. CLARK ST. ROOM 420
CHICAGO, IL ECSOL 1327

2301808080 Page: 58 of 99

UNOFFICIAL COPY

STATE OF ILLINOIS)
) ss
COUNTY OF COOK)

I, the undersigned, a Notary Public, in and for said County, in the State aforesaid, DO HEREBY CERTIFY that Earl Chase, the Vice President and Executive Director of HEARTLAND HOUSING, INC., an Illinois not for profit corporation, who is personally known to me to be the same person whose name is subscribed to the foregoing instrument as such Vice Presdident and Executive Director, appeared before me this day in person and acknowledged that he/she signed and delivered said instrument as his/her own free and voluntary act and as the free and voluntary act of said not for profit corporation on behalf of Tenant, for the uses and purposes therein set forth.

GIVEN under my hand and official seal this 3'd day of 12 nuary

OFFICIAL SEAL STEPHANIE Y BROWNLEE NOTARY PUBLIC - STATE OF ILLINOIS MY COMMISSION EXPIRES:01/12/25

Office My Commission Expires 12025

EXHIBIT A

LEGAL DESCRIPTION OF REAL ESTATE

LEASEHOLD ESTATE CREATED BY GROUND LEASE DATED AS OF JANUARY 10. 2023, BETWEEN THE CHICAGO HOUSING AUTHORITY, A MUNICIPAL CORPORATION, AS GROUND LESSOR AND HEARTLAND HOUSING, INC., AN ILLINOIS NOT-FOR-PROFIT CORPORATION. AS GROUND LESSEE; ASSUMPTION AND ASSIGNMENT OF GROUND LEASE MADE BY HEARTLAND HOUSING, INC., AN ILLINOIS NOT-FOR-PROFIT CORPORATION, AS ASSIGNOR, RS AFFORDABLE 1 LLC, AN ILLINOIS LIMITED LIABILITY COMPANY, AS ASSIGNEE, AND THE CHICAGO HOUSING AUTHORITY, A MUNICIPAL CORPORATION, DATED AS OF JANUARY 10, 2023: DEMISING THE FOLLOWING DESCRIBED LAND FOR A TERM OF 82 YEARS:

TRACT 1:

THAT PART OF BLOCK 15 IN VERNON PARK ADDITION TO CHICAGO, BEING AN ANTE FIRE SUBDIVISION OF BLOCKS 38, 39, 44 & 45 IN CANAL TRUSTEE'S SUBDIVISION OF THE WEST HALF OF THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHWEST CORNER OF BLOCK 15 IN VERNON PARK ADDITION TO CHICAGO AFORESAID: THENCE SOUTH 01°49'49" EAST ALONG THE WEST LINE OF BLOCK 15 A DISTANCE OF 111.20 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 01°49'49" EAST ALONG THE WEST LINE OF BLOCK 15 A DISTANCE OF 39.69 FEET; THENCE NORTH 88°10'11" EAST 17.08 FEET; THENCE SOUTH 01°49'49" EAST 20.39 FEET; THENCE NORTH 88°10'11" EAST 27.21 FEET; THENCE NORTH 01°49'49" WEST 20.23 FEET; THENCE NORTH 88°10'11" EAST 30.70 FEET; THENCE SOUTH 01°49'49" EAST 18.16 FEET; THENCE NORTH 88°10'11" EAST 26.95 FEET, THENCE NORTH 01°49'49" WEST 18.09 FEET; THENCE NORTH 88°10'11" EAST 13.24 FEET; THENCE NORTH 01°49'49" WEST 39.93 FEET; THENCE SOUTH 88°10'11" WEST 115.18 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS. 7//_C

For informational purposes only: Commonly known as 907-909 South Ada Street, Chicago, IL 60607; PIN No. 17-17-320-002-0000

R	REAL ESTATE TRANSFER TAX			12-Jan-2023
		The state of the s	COUNTY	0.00
		S04.	ILLINOIS:	0.00
			TOTAL:	0.00
	17-17-320	-002-000C	120230101628678	-910-027-088

REAL ESTATE TRA	NSFER TAX	12-Jan-2023
	CHICAGO: ,	0.00
	CTA:	0.00
	TOTAL:	0.00 1

17-17-320-002-0000 | 20230101628678 | 0-836-285-264

^{*} Total does not include any applicable penalty or interest due.

TRACT 2:

LIHTC PARCEL (REMAINDER)

LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, EXCEPT THEREFROM THE FOLLOWING DESCRIBED PARCELS:

PARCEL L1-1

THAT PART OF LOT 16 IN PLAT I ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SQUITHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEZATION OF +31.08 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +13.75 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID: THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 7.08 FEET; THENC! SOUTH 88°23'27" WEST 17.46 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 0. 35'33" EAST 30.86 FEET; THENCE SOUTH 88°23'27" WEST 25.23 FEET; THENCE NORTH U 36'33" WEST 9.45 FEET; THENCE SOUTH 84°15'03" WEST 21.12 FEET; THENCE SOUTH 05° 4'57" EAST 3.45 FEET; THENCE SOUTH 84°15'03" WEST 6.00 FEET; THENCE SOUTH 01°36'33" LAST 6.90 FEET; THENCE SOUTH 88°23'27" WEST 17.47 FEET; THENCE SOUTH 01°36'33" EAST 16.65 FEET; THENCE SOUTH 88°23'27" WEST 33.94 FEET; THENCE NORTH 01°36'33" WEST 4.67 FEET; THENCE SOUTH 88°23'27" WEST 5.66 FEET; THENCE NORTH 01°36'33" WEST 5.34 FEET; THENCE SOUTH 88°23'27" WEST 40.56 FEET; THENCE SOUTH 01°36'33" EAST 4.82 FEET; THENCE SOUTH 88°23'27" WEST 7.48 FEET; THENCE SOUTH 01°36'33" EAST 4.60 F LET, THENCE SOUTH 84°13'18" WEST 20.58 FEET; THENCE SOUTH 01°36'33" EAST 9.14 FEET: THENCE SOUTH 88°23'27" WEST 67.60 FEET; THENCE NORTH 01°36'33" WEST 58.41 FEET; THENCE NORTH 88°23'27" EAST 68.20 FEET; THENCE SOUTH 86°15'43" EAST 78.12 I/EET; THENCE NORTH 84°15'03" EAST 73.14 FEET; THENCE NORTH 88°23'27" EAST 26.34 FRET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L2-1

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +31.08 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 6.98 FEET; THENCE SOUTH 88°23'27" WEST 6.81 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°36'33" EAST 48.43 FEET; THENCE SOUTH 88°23'27" WEST 30.14 FEET; THENCE NORTH

88°23'27" EAST 6.32 FEET; THENCE NORTH 01°36'33" WEST 6.50 FEET; THENCE SOUTH 88°23'27" WEST 6.10 FEET; THENCE SOUTH 01°36'33" EAST 0.67 FEET; THENCE SOUTH 88°23'27" WEST 2.93 FEET; THENCE NORTH 01°36'33" WEST 4.77 FEET; THENCE SOUTH 88°23'27" WEST 3.21 FEET; THENCE NORTH 01°36'33" WEST 2.77 FEET; THENCE SOUTH 84°15'03" WEST 20.74 FEET; THENCE SOUTH 05°44'57" EAST 8.85 FEET; THENCE SOUTH 84°15'03" WEST 2.43 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 29.31 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 14.73 FEET: THENCE NORTH 05°44'57" WEST 0.66 FEET: THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 13.92 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WESP 5 66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 35.40 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FLET: THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 9.25 FLET; THENCE SOUTH 88°23'27" WEST 6.85 FEET; THENCE NORTH 01°36'33" WEST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 5.66 FEET; THENCE SOUTH 01°36'33" EAST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 6.57 FEET; THENCE NORTH 01°36'33" WEST 26.97 FEET; THENCE NORTH 88°23'27" EAST 19.20 FEET; THENCE NORTH 84°15'03" EAST 153.32 FFFT; THENCE NORTH 01°36'33" WEST 3.63 FEET; THENCE NORTH 88°23'27" EAST 36.0. SEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT

PARCEL L2-2

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQ'JARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOV. NSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELDW A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +31.08 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 78.99 FEET; THENCE SOUTH 88°23'27" WEST 7.01 FEET TO 1/4F POINT OF BEGINNING; THENCE SOUTH 01°36'33" EAST 26.53 FEET; THENCE SOUTH 88°23'27" WEST 36.04 FEET; THENCE NORTH 01°36'33" WEST 26.53 FEET; THENCE NORTH 88°23'27" EAST 36.04 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS; ALSO EXCEPT

PARCEL L2-3

THAT PART OF LOT 16 IN PLAT I ROOSEVELT SQUARE SUBDIVISION. IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142. LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +31.08 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED

VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID: THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 73.68 FEET; THENCE SOUTH 88°23'27" WEST 90.41 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 84°15'03" WEST 105.88 FEET; THENCE SOUTH 88°23'27" WEST 19.16 FEET; THENCE NORTH 01°36'33" WEST 26.97 FEET; THENCE NORTH 88°23'27" EAST 6.49 FEET; THENCE SOUTH 01°36'33" EAST 0.66 FEET; THENCE NORTH 88°23'27" EAST 5.66 FEET; THENCE NORTH 01°36'33" WEST 0.66 FEET; THENCE NORTH 88°23'27" EAST 7.89 FEET; THENCE SOUTH 01°36'33" EAST 8.82 FEET; THENCE NORTH 84°15'03" EAST 20.65 FEET; THENCE NORTH 05°44'57" WEST 8.85 FEET; THENCE NORTH 84°15'03" £AST 2.60 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 26.19 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3 49 FEET; THENCE SOUTH 05°44'57" EAST 5.55 FEET; THENCE NORTH 84°15'03" EAST 21.53 FEET; THENCE NORTH 05°44'57" WEST 5.55 FEET; THENCE NORTH 84°15'03" EAST 3.44 FLET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET THENCE NORTH 05°44'57" WEST 0.66 FEET: THENCE NORTH 84°15'03" EAST 8.78 FEET; THENCE SOUTH 05°44'57" EAST 26.89 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L2-4

THAT PART OF LOT 16 IN PLAT 1 ROOSLVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACC OF DING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CAICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +31.08 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 88°23'27" WLST ALONG THE NORTH LINE THEREOF 233.94 FEET; THENCE SOUTH 01°36'33" EAST 42.15 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°36'33" EAST 12.31 FEET; THENCE SOUTH 88°23'27" WEST 5.55 FEET; THENCE SOUTH 01°36'33" WEST 31.94 FEET; THENCE NORTH 88°23'27" EAST 25.16 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS; ALSO EXCEPT

PARCEL L3-1

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE

THEREOF 6.98 FEET; THENCE SOUTH 88°23'27" WEST 6.81 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°36'33" EAST 72.01 FEET; THENCE SOUTH 88°23'27" WEST 30.14 FEET; THENCE NORTH 01°36'33" WEST 10.90 FEET; THENCE NORTH 88°23'27" EAST 0.66 FEET; THENCE NORTH 01°36'33" WEST 5.66 FEET; THENCE SOUTH 88°23'27" WEST 0.66 FEET; THENCE NORTH 01°36'33" WEST 20.36 FEET; THENCE NORTH 88°23'27" EAST 6.32 FEET; THENCE NORTH 01°36'33" WEST 6.50 FEET; THENCE SOUTH 88°23'27" WEST 6.10 FEET; THENCE SOUTH 01°36'33" WEST 0.67 FEET; THENCE SOUTH 88°23'27" WEST 2.93 FEET; THENCE NORTH 01°36'33" WEST 4.77 FEET; THENCE SOUTH 88°23'27" WEST 3.21 FEET; THENCE NORTH 01°36'33" WEST 24.49 FEET; THENCE NORTH 88°23'27" FAST 36.06 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS; ALSO EXCEPT

PARCEL L3-2

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 81.32 FEET; THENCE SOUT A 88°23'27" WEST 43.08 FEET; THENCE NORTH 01°36'33" WEST 11.07 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 84°15'03" WEST 47.48 FEET; THENCE NORTH 05°44'57" WEST 20.62 FEET; THENCE NORTH 84°15'03" EAST 13.32 FEET; THENCE NORTH 05 14 57" WEST 6.27 FEET; THENCE NORTH 84°15'03" EAST 7.77 FEET; THENCE SOUTH 05°44'5/" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3.38 FEET; THENCE SOUTH 05°44'57" EAST 10.28 FEET; THENCE NORTH 84°15'03" EAST 18.55 FEET; THENCE SOUTH 01°36'33" EAST 16.65 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L3-3

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 81.32 FEET; THENCE SOUTH 88°23'27" WEST 196.04 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°23'27" WEST 19.16 FEET; THENCE NORTH 01°36'33" WEST 26.97 FEET; THENCE NORTH 88°23'27" EAST 5.66 FEET; THENCE SOUTH 01°36'33" WEST 0.66 FEET; THENCE NORTH 88°23'27" EAST 7.89 FEET; THENCE SOUTH 01°36'33" WEST 0.66 FEET; THENCE NORTH 88°23'27" EAST 7.89 FEET; THENCE SOUTH 01°36'33"

EAST 8.82 FEET; THENCE NORTH 84°15'03" EAST 20.65 FEET; THENCE NORTH 05°44'57" WEST 8.85 FEET; THENCE NORTH 84°15'03" EAST 2.60 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 26.19 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3.49 FEET; THENCE SOUTH 05°44'57" EAST 5.55 FEET; THENCE NORTH 84°15'03" EAST 10.79 FEET; THENCE SOUTH 05°44'57" EAST 21.34 FEET; THENCE SOUTH 84°15'03" WEST 77.24 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L3-4

THAT PAR? OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUT FIWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD I RINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID, T.JENCE SOUTH 88°23'27" WEST ALONG THE NORTH LINE THEREOF 90.36 FEET; THENCE SOUTH 01°36'33" EAST 14.05 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 05°44'57" EAST 26.89 FEET; THENCE SOUTH 84°15'03" WEST 11.81 FEET; THENCE NORTH 05°44', 7" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 14.73 FEET; THENCE NORTH 05°44'57" WEST 2 66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 6.64 FEET; THENCE NORTH 05°44'57" WEST 26 89 FEET; THENCE NORTH 84°15'03" EAST 44.50 FEET TO THE POINT OF BEGINNING. IN COCK COUNTY, ILLINOIS:

ALSO EXCEPT

PARCEL L3-5

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST OUARTER OF SECTION 17. TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +41.75 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 88°23'27" WEST ALONG THE NORTH LINE THEREOF 159.16 FEET: THENCE SOUTH 01°36'33" EAST 19.03 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 05°44'57" EAST 26.89 FEET; THENCE SOUTH 84°15'03" WEST 23.86 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 9.25 FEET: THENCE SOUTH 88°23'27" WEST 6.85 FEET: THENCE NORTH 01°36'33" WEST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 5.66 FEET; THENCE SOUTH 01°36'33" EAST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 18.35 FEET; THENCE NORTH 01°36'33"

WEST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 7.20 FEET; THENCE SOUTH 01°36'33" EAST 6.37 FEET; THENCE SOUTH 88°23'27" WEST 5.55 FEET; THENCE SOUTH 01°36'33" EAST 19.64 FEET; THENCE SOUTH 88°23'27" WEST 19.61 FEET; THENCE NORTH 01°36'33" WEST 52.31 FEET; THENCE NORTH 88°23'27" EAST 63.35 FEET; THENCE NORTH 84°15'03" EAST 36.69 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L4-1

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORES AID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 10.61 FEET; THENCF SOUTH 88°23'27" WEST 42.88 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01 36'33" EAST 18.09 FEET; THENCE SOUTH 84°15'03" WEST 20.74 FEET; THENCE SOUT (1)5°44'57" EAST 8.85 FEET; THENCE SOUTH 84°15'03" WEST 2.43 FEET; THENCE NORTH 0.5°4/57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 17.51 FEET: THENCE NORTH 05 44 57" WEST 26.89 FEET; THENCE NORTH 84°15'03" EAST 47.64 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS; ALSO EXCEPT

PARCEL L4-2

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARL SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNS 11P 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A LIAT THEREOF RECORDED MAY 27, 2004. AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +57.92 FEET ABOVE, CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUND ARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 88°23'27" WEST ALONG THE NORTH LINE THEREOF 114.78 FEET; THENCE SOUTH 01°36'33" EAST 15.82 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 05°44'57" EAST 26.89 FEET; THENCE SOUTH 84°15'03" WEST 7.72 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 13.92 FEET: THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 11.54 FEET: THENCE NORTH 05°44'57" WEST 26.89 FEET; THENCE NORTH 84°15'03" EAST 44.50 FEET TO THE POINT OF BEGINNING. IN COOK COUNTY, ILLINOIS: ALSO EXCEPT

PARCEL L4-3

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 88°23'27" WEST ALONG THE NORTH LINE THEREOF 194.96 FEET; THENCE SOUTH 01°36'33" EAST 21.68 FEET TO THE POINT OF BECANNING; THENCE SOUTH 01°36'33" EAST 26.97 FEET; THENCE SOUTH 88°23'27" WEST 7.77 FEET; THENCE NORTH 01°36'33" WEST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 5.66 FECT, THENCE SOUTH 01°36'33" EAST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 18.35 FEET, THENCE NORTH 01°36'33" WEST 0.66 FEET; THENCE SOUTH 88°23'27" WEST 7.20 FEET; THENCE NORTH 01°36'33" WEST 5.94 FEET; THENCE SOUTH 88°23'27" WEST 25.16 FEET; THENCE NORTH 01°36'33" WEST 20.37 FEET; THENCE NORTH 88°23'27" EAST 64.14 FEE" 1'O THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS: ALSO EXCEPT

PARCEL L4-4

THAT PART OF LOT 16 IN PLAT I RODSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN LIEVATION OF +52.92 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HOUZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" LAST ALONG THE EAST LINE THEREOF 73.68 FEET; THENCE SOUTH 88°23'27" WEST 90.41 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 84°15'03" WEST 57.30 FEET; THENCE NORTH 05°44'57" WEST 26.89 FEET; THENCE NORTH 84°15'03" EAST 8.73 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3.49 FEET; THENCE SOUTH 05°44'57" EAST 5.55 FEET; THENCE NORTH 84°15'03" EAST 21.55 FEET; THENCE NORTH 84°15'7" WEST 5.55 FEET; THENCE NORTH 84°15'03" EAST 3.43 FEET; THENCE SOUT J 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 0.5°14'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 8.78 FEET; THENCE SOUTH 05°44'57" EAST 26.89 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L4-5

THAT PART OF LOT 16 IN PLAT I ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN. ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +52.92 FEET ABOVE

CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46′13″ EAST ALONG THE EAST LINE THEREOF 81.32 FEET; THENCE SOUTH 88°23′27″ WEST 195.15 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°23′27″ WEST 44.57 FEET; THENCE NORTH 01°36′33″ WEST 26.97 FEET; THENCE NORTH 88°23′27″ EAST 5.55 FEET; THENCE SOUTH 01°36′33″ EAST 0.66 FEET; THENCE NORTH 88°23′27″ EAST 7.20 FEET; THENCE NORTH 01°36′33″ EAST 0.66 FEET; THENCE NORTH 88°23′27″ EAST 18.26 FEET; THENCE SOUTH 01°36′33″ EAST 0.66 FEET; THENCE NORTH 88°23′27″ EAST 5.66 FEET; THENCE NORTH 01°36′33″ EAST 0.66 FEET; THENCE NORTH 88°23′27″ EAST 7.89 FEET; THENCE SOUTH 01°36′33″ EAST 26.97 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS; ALSO EXCLPT

PARCEL L5-1

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.25 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS JOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 10.61 FEET; THENCE SOUTH & 8°23'27" WEST 42.88 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°36'33" EAST 18.09 FEET; THENCE SOUTH 84°15'03" WEST 20.74 FEET; THENCE SOUTH 05°44'57" EAST 8 85 FEET; THENCE SOUTH 84°15'03" WEST 2.43 FEET; THENCE NORTH 05°44'57" WEST 4.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0 65 FEET; THENCE SOUTH 84°15'03" WEST 29.31 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEE1; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET: THENCE SOUTH 84°15'03" WEST 14.73 FEET: THENCE NORTH 05°44'57" WEST 0.66 FEET, TAFNCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 13.92 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 11.54 FEET; THENCE NORTH 05°44'57" WEST 26.89 FEET; THENCE NORTH 84°15'03" EAST 116.62 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, H. LINOIS: ALSO EXCEPT

PARCEL L5-2

THAT PART OF LOT 16 IN PLAT I ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.25 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE

THEREOF 70.25 FEET; THENCE SOUTH 88°23'27" WEST 43.05 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 84°15'03" WEST 76.12 FEET; THENCE NORTH 05°44'57" WEST 21.34 FEET; THENCE NORTH 84°15'03" EAST 10.77 FEET; THENCE NORTH 05°44'57" WEST 5.55 FEET; THENCE NORTH 84°15'03" EAST 3.43 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 13.32 FEET; THENCE NORTH 05°44'57" WEST 6.27 FEET; THENCE NORTH 84°15'03" EAST 13.32 FEET; THENCE NORTH 05°44'57" WEST 6.27 FEET; THENCE NORTH 84°15'03" EAST 7.77 FEET; THENCE SOUTH 05°44'57" VEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3.38 FEET; THENCE NORTH 05°44'57" VEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3.38 FEET; THENCE SOUTH 01°36'33" EAST 10.28 FEET; THENCE NORTH 84°15'03" EAST 18.55 FEET; THENCE SOUTH 01°36'33" EAST 16.65 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L5-3

THAT PART OF LOT 16 IN PLAT I ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MEPIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74 2' FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +63.58 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 77.82 FEET; THENCE SOUTH 88°23'27' WEST 147.58 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 84°15'03" WEST 47.5° FEET; THENCE NORTH 01°36'33" WEST 18.09 FEET; THENCE NORTH 84°15'03" E4.37 20.65 FEET; THENCE NORTH 05°44'57" WEST 8.85 FEET; THENCE NORTH 84°15'03" E/ST 2.60 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5 66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 17.45 FEET; THENCE SOUTH 05°44'57" EAST 26.89 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS: ALSO EXCEPT

PARCEL L6-1

THAT PART OF LOT 16 IN PLAT 1 ROOSEVELT SQUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF REGORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +85.42 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.25 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID; THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 10.61 FEET; THENCE SOUTH 88°23'27" WEST 42.88 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°36'33" EAST 18.09 FEET; THENCE SOUTH 84°15'03" WEST 20.74 FEET; THENCE SOUTH 05°44'57" EAST 8.85 FEET; THENCE SOUTH 84°15'03" WEST 2.43 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03"

WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 17.51 FEET; THENCE NORTH 05°44'57" WEST 26.89 FEET; THENCE NORTH 84°15'03" EAST 47.64 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS; ALSO EXCEPT

PARCEL L6-2

THAT PART OF LOT 16 IN PLAT I ROOSEVELT SOUARE SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +85.42 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.25 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT IT AFORESAID; THENCE SOUTH 88°23'27" WEST ALONG THE NORTH LINE THEREOF 114.78 FEET; THENCE SOUTH 01°36'33" EAST 15.82 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 05°44'57" EAST 26.89 FEET; THENCE SOUTH 84°15'03" WEST 7.72 FEET; THENCI, NORTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 13.92 FEET; THENCE NO RTH 05°44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 5.66 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 34.23 FEET; THENCE NORTH 05' 44'57" WEST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 6.83 FEET; THENCE SOUTH 05°44'57" FAST 0.66 FEET; THENCE SOUTH 84°15'03" WEST 8.32 FEET; THENCE NORTH 01°36'3J" WEST 26.96 FEET; THENCE NORTH 84°15'03" EAST 80.39 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS ALSO EXCEPT

PARCEL L6-3

THAT PART OF LOT 16 IN PLAT I ROOSEVELT SQUARL'SUBDIVISION, IN THE EAST HALF OF THE SOUTHWEST QUARTER OF SECTION 17, TOWNS IIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED MAY 27, 2004, AS DOCUMENT NO. 0414831142, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +85.42 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.25 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF LOT 16 AFORESAID: THENCE SOUTH 01°46'13" EAST ALONG THE EAST LINE THEREOF 70.25 FEET; THENCE SOUTH 88°23'27" WEST 43.05 FEET TO THE FOULT OF BEGINNING; THENCE SOUTH 84°15'03" WEST 104.79 FEET; THENCE NORTH 05°44'57" WEST 26.89 FEET; THENCE NORTH 84°15'03" EAST 8.73 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3.49 FEET; THENCE SOUTH 05°44'57" EAST 5.55 FEET; THENCE NORTH 84°15'03" EAST 21.55 FEET; THENCE NORTH 05°44'57" WEST 5.55 FEET; THENCE NORTH 84°15'03" EAST 3.43 FEET; THENCE SOUTH 05°44'57" EAST 0.66 FEET: THENCE NORTH 84°15'03" EAST 5.66 FEET: THENCE NORTH 05°44'57" WEST 0.66 FEET: THENCE NORTH 84°15'03" EAST 8.78 FEET; THENCE SOUTH 05°44'57" EAST 6.27 FEET; THENCE NORTH 84°15'03" EAST 13.32 FEET; THENCE NORTH 05°44'57" WEST 6.27 FEET; THENCE NORTH 84°15'03" EAST 7.77 FEET; THENCE SOUTH 05°44'57"

EAST 0.66 FEET; THENCE NORTH 84°15'03" EAST 5.66 FEET; THENCE NORTH 05°44'57" WEST 0.66 FEET; THENCE NORTH 84°15'03" EAST 3.38 FEET; THENCE SOUTH 05°44'57" EAST 10.28 FEET; THENCE NORTH 84°15'03" EAST 18.55 FEET; THENCE SOUTH 01°36'33" EAST 16.65 FEET TO THE POINT OF BEGINNING), IN COOK COUNTY, ILLINOIS.

For informational purposes only:

Commonly known as 1002 South Racine Avenue, Chicago, IL 60607;

PIN No. 17-17-334-006-0000

TRACT 3:

LIHTC PARCEL (REMAINDER)

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33. BOTH INCLUSIVE. TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24. AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO. AN ANTE FIRE SUBMY ISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT AND DESCRIBED AS FOLLOWS: BEGINNING AT THE NORTHEAST CORNER OF SAID LOT 16; THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF LOTS 16 AND 33, AND A LINE DRAWN FROM THE SOUTHEAST CORNER OF LOT 16 TO THE NORTHEAST CORNER OF LOT 33 % DISTANCE OF 109.78 FEET TO A POINT ON THE NORTH LINE OF AN 18 FOOT PUBLIC ALLEY DEDICATED PER ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO RECORDED JULY 20, 2007 AS DOCUMENT NO. 0720115116; THENCE SOUTH 88°22'20" WEST ALONG THE NORTH LINE OF THE 18 FOOT PUBLIC ALLEY AFORESAID 224.54 FEET TO A POINT ON THE EAST LINE OF S. THROOP STREET; THENCE NORTH 01°37'40" VEST ALONG THE EAST LINE OF S. THROOP STREET 109.81 FEET TO A POINT ON THE SOUTH LINE OF W. ROOSEVELT ROAD AS WIDENED; THENCE NORTH 88°22'51" EAST ALONG THE SOUTH LINE OF W. ROOSEVELT ROAD AS WIDENED 224.54 FEET TO THE POINT OF BEGINNING, EXCEPT THEREFROM THE FOLLOWING DESCRIBED PARCELS:

PARCEL L2-1

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOU'. H C F THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN 31-OCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 2.86 FEET: THENCE SOUTH 88°22'51" WEST 7.04 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 22.39 FEET; THENCE SOUTH 88°22'51" WEST 15,21 FEET; THENCE SOUTH 01°37'09" EAST 5.59 FEET; THENCE SOUTH 88°22'51" WEST 6.93 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 12.43 FEET; THENCE NORTH 01°37'09" WEST 4.39 FEET; THENCE SOUTH

88°22'51" WEST 8.31 FEET; THENCE NORTH 01°37'09" WEST 24.00 FEET; THENCE NORTH 88°22'51" EAST 42.88 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L2-2

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WATTA ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG T IF NORTH LINE OF SAID TRACT 99.50 FEET; THENCE SOUTH 01°37'09" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 28.39 FEET: THENCE SOUTH 88°22'51" WFST 11.36 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6 JZ FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 14.78 FEET: THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 10.61 FEET; THENCE NORTH 01°37 PJ" WEST 2.70 FEET; THENCE SOUTH 88°22'51" WEST 0.79 FEET; THENCE NORTH 01°37'09" WEST 25.49 FEET; THENCE NORTH 88°22'51" EAST 49.58 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT PARCEL L2-3

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALL IT LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2135719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FILE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 11 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88'22'51" WEST 122.14 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 88°22'51" WEST 24.48 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 7.06 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 11.47 FEET; THENCE SOUTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT PARCEL L2-4

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE. TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33,

VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011. ALI. IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.92 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 19.24 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 6.95 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.59 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.59 FEET; THENCE NORTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L3-1

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE LAST WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECURDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.96 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION O + 34.92 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PPOJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST COPNER OF SAID TRACT: THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 2.86 FEET; THENCE SOUTH 88°22'51" WEST 74.74 FEET TO THE POINT OF BEGINNING; THENCE SCUTH 01°37'09" EAST 28.39 FEET; THENCE SOUTH 88°22'51" WEST 7.35 FEET; THENCE NORTH 0.1°37'/9" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09" EAST 1.40 FEET; THENCE SOUTH 88°22'51" WEST 11.40 FEET; THENCE NORTH 01°37'09" WEST 28.39 FEET; THENCE NORTH 88°22'51" EAST 24.77 FEET TO THE POINT OF BEGINNING, IN COOK CCUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L3-2

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE POGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011. ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.96 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.92 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST 97.08 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" EAST 11.20 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET: THENCE NORTH 88°22'51" EAST 11.20 FEET;

THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.18 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 7.68 FEET; THENCE SOUTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT

PARCEL L4-1

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33. VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.98 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.96 FEET ABOVE CHICAGO CITY DATUM. AND LYING WITHIN 'AS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 2.86 FEET; THENCE SOUTH 88°22'51" WEST 7.04 FEET TO THE POLY, OF BEGINNING; THENCE SOUTH 01°37'09" EAST 22.39 FEET; THENCE SOUTH 88°22'51" WEST 15.21 FEET; THENCE SOUTH 01°37'09" EAST 5.59 FEET; THENCE SOUTH 88°22'51" WEST 6.93 FEET: THENCE SOUTH 01°37'09" EAST 0.40 FEET: THENCE SOUTH 88°22'51" WEST 12.43 FEET; THENCL NORTH 01°37'09" WEST 4.39 FEET; THENCE SOUTH 88°22'51" WEST 8.31 FEET; THENCE NOXTH 01°37'09" WEST 24.00 FEET; THENCE NORTH 88°22'51" EAST 42.88 FEET TO THE POINT OF PEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L4-2

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST LE FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33. VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL JAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.98 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.96 FEET ABOVE CHICAGO CITY DATUM, AND LYING YJTHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 74.74 FEET; THENCE SOUTH 01°37'09" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 28.39 FEET; THENCE SOUTH 88°22'51" WEST 7.35 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET: THENCE SOUTH 88°22'51" WEST 6.02 FEET: THENCE SOUTH 01°37'09" EAST 0.40 FEET: THENCE SOUTH 88°22'51" WEST 11.40 FEET: THENCE NORTH 01°37'09" WEST 28.39 FEET: THENCE NORTH 88°22'51" EAST 24.77 FEET TO THE POINT OF BEGINNING. IN COOK COUNTY. ILLINOIS:

ALSO EXCEPT PARCEL L4-3

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33. VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.98 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.96 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 88°22'51' WEST ALONG THE NORTH LINE OF SAID TRACT 124.28 FEET; THENCE SOUTH 01°37'09" EAST 2.86 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 01°37'09" EAST 28.39 FEET; THENCE SOUTH 88°22'51" WEST 7.39 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 68°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 19.61 FEET; THENCE NORTH 01°37'09" WEST 2.70 FEET; THENCE SOUTH 88°22'51" WEST 0.79 FEET; THENCE NORTH 01°37'09" WEST 25.69 FEET; THENCE NORTH 88°22'51" EAST 24.81 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT PARCEL L4-4

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST IC FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS LOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING 'JFLOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.98 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.96 FEET AROVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 202./2 FEET; THENCE SOUTH 01°37'09" EAST 30.85 FEET TO THE POINT OF BEGINNING: THENCE SCUT 4' 01°37'09" EAST 9.42 FEET; THENCE NORTH 88°22'51" EAST 7.80 FEET; THENCE SOUTH 01°57'09" EAST 6.07 FEET; THENCE NORTH 88°22'51" EAST 9.61 FEET; THENCE SOUTH 01°37'09" EAST 18 90 LEET; THENCE SOUTH 88°22'51" WEST 36.63 FEET; THENCE NORTH 01°37'09" WEST 34.38 FLFT; THENCE NORTH 88°22'51" EAST 19.23 FEET TO THE POINT OF BEGINNING, IN COOK COUNT, 'ULINOIS;

ALSO EXCEPT

PARCEL L5-1

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.98 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED

AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 2.86 FEET; THENCE SOUTH 88°22'51" WEST 7.04 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 01°37'09" EAST 22.39 FEET; THENCE SOUTH 88°22'51" WEST 15.21 FEET; THENCE SOUTH 01°37'09" EAST 5.59 FEET; THENCE SOUTH 88°22'51" WEST 6.93 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 12.43 FEET; THENCE NORTH 01°37'09" WEST 4.39 FEET; THENCE SOUTH 88°22'51" WEST 8.31 FEET; THENCE NORTH 01°37'09" WEST 24.00 FEET; THENCE NORTH 88°22'51" EAST 42.88 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L5-2

LOTS 16 10 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORD'NANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FLET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN LILEVATION OF +54.98 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 149.09 FEET; THENCE SOUTH 01°37'40" EAST 2.86 FEET TO THE POINT OF BFGINNING: THENCE SOUTH 01°37'09" EAST 25.69 FEET; THENCE NORTH 88°22'51" EAST 0.79 LEET; THENCE SOUTH 01°37'09" EAST 2.70 FEET; THENCE SOUTH 88°22'51" WEST 12.52 FEET; THENCE NORTH 01°37'09" WEST 2.25 FEET; THENCE SOUTH 88°22'51" WEST 8.92 FEET; THENC £ SOUTH 01°37'09" EAST 1.85 FEET; THENCE SOUTH 88°22'51" WEST 5.70 FEET; THENCE NORTH 01°3' '('9)" WEST 10.46 FEET; THENCE SOUTH 88°22'51" WEST 21.97 FEET; THENCE NORTH 01°37'05" WEST 17.53 FEET; THENCE NORTH 88°22'51" EAST 48.32 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L5-3

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL P. PLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.98 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 122.14 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 24.48 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET: THENCE NORTH 88°22'51" EAST 7.06 FEET: THENCE SOUTH 01°37'09" EAST 0.40 FEET: THENCE NORTH 88°22'51" EAST 5.95 FEET: THENCE NORTH 01°37'09" WEST 0.40 FEET: THENCE NORTH 88°22'51" EAST 11.47 FEET; THENCE SOUTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L6-1

LOTS 16 TO 24. TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33. VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FEET ABOVE CHICAGO CITY DATUM. AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 49.92 FEET; THENCE SOUTH 01°37'40" EAST 2.86 FEEL TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 24.00 FEET: THENCE SOUTH 88°22'51" WEST 0.46 FEET; THENCE SOUTH 01°37'09" EAST 4.39 FEET; THENCE SOUTH 88°22'51" WEST 1', 90 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FELT; TI:ENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 7.44 FEET; THENCE NORTH 88°22'51" EAST 24.82 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L6-2

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOCT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NOKT/L RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FEET ALOUE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERSY ALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 99.50 FEET; THENCE SOUTH 01°37'40" EAST 2.86 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 01°37'09" EAST 28.39 FEET: THENCE SOUTH 88°22'51" WEST 11.36 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET: THENCE SOUTH 01°37'09" EAST 0.40 FEET: THENCE SOUTH 88°22'51" WEST 7.39 FEET; THENCE NORTH 01°37'09" WEST 28.39 FEET; THENCE NORTH 88°22'51" EAST 24.77 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS,

ALSO EXCEPT PARCEL L6-3

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A

HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 149.09 FEET; THENCE SOUTH 01°37'40" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 2.70 FEET; THENCE NORTH 88°22'51" WEST 12.52 FEET; THENCE SOUTH 01°37'09" WEST 2.25 FEET; THENCE SOUTH 88°22'51" WEST 8.92 FEET; THENCE SOUTH 01°37'09" EAST 1.85 FEET; THENCE SOUTH 88°22'51" WEST 5.70 FEET; THENCE NORTH 01°37'09" WEST 10.46 FEET; THENCE SOUTH 88°22'51" WEST 18.87 FEET; THENCE SOUTH 01°37'09" EAST 10.46 FEET; THENCE SOUTH 88°22'51" WEST 27.34 FEET; THENCE NORTH 01°37'09" WEST 27.99 FEET; THENCE NORTH 88°22'51' EAST 72.56 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L6-4

LOTS 16 TO 24. TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE FAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE PLCORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20 TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVAT ON OF +65.00 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUN DARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TVACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 122.14 FEET TO THE POINT OF BEGINNING; THEY.CE SOUTH 88°22'51" WEST 24.48 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 7.06 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88° 22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 11.47 FEET; THENCE SOUTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COCASCOUNTY, ILLINOIS;

ALSO EXCEPT

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCL ISIVE. TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH CFT HE SOUTH LINES OF SAID LOTS 16 TO 24, AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20. TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FEET ABOVE CHICAGO CITY DATUM. AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 80°22'51" WEST 77.84 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 19.24 FEET: THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 6.95 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET: THENCE NORTH 88°22'51" EAST 5.59 FEET; THENCE

NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT

PARCEL L6-6

LOTS 16 TO 24, TOGETHER WITH THAT PART OF LOTS 25 TO 33, BOTH INCLUSIVE, TOGETHER WITH THAT PART OF THE EAST-WEST 16 FOOT VACATED ALLEY LYING SOUTH OF THE SOUTH LINES OF SAID LOTS 16 TO 24. AND NORTH OF THE NORTH LINE OF SAID LOT 25 TO 33, VACATED BY ORDINANCE RECORDED AS DOCUMENT NO. 2133719011, ALL IN BLOCK 1 IN SAMPSON & GREENE'S ADDITION TO CHICAGO, AN ANTE FIRE SUBDIVISION OF THE NORTHWEST 1/4 OF SECTION 20. TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIFAL MERIDIAN, ALL TAKEN AS A TRACT, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +65.00 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITTING ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMUNICING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 01°37'40" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET: THENCE SOUTH 88°22'51" WEST 7.05 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 46.33 FEET; THENCE NORTH 01°37'09" V. F.J. 18.50 FEET; THENCE NORTH 88°22'51" EAST 25.56 FEET; THENCE NORTH 01°37'09" WEST 9.51 FEET; THENCE NORTH 88°22'51" EAST 5.57 FEET; THENCE NORTH 01°37'09" WEST 11.97 FEET: THENCE NORTH 88°22'51" EAST 13.65 FEET: THENCE SOUTH 01°37'09" EAST 11.30 FEET; THENCE NORTH 88°22'51" EAST 1.56 FEET; THENCE SOUTH 01°37'09" EAST 28.68 FEET TO THE POINT OF BECLIVING), IN COOK COUNTY, ILLINOIS.

For informational purposes only:

Commonly known as 1257 West Roosevelt Road, Chicago, 1L 60608;

PIN No. 17-20-103-001-0000

17-20-103-002-0000

17-20-103-003-0000

17-20-103-004-0000

17-20-103-005-0000

17-20-103-006-0000

17-20-103-007-0000

17-20-103-008-0000

17-20-103-063-0000

TRACT 4: LIHTC PARCEL

160k, LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO FEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20. TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007. AS DOCUMENT NO. 0720115116, EXCEPT THEREFROM THE FOLLOWING DESCRIBED PARCELS:

PARCEL L2-1

THAT PART OF LOT 142 IN ROOSEVELT SOUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH. RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A

HORIZONTAL PLANE HAVING AN ELEVATION OF +34.82 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.80 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 2.86 FEET; THENCE SOUTH 88°22'51" WEST 2.97 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 22.39 FEET; THENCE SOUTH 88°22'51" WEST 15.21 FEET; THENCE SOUTH 01°37'09" EAST 5.59 FEET; THENCE SOUTH 88°22'51" WEST 6.93 FEET; THENCE SOUTH 01°37'09" WEST 4.39 FEET; THENCE SOUTH 88°22'51" WEST 12.43 FEET; THENCE NORTH 01°37'09" WEST 4.39 FEET; THENCE SOUTH 88°22'51" WEST 8.31 FEET; THENCE NORTH 01°37'09" WEST 24.00 FEET; THENCE NORTH 88°22'51" EAST 42.88 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L2-2

THAT PART OF LOT 142 'N ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANCE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.82 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONIAL PLANE HAVING AN ELEVATION OF +24.80 FEET ABOVE CHICAGO CITY DATUM, AND LYN'G WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 95.44 FEET; THENCE SOUTH 01°37'09" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 28.39 FEET; THENCE SOU (H 88°22'51" WEST 11.36 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 14./8 FFET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 10.61 FEET; THENCE NORTH 01°37'09" WEST 2.70 FEET; THENCE SOUTH 88°22'51" WEST 0.79 FEET; THENCE NONTH 1/1°37'09" WEST 25.69 FEET: THENCE NORTH 88°22'51" EAST 49.58 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT PARCEL L2-3

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLATTWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.82 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.80 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 118.05 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 24.48 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 11.47 FEET; THENCE NORTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L2-4

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.82 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +24.80 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 80°22'51" WEST 73.75 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 19.24 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" FAST 6.95 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.59 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.69 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE NORTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L3-1

THAT PART OF LOT 142 IN ROOSEVEL' SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST 'I'LLF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.86 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.82 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS FORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 2.86 FEET; THENCE SOUTH 88°22'51" WEST 70.68 FEET TO THE POINT OF PEGINNING; THENCE SOUTH 01°37'09" EAST 28.39 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET, THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 11.40 FEET; THENCE SOUTH 01°37'09" WEST 28.39 FEET; THENCE SOUTH 88°22'51" WEST 11.40 FEET; THENCE SOUTH 01°37'09" WEST 28.39 FEET; THENCE SOUTH 88°22'51" EAST 24.77 FEET TO THE POINT OF PEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L3-2

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.86 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +34.82 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 92.98 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 25.06 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET: THENCE NORTH

88°22'51" EAST 6.18 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 7.68 FEET: THENCE SOUTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L4-1

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.88 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.86 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 2.86 FEET; THENCE SOUTH 88°22'5!" WEST 2.97 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 01°37'09" EAST 22.39 FLET; THENCE SOUTH 88°22'51" WEST 15.21 FEET; THENCE SOUTH 01°37'09" EAST 5.59 FEET; THENCE SOUTH 88°22'51" WEST 6.93 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 12.43 FEET; THENCE NORTH 01°37'09" WEST 4.39 FEET; THENCE SOUTH 88°2751" WEST 8.31 FEET; THENCE NORTH 01°37'09" WEST 24.00 FEET; THENCE NORTH 88°22'51' FAST 42.88 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT PARCEL L4-2

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST NALE OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 FAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HOR ZONTAL PLANE HAVING AN ELEVATION OF +54.88 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.86 FEET APOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAIL TRACT 70.68 FEET; THENCE SOUTH 01°37'09" EAST 2.86 FEET TO THE POINT OF LEGINNING; THENCE SOUTH 01°37'09" EAST 28.39 FEET; THENCE SOUTH 88°22'51" WEST / 35 FEET: THENCE NORTH 01°37'09" WEST 0.40 FEET: THENCE SOUTH 88°22'51" WEST 6.02 FEET: THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 11.46 FEET; THENCE NORTH 01°37'09" WEST 28.39 FEET; THENCE NORTH 88°22'51" EAST 24.77 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT

PARCEL L4-3

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO. PLAT TWO. BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.88 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.86 FEET ABOVE

CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 120.22 FEET; THENCE SOUTH 01°37'09" EAST 2.86 FEET TO THE POINT OF BEGINNING: THENCE SOUTH 01°37'09" EAST 28.39 FEET; THENCE SOUTH 88°22'51" WEST 7.39 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09" WEST 2.70 FEET; THENCE SOUTH 88°22'51" WEST 10.61 FEET; THENCE NORTH 01°37'09" WEST 2.70 FEET; THENCE SOUTH 88°22'51" EAST 24.81 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCELT PARCEL L4-4

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF TART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, PANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN. ACCORDING TO A PLAT THEREOF RECONDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.88 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +44.86 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22 5.1' WEST ALONG THE NORTH LINE OF SAID TRACT 198.36 FEET; THENCE SOUTH 01°37'09" EAST 9.42 FEET; THENCE NORTH 88°22'51" EAST 7.80 FEET; THENCE SOUTH 01°37'09" EAST 6.07 FEET; THENCE NORTH 88°22'51" EAST 9.61 FEET; THENCE SOUTH 01°37'09" EAST 18.90 FEET; THENCE SOUTH 88°22'51" WEST 6.63 FEET; THENCE NORTH 01°37'09" WEST 34.38 FEET; THENCE NORTH 88°22'51" EAST 19.23 F.FT TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS:

ALSO EXCEPT PARCEL L5-1

THAT PART OF LOT 142 IN ROOSEVELT SOUARE SUBDIVISION PLANE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL ME VIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115 16, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.88 FLET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 2.85 FEET: THENCE SOUTH 88°22'51" WEST 2.97 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 22.39 FEET: THENCE SOUTH 88°22'51" WEST 15.21 FEET: THENCE SOUTH 01°37'09" EAST 5.59 FEET; THENCE SOUTH 88°22'51" WEST 6.93 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET: THENCE SOUTH 88°22'51" WEST 12.43 FEET: THENCE NORTH 01°37'09" WEST 4.39 FEET; THENCE SOUTH 88°22'51" WEST 8.31 FEET; THENCE NORTH 01°37'09" WEST 24.00 FEET: THENCE NORTH 88°22'51" EAST 42.88 FEET TO THE POINT OF BEGINNING. IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L5-2

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST OUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.88 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 145.03 FEET; THENCE SOUTH 01°37'40" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01237'09" EAST 25.69 FEET; THENCE NORTH 88°22'51" EAST 0.79 FEET: THENCE SOUTH 01°37'09' E.ST 2.70 FEET; THENCE SOUTH 88°22'51" WEST 12.52 FEET: THENCE NORTH 01°37'09" WLST 2.25 FEET; THENCE SOUTH 88°22'51" WEST 8.92 FEET; THENCE SOUTH 01°37'09" EAST 1.85 FEET: THENCE SOUTH 88°22'51" WEST 5.70 FEET; THENCE NORTH 01°37'09" WEST 10.46 FEET; THENCE SOUTH 88°22'51" WEST 21.97 FEET; THENCE NORTH 01°37'09" WEST 17.53 FEET; THENCE NORTH 88°22'51" EAST 48.32 FEET TO THE POINT OF BEGINNING. IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L5-3

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST, HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +54.88 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN INS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 118.05 FEET TO THE TOINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 24.48 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET, THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET, THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET, THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET, THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L6-1

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT 1WC, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.80 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 45.86 FEET; THENCE SOUTH 01°37'40" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 24.00 FEET; THENCE SOUTH 88°22'51" WEST 0.46 FEET; THENCE SOUTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 10.90 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09"

EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 7.44 FEET; THENCE NORTH 01°37'09" WEST 28.39 FEET; THENCE NORTH 88°22'51" EAST 24.82 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L6-2

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.80 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THEN'CE SOUTH 88°22'51" WEST ALONG THE NORTH LINE OF SAID TRACT 95.44 FEET; THENCE SOUTH, 91°37'40" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 6.02 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE SOUTH 88°22'51" WEST 7.39 FEET; THENCE NORTH 01°37'09" WEST 28.39 FEET; THENCE NORTH 88°22'51" WEST 7.39 FEET; THENCE NORTH 01°37'09" WEST 28.39 FEET; THENCE NORTH 88°22'51" WEST 7.39 FEET; THENCE NORTH 01°37'09" WEST 28.39 FEET; THENCE NORTH 88°22'51" EAST 24.77 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL L6-3

THAT PART OF LOT 142 IN ROOSEVELT SOUL RE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF O. THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE TH RD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.8/ I FET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 88°22'51" WEST ALONG THE NORTH LAND OF SAID TRACT 145.03 FEET; THENCE SOUTH 01°37'09" EAST 2.86 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 01°37'09" EAST 25.69 FEET; THENCE NORTH 88°22'51" EAST 0.79 FEET; THENCE SOUTH 01°37'09" EAST 2.70 FEET; THENCE SOUTH 88°22'51" WEST 12.52 FEET; THENCE NORTH 01°37'09" WEST 2.25 FEET; THENCE SOUTH 88°22'51" WEST 8.92 FEET; THENCE SOUTH 88°22'51" WEST 88°22'51" EAST 1.85 FEET; THENCE SOUTH 88°22'51" WEST 5.70 FEET: THENCE NORTH 0.1°27'09" WEST 10.46 FEET; THENCE SOUTH 88°22'51" WEST 18.87 FEET; THENCE SOUTH 01°37'09' FAST 10.46 FEET; THENCE SOUTH 88°22'51" WEST 27.34 FEET; THENCE NORTH 01°37'09" WEST 27.29 FEET; THENCE NORTH 88°22'51" EAST 72.56 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY. ILLINOIS:

ALSO EXCEPT

PARCEL L6-4

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO. BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.80 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE

CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 118.05 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 24.48 FEET; THENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 7.06 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 5.95 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 11.47 FEET; THENCE SOUTH 01°37'09" EAST 28.41 FEET TO THE POINT OF BEGINNING, IN COOK COUNTY, ILLINOIS;

ALSO EXCEPT PARCEL LG 3

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NGPTI, RANGE 14 EAST OF THE THIRD PRINCIPAL MERIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115116, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.80 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF +64.90 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT; THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID TRACT 65.24 FEET; THENCE SOUTH 88°22'51" WEST 73.75 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 19.24 FEET; FLENCE NORTH 01°37'09" WEST 28.41 FEET; THENCE NORTH 88°22'51" EAST 5.59 FEET; THENCE NORTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE NORTH 01°37'09" WEST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51" EAST 6.69 FEET; THENCE SOUTH 01°37'09" EAST 0.40 FEET; THENCE NORTH 88°22'51"

ALSO EXCEPT PARCEL L6-6

THAT PART OF LOT 142 IN ROOSEVELT SQUARE SUBDIVISION PHASE TWO, PLAT TWO, BEING A SUBDIVISION OF PART OF THE EAST HALF OF THE NORTHWEST QUARTER OF SECTION 20, TOWNSHIP 39 NORTH, RANGE 14 EAST OF THE THIRD PRINCIPAL METIDIAN, ACCORDING TO A PLAT THEREOF RECORDED JULY 20, 2007, AS DOCUMENT NO. 0720115/16, LYING BELOW A HORIZONTAL PLANE HAVING AN ELEVATION OF +74.80 FEET ABOVE CHICAGO CITY DATUM AND LYING ABOVE A HORIZONTAL PLANE HAVING AN ELEVATION OF ±(4.96 FEET ABOVE CHICAGO CITY DATUM, AND LYING WITHIN ITS HORIZONTAL BOUNDARY PROJECTED VERTICALLY AND DESCRIBED AS FOLLOWS: COMMENCING AT THE NORTHEAST CORNER OF SAID TRACT: THENCE SOUTH 01°35'55" EAST ALONG THE EAST LINE OF SAID 1.P.A.CT 65.24 FEET: THENCE SOUTH 88°22'51" WEST 2.96 FEET TO THE POINT OF BEGINNING; THENCE SOUTH 88°22'51" WEST 46.33 FEET: THENCE NORTH 01°37'09" WEST 18.50 FEET; THENCE NORTH 88°22'51" EAST 25.56 FEET; THENCE NORTH 01°37'09" WEST 9.51 FEET; THENCE NORTH 88°22'51" EAST 5.57 FEET; THENCE NORTH 01°37'09" WEST 11.97 FEET; THENCE NORTH 88°22'51" EAST 13.65 FEET; THENCE SOUTH 01°37'09" EAST 11.30 FEET; THENCE NORTH 88°22'51" EAST 1.56 FEET; THENCE SOUTH 01°37'09" EAST 28.68 FEET TO THE POINT OF BEGINNING), IN COOK COUNTY, ILLINOIS.

For informational purposes only: Commonly known as 1357 West Roosevelt Road, Chicago, IL 60608; PIN No. 17-20-102-057-0000

Together with:

FEE SIMPLE TITLE TO ALL BUILDINGS AND IMPROVEMENTS LOCATED, OR TO BE LOCATED AFTER THE DATE OF THE AFORESAID GROUND LEASE ON THE LEASEHOLD ESTATE HEREINABOVE DESCRIBED.

COOK COUNTY CLERK OFFICE
INLUMBING DOUBLE 14
118 N. CLASK ST. KIND IN 120
CHICAGO, IL ECEO2-1387

COOK COUNTY CLERK DATES RECORDING DIVISION 118 N. CLARK ST. ROOM 120 CHICAGO, IL 60602-1387

> COCK COUNTY CLERK OFFICE RECORDING DIVISION 1.8 N. CLARK ST. ROOM 120 CHICAGO, IL 60602-1387

EXHIBIT B

PERMITTED EXCEPTIONS

- 1. General Real Estate Taxes not yet due and payable.
- 2. Rights of Quasi-Public utilities for maintenance of utility facilities.
- 3. Exceptions listed on ALTA Loan Policy of Title Insurance Policy Number 41065324PB issue2 by Stewart Title Guaranty Company to Chicago Housing Authority.

COOK COUNTY CLIKE DEFICE
RUDROWN SWINDS
RES M. CLAPK ST. NOUN 120
CHICAGO, IL 60002-1327

T'S OFFICE

COUR COUNTY CLERK OF THE RECORDING DIVISION 1) & N. CLARK ST. ROOM 120 CHICAGO, IL SCEOZ-1327

EXHIBIT C INITIAL LEASEHOLD MORTGAGEES

BMO Harris Bank, N.A. 320 South Canal Street, 15th Floor Chicago, IL 60606

CIBC Bank USA 120 South LaSalle Street Chicago, IL 30603

With a copy to:

Charity & Associates, P.C. 20 S. Clark Street, Suite 500 Chicago, IL 60602 Attention: Elvin Charity

Chicago Housing Authority 60 E. Van Buren Street, 12th Floor Chicago, Illinois 60605 Attention: Chief Executive Officer

With a copy to:

Chicago Housing Authority Office of the General Counsel 60 E. Van Buren Street, 12th Floor Chicago, Illinois 60605

TOIT OF COUNTY CLOSES OFFICE Heartland Housing, Inc. 208 South LaSalle Street, Suite 1300 Chicago, IL 60604 Attention: Mark Kruse

With a copy to:

Applegate & Thorne-Thomsen P.C. 425 S. Financial Place. Suite 1900 Chicago, IL 60605 Attention: Gregory Whitehead

EXHIBIT D

ENCUMBRANCES

None

COOK SOURTY CLUK! OF JEE HECORDING DIVISION 128 N. C. AFK ST. ROCM?

EXHIBIT E

REGULATORY AGREEMENTS

- 1. Roosevelt Square 3B Redevelopment Agreement dated as of January 1, 2023 by and among City of Chicago, an Illinois municipal corporation (the "City"), RS Affordable I LLC, an Illinois limited liability company ("Owner"), and Heartland Housing, Inc., an Illinois not-for-profit corporation;
- 2. Low-Income Housing Tax Credits Regulatory Agreement dated as of January 1, 2023 by and between the City by and through the Department of Housing and Owner;
- 3. Lead Use Restriction Agreement dated as of January 1, 2023 by and between the City and Owner; and
- 4. Donations fax Regulatory Agreement dated as of January 1, 2023 by and among the City by and through the Denartment of Housing, Tenant and Owner.

E-1

EXHIBIT F

INSURANCE REQUIREMENTS

Tenant must procure and maintain at all times during the term of this Agreement the types of insurance specified below in order to protect the Authority from the acts, omissions and negligence of Tenant, its officers, officials, subcontractors, joint venture partners, agents or employees. The insurance carriers used by Tenant must be authorized to conduct business in the State of Illinois and shall have a BEST rating of not less than "A". The insurance provided shall cover all operations under this Agreement, whether performed by Tenant, any general contractor (including without limitation the General Contractor or by any subcontractors.

I. CONSTRUCTION INSURANCE REQUIREMENTS

A. Required Insurance Coverages for Owner:

- 1. <u>Builder's Risk.</u> The Developer and/or Owner Entity shall provide directly, on behalf of the Contractor and Subcontractors, for each Development, an All Risk Builder's Risk Insurance policy in accordance with HUD's 5370 form (paragraph 36b) covering new construction, improvements, betterments, and/or repairs, at replacement cost, for all materials, supplies, equipment, machinery and fixtures that are or will be part of the permanent project. The Authority and be named as loss payee.
- 2. General Liability Insurance. General Liability Insurance provided shall have a limit of not less than One Million Dollars (\$1,000,000) per occurrence with an aggregate of not less than Two Million Dollars (\$2,000,000) (i.e. \$1,000,000/\$2,000,000). In addition to the stipulations outlined above, the insurance policy is to include coverage for Contractual Liability, Products-Completed Operations, Fersonal & Advertising Injury and will also cover injury to Developer's and/or Owner Entity's and General Contractor's respective officers, employees, agents, subcontractors, invitees and guests and their personal property. The Authority is to be endorsed as additional insureds on the policy and such insurance will be endorsed as primary and non-contributory with any other insurance available to the Authority.
- 3. Excess Liability*. The Developer and/or Owner Entity shall secure Excess Liability insurance in the amount of not less than Ten Million Dollars (\$ 10,000,000). This coverage will be excess of the General Liability, Auto Liability and Employer's Liability coverage. The Developer's and/or Owner Entity's coverage will follow-form for all primary, liability and employer's liability coverages.
- 4. <u>Automobile Liability Insurance.</u> When any motor vehicle (owned, non-owned and/or hired) is used in connection with the construction to be performed for such Development, Comprehensive Automobile Liability

Insurance with limits of not less than One Million Dollars (\$1,000,000) per occurrence CSL, for bodily injury and property damage shall be provided. The Authority is to be endorsed as additional insured on the policy and such insurance will be endorsed as primary and non-contributory with any other insurance available to the Authority.

- 5. Workers Compensation and Occupational Disease Insurance. Workers Compensation and Occupational Disease Insurance shall be in accordance with the laws of the State of Illinois (Statutory) Coverage A, and Employer's Liability, Coverage B, in an amount of not less than \$500,000/\$500,000/\$500,000.
 - Professional Liability (Errors & Omissions). The Developer and/or Owner Entity shall require any architects and engineers of record, construction manager, property manager, security companies and/or other professional consultants who perform work in connection with the Redevelopment to provide Professional Liability Insurance. Such insurance covering acts, errors or omissions of architects and engineers of record, and the construction manager shall be maintained with limits of not less than Two Million Potlars (\$2,000,000) per occurrence. Such insurance covering acts, errors or omissions of the property manager, security companies and/or other professional equivalents shall be maintained with limits of not less than Five Million Dollars (\$5,000,000) per occurrence. When policies are renewed or replaced, the policy retroactive date must coincide with, or precede, start of Services under this Agreement. A Claims-made policy which is not renewed or replaced must have an extended reporting period of two (2) years.

B. Required Insurance Coverages for the General Contractor:

1. General Liability Insurance. Liability Insurance provided shall have a limit of not less than One Million Dollars (\$1.000.00%) per occurrence with a Per Project aggregate of not less than Two Million Dollars (\$2,000,000) (i.e. \$1,000,000/\$2,000,000). Included without limitation the following coverages: Property/Operations, including deletion of explosion, collapse and underground (XCU) exclusions; Independent Contractors Protective Liability; Broad Form Contractual Liability, specifically referring to the indemnity obligations under and pursuant to this Agreement, subject to the standard industry terms, conditions and exclusions of the policy; Broad Form Property Damage, including Products/Completed Operations; Personal Injury Liability, with employee and contractual exclusions deleted. In addition to the stipulations outlined above, the insurance policy is to include coverage for Contractual Liability, Products-Completed Operations, Personal & Advertising Injury. The Authority, Developer and/or Owner Entities ("Additional Insureds"), are to be endorsed as additional insureds on the policy and such insurance will be endorsed as

primary and non-contributory with any other insurance available to the Additional Insureds.

Products-Completed Operations. General Contractor and subcontractors shall procure and maintain (and require subcontractor's subcontractors of any tier to procure and maintain) until expiration of the Roosevelt Square 3 Project's warranty period and, with regard to Products/Completed Operations coverage for two (2) years after final completion of the Work.

It is further agreed that the coverage afforded to the Additional Insureds shall exclude indemnification of the architect for claims arising out of (1) the preparation or approval of maps, drawings, opinions, reports, surveys, change orders, designs or specifications, or (2) the giving of or the failure to give directions or instructions by the architects, his agents or employees provided such giving or failure to give is the primary cause of the injury or damage.

- 2. Excess Liability*. The General Contractor shall secure Excess Liability insurance in the amount of not less than Ten Million Dollars (\$10,000,000) Per Occurrence. This coverage will be excess of the General Liability, Auto Liability and Employers Liability coverages. The General Contractor's insurance coverage will be excess of all subcontractors with which it contracts to provide services for this development. The Authority and the Developer and/or Owner Entities are to be endorsed as additional insureds on the General Contractor's Excess Liability policy. Subcontractor's excess limits will be determined by the General Contractor as they deem appropriate.
- Automobile Liability Insurance. When any motor vehicles (owned, non-owned and/or hired) are used in connection with the construction to be performed for the Development, Comprehensive Automobile Liability Insurance with limits of not less than One Million Deilars (\$1,000,000) per occurrence CSL, for bodily injury and property damage shall be provided. The Authority, Developer and/or Owner Entities, are to be endorsed as additional insureds on the policy and such insurance will be endorsed as primary and non-contributory with any other insurance available to the Authority.
- 4. Workers Compensation and Occupational Disease Insurance. Workers Compensation and Occupational Disease Insurance shall be in accordance with the laws of the State of Illinois (Statutory) Coverage A, and Employer's Liability, Coverage B, in an amount of not less than \$500.000/\$500,000/\$500,000.
- 5. <u>Lead/Asbestos Abatement Liability</u>. When any lease and/or asbestos abatement liability work is performed in connection with the contract, Lead/Asbestos Abatement Liability Insurance shall be provided with limits

of not less than Five Million Dollars (\$5,000,000) per occurrence insuring bodily injury, property damage and environmental clean-up. The Authority, Developer and/or Owner Entities, are to be endorsed as additional insureds on the policy and such insurance will be endorsed as primary and non-contributory with any other insurance available to the Authority. When claims made policies are renewed or replaced, the policy retroactive date must coincide with, or precede, start of Services under this Agreement. A Claims-made policy which is not renewed or replaced must have an extended reporting period of two (2) years.

- 6. Contractor's Pollution Liability. The General Contractor shall require a separate Contractor's Pollution Liability insurance policy, covering any bodily injury, liability and property damage liability, arising out of pollutants including hazardous materials such as asbestos, lead, etc. or contaminated soil, including while in transit to a permanent disposal facility which may arise from activities under or incidental to the contract, whether such activities be by the General Contractor or by any of his subcontractors or by any one directly or indirectly employed or otherwise contracted by any of them. This policy shall be maintained with limits of not less than Three Million Dolla's (\$3,000,000) per occurrence. The Authority is to be endorsed as an additional insured on the policy and such insurance will be endorsed as primary and non-contributory with any other insurance available to the Authority.
- Railroad Protective Liability Insurance. When, in connection with a Development, any work is to be done adjacent to or on property owned by a railroad or public transit entity, the Ceneral Contractor shall procure and maintain, or cause to be procured and maintained, with respect to the operations that contractor or any subcontractor shall perform, railroad protective liability insurance in the name of such railroad or public transit entity. The policy shall have limits of not less a an Two Million Dollars (\$2,000,000) per occurrence, combined single limits, and Six Million Dollars (\$6,000,000) in the aggregate for losses arising out of injuries to or death of all persons, and for damage to or destruction of property, including the loss of use thereof.
- C. <u>Evidence of Insurance</u>. Within five days of initial construction closing disbursement on a Phase and prior to the commencement of construction activities, the Developer and/or Owner Entity directly or through the General Contractor shall furnish the Authority, for record keeping purposes only, with satisfactory evidence that the Developer and/or Owner Entity, General Contractor and subcontractors have the insurance coverages set forth above. The Developer, Owner Entity, and/or General Contractor shall be required to ensure that all subcontractors comply with the Authority's minimum coverage requirements. It is the responsibility of the Developer, Owner Entity and/or General Contractor to secure and maintain proof of coverage for all entities that it contracts with that provide services to this Development. At closing, final draft certificates of insurance records previously approved by the Authority must be delivered. Post-closing certificates must be made available for review by the Authority within

twenty-four (24) hours of being requested. Said coverages shall not be modified, canceled, non-renewed, or permitted to lapse until final completion and approval of the performance of the General Contractor's contract and the policies shall contain a provision that the coverages will not be modified, canceled, non-renewed or permitted to lapse until not less than 30 days after the Authority has received written notice, by certified or registered mail, that the modification, cancellation, non-renewal or lapse of such coverages is contemplated.

ALL REQUIRED DOCUMENTATION MUST BE RECEIVED FOR APPROVAL PRIOR TO DEVELOPER AND/OR OWNER ENTITY COMMENCING WORK UNDER THIS AGREEMENT.

- Developer and/or Owner Entity shall advise, and cause each General Contractor for a Development c advise, all insurers of the contract provisions regarding insurance. The failure of the Developer, Gwner Entity, or any General Contractor to notify insurers of the contract provisions shall not relieve Developer and/or Owner Entity from its insurance obligations under this contract or any Authority Closing Document and such insurance obligations shall survive the term of this Development Agreement. Nonfulfillment of the insurance provisions stated herein shall constitute a breach of the Ceneral Contractor's contract and of this Agreement. The Authority retains the right to stop work until proper evidence of insurance is provided.
- Renewal Certificates of Insurance, requested endorsements, or such similar evidence is to be received by the Authority's Risk Management Department, with a copy to the Authority's designated representative, prior to (xp) ation of insurance coverage. At the Authority's option, non-compliance may result in one or more of the following actions, in addition to any rights or remedies in the Closing Documents: (1) In: Authority will purchase insurance on behalf of Developer and/or Owner Entity and will charge back all costs to Developer and/or Owner Entity; (2) Developer or Owner Entity shall cause the General Contractor and any subcontractors to be immediately removed from the property; (3) this Development Agreement will be terminated; or (4) all payments due Developer and/or Owner Entity and General Contractor will be held until Developer and/or Owner Entity has complied with the contract. The receipt of any certificate by the Authority does not constitute agreement by the Authority that the insurance requirements in the Agreement have been fully met or that the insurance policies indicated on the certificate are in compliance with the requirements of the Agreement. Developer, Owner Entity, and/or General Contractor shall be required to ensure that all subcontractors comply with the Authority's minimum coverage requirements. It is the responsibility of the Developer, Owner Entity, and/or General Contractor to secure and maintain proof of coverage for all entities that it contracts with, that provides services to this Development. Proof of insurance records must be available for review by the Authority within twenty-four (24) hours of being requested.
- F. If any of the required insurance is underwritten on a claims made basis, the retroactive date shall be prior to or coincident with the date of the General Contractor's contract, and the Certificate of Insurance shall state the coverage is "claims made" and also the Retroactive Date. A Claims-Made policy which is not renewed or replaced must have an extended reporting period (tail coverage) of two (2) years. Any extended reporting period premium (tail coverage) shall be paid by Developer and/or Owner Entity, directly or through the General Contractor for the Development. It is further agreed that all insurance policies required hereunder shall provide the Authority with not less than a thirty (30) days notice in the event of the occurrence of any of the

following conditions: aggregate erosion in advance of the Retroactive Date, cancellation and/or non-renewal.

- **G.** Developer and/or Owner Entity shall provide to the Authority, prior to commencement of construction and upon each renewal or replacement of a builder risk policy required hereunder, and in any event not less than annually, a certified copy of the insurance policies required hereunder and all endorsements.
- H. Developer and/or Owner Entity shall require, directly or through the General Contractor for each Development, that all subcontractors performing work for the Development carry insulance required herein or the Developer, or General Contractor may provide the coverages for any or all of its subcontractors, and if so, the evidence of insurance submitted shall so stipulate and adhere to the same requirements and conditions as outlined in Section "B" above. Evidence of such coverage must be submitted to the Authority for record keeping purposes only.

II. OPERATIONS VERIOD INSURANCE REQUIREMENTS

The Owner Entity must procure and maintain at all times during the operation of the Development the types of insurance specified below in order to protect the Landlord from the acts, omissions and negligence of the selected respondent, its officers, officials, subcontractors, joint venture, partners, agents or employees. The insurance carriers used by the Owner Entity must be authorized to conduct business in the State of Illinois and shall have a BEST Rating of not less than an "A". The insurance provided shall cover all operations under the Agreement, whether performed by the Owner Entity or by its subcontractors.

A. Required Insurance Coverages:

- All-Risk Property Damage: The Owner Entity shall obtain an all-risk property policy in the amount of the full replacement value, including improvements and betterments, covering damage to or loss of the Property. The insurance shall include the following extensions: business interruption/loss of rents, and boiler and machinery if applicable. The policy shall list the Authority as loss payee.
- 2. General Liability Insurance. General Liability Insurance provided shall have a limit of not less than One Million Dollars (\$1,000,000) per occurrence with a Per Project aggregate of not less than Two Million Dollars (\$2,000,000) (i.e. \$1,000,000/\$2,000,000). In addition to the stipulations outlined above, the insurance policy is to include coverage for Contractual Liability, Products-Completed Operations, Personal & Advertising Injury and will also cover injury to the Owner Entity's officers, employees, agents, subcontractors, invitees and guests and their personal property. The Landlord is to be endorsed as an additional insured on the policy and such insurance will be endorsed as primary and non-contributory with any other insurance available to the Authority.
- **3. Automobile Liability Insurance.** When any motor vehicles (owned, nonowned and hired) are used in connection with the Services to be performed,

the Owner Entity shall provide Comprehensive Automobile Liability Insurance with limits of not less than One Million Dollars (\$1,000,000) per occurrence CSL, for bodily injury and property damage. The Landlord shall be endorsed as additional insureds on the Owner Entity's policy and such insurance will be endorsed as primary and non-contributory with any other insurance available to the CHA.

- 4. Workers Compensation and Occupational Disease Insurance. Workers Compensation and Occupational Disease Insurance shall be in accordance with the laws of the State of Illinois (Statutory) Coverage A, and Employer's Liability, Coverage B, in an amount of not less than \$500,000/\$500,000/\$500,000.
- Blanket Crime. The Owner Entity shall provide Blanket Crime coverage in a form reasonably acceptable to the Authority, against loss by dishonesty, to bery, burglary, theft, destruction or disappearance, computer fraud, credit card forgery and other related crime risks. The policy limit shall be written 15 cover losses in the amount of the maximum monies collected, received and/or in the Owner Entity's care at any given time, but shall in no event be less that the aggregate amount of two months operating subsidy.
- **Professional Liabitity.** When any architects (of record), engineers (of record), construction managers, property managers or other professional consultants perform work in connection with this contract, Professional Liability insurance covering acts, errors or omissions shall be maintained with limits of not less than Five Million Dollars (\$2,000,000) per occurrence for architects and engineers or record, and construction managers and of not less than Five Million Dollars (\$5,000,000) per occurrence property managers and other professional consultants. When policies are renewed or replaced, the policy retroactive date must coincide with, or precede, start of Services under this Agreement. A Claims-made policy which is not renewed or replaced must have an extended reporting period of two (2) years.

B. Related Requirements

- 1. The Owner Entity shall advise all insurers of the contract provisions regarding insurance. The failure of the Owner Entity to notify insurers of the contract provisions shall not relieve the Owner Entity from its insurance obligations herein. Nonfulfillment of the insurance provisions shall constitute a default under the Authority Closing Documents. The Authority retains the right to stop work until proper evidence of insurance is provided.
- 2. The Owner Entity shall furnish the Chicago Housing Authority, Risk Management Department, 60 E. Van Buren St., 11th Floor, Chicago, Illinois 60605 original Certificates of Insurance evidencing the required coverages to be in force on the Effective Date of this Agreement. In addition, copies

of the endorsement(s) adding the Landlord to the policies as additional insured are required.

- 3. Renewal Certificates of Insurance, requested endorsements, or such similar evidence is to be received by the Risk Management Department, with a copy to the Authority's Designated Representative prior to expiration of insurance coverage. At the Authority's option, non-compliance may result in one or more of the following actions, in addition to any rights or remedies in any Closing Documents: (1) The Authority will purchase insurance on behalf of the Owner Entity and will charge back all costs to the Owner Entity; (2) the Owner Entity will be immediately removed from Authority property and contract revoked; (3) all payments due the Owner Entity will be held until the Owner Entity has complied with the contract; or (4) the Owner Entity will be assessed Five Hundred Dollars (\$500) for every day of non-compliance. The receipt of any certificate does not constitute agreement by the Authority that the insurance requirements in the Agreement have been fully met or that the insurance policies indicated on the certificate are in compliance with the requirements of the Agreement. The insurar ce policies shall provide for thirty (30) days written notice to be given to the Authority in the event coverage is substantially changed, canceled or non-tenewed.
- 4. If any of the required insurance is underwritten on a claims made basis, the retroactive date shall be prior to or coincident with the date of this Agreement and the Certificate of Insurance shall state the coverage is "claims made" and also the Retroactive Date. The Owner Entity shall maintain coverage for the duration of this Agreement. A Claims-Made policy which is not renewed or replaced must have an extended reporting period (tail coverage) of two (2) years. Any extended reporting period premium (tail coverage) shall be paid by the Owner Entity. The Owner Entity shall provide to the Authority, annually, a certified copy of the insurance policies obtained pursuant hereto. It is further agreed that the Owner Entity shall provide the Authority a thirty (30) days notice in the event of the occurrence of any of the following conditions: aggregate erosion in advance of the Retroactive Date, cancellation and/or non-renewal.
- 5. The Owner Entity shall require all subcontractors to carry the insurance required herein or the Owner Entity may provide the coverage for any or all of its subcontractors, and if so, the evidence of insurance submitted shall so stipulate and adhere to the same requirements and conditions as outlined in Section "A" above.

^{*} Excess Liability can be satisfied by \$10,000,000 each of Developer and General Contractor, \$20,000,000 in Excess Liability coverage by either the Developer or General Contractor or any combination equal to \$20,000,000 in total Excess Liability coverage.

2301808080 Page: 99 of 99

UNOFFICIAL COPY

RIDER TO GROUND LEASE

THIS RIDER is attached to and made a part of that certain Ground Lease, dated January 10, 2023 (the "Lease") by and between Chicago Housing Authority ("Landlord") and Heartland Housing, Inc., an Illinois not for profit corporation ("Tenant"). All of the capitalized terms used in this Rider that are not defined herein shall have the same meanings as defined in the Lease. In the event of a conflict, inconsistency or ambiguity between the provisions contained in the Lease and this Rider, the provisions of this Rider shall prevail.

R-1 The portion of the overall Development to be constructed on the Real Estate consists of four (4) Exillings containing one hundred thirty (130) affordable housing units (the "Units") and related improvements which include Improvements more particularly described on the plans and specifications submitted to and approved by Landlord ("Plans and Specifications"). The Plans and Specifications also include the construction of ninety two (92) Market Units and related improvements for the Market Development. The number and size of the Units in the Development are as follows:

9 - Studios

74 - One-bedroom units

39 - Tv o-bedroom units

8 - Threy-bedroom units

R-2 The Construction Completion Deadline for the Development is thirty-six (36) months after the date of the Lease, subject to extensions for Upavoidable Delays.

Tenant (and any successors and permitted assigns) shall construct the Development in accordance with the Plans and Specifications.